

PIRLS 2021: National Report for England

Research report

April 2024

Ariel Lindorff, Jamie Stiff & Heather Kayton: University of Oxford



Government Social Research

Acknowledgements

This report is the culmination of a collaborative effort between several organisations over several years.

We would like to acknowledge the International Association for the Evaluation of Educational Achievement (IEA), the International TIMSS & PIRLS Study Centre at Boston College, and Statistics Canada for their support and guidance throughout the project.

We are grateful for the ongoing feedback, support and flexibility of the team in the Department for Education (DfE) that oversaw the work, and in particular Naomi Maguire, Aliki Pareas, Rachel Grant and Thomas Lockhart. We would also like to acknowledge the DfE Data Sharing team for their stellar support and responsiveness, without which the linking of National Pupil Database and PIRLS data used in some of the key analyses in this report would not have been possible.

We would also like to thank the team at Pearson, particularly Grace Grima, Kate Miller, Alistair Hooper, Ellen Barrow, Mish Mohan and Sarah Turner for their exemplary project management, teamwork and support, and in particular for conducting the PIRLS assessment and data collection process in England to a very high standard despite the challenges introduced by the COVID-19 pandemic. We also acknowledge the test administrators, the scorers responsible for coding the assessments, and the data processing team.

We would like to acknowledge several additional researchers who were part of the original Oxford University Centre for Educational Assessment research team, including Joshua McGrane, Therese Hopfenbeck and Kit Double, who contributed to the planning of the structure of this report and contributed outputs in the earlier stages of the project.

Finally, we are especially grateful to the staff and pupils in participating schools for their time and effort in administering and completing the PIRLS reading assessment and questionnaires.

Corrections

The authors have made minor amendments to the original version of this report, detailed in Appendix E.

Contents

E	xecutiv	e Summary	5
	About	PIRLS	5
	Impac	t of COVID-19 on PIRLS 2021	5
	Key hi	ghlights from PIRLS 2021	6
	Overa	Il reading performance	7
	The at	tainment gap in reading performance	8
	Readi	ng performance and prior attainment	8
	Readi	ng performance differences by pupil characteristics	9
	Pupils	' attitudes towards reading	10
	Teach	er and school characteristics	11
	Educa	tional policy in England and PIRLS	12
1	Intro	oduction to PIRLS	13
	1.1	What is PIRLS?	13
	1.2	Interpreting data from PIRLS: a reader's guide	17
	1.3	Overview of the report structure	27
2	Rea	ding performance	30
	2.1	Overall reading performance	30
	2.2	Trends in overall reading performance	34
	2.3	Overall performance according to International Benchmarks	38
	2.4	Contextualisation: Average PIRLS performance in European education system 41	ms
3	Per	formance in reading purpose and process scales	43
	3.1	Performance in reading purposes	43
	3.2	Performance in reading comprehension processes	49
	3.3 Northe	Contextualisation: Reading for Literary Purposes and curricula in Ireland and ern Ireland	55
4	Rea	ding performance of higher and lower performing pupils	58
	4.1	Distribution of reading performance	58
	4.2	Trends in performance of lower and higher performing pupils	60

	4.3 Singaj	Contextualisation: Achievement of higher and lower performing pupils in pore	66
5	Rea	iding performance by prior attainment	69
	5.1	PIRLS performance by prior performance in phonics screening checks	70
	5.2	PIRLS performance by prior performance in key stage 1 reading	73
	5.3	Contextualisation: Approaches to reading screening across education system 75	ns
6	Rea	ding performance by pupil characteristics	78
	6.1	The relationship between PIRLS performance and pupil characteristics	79
	6.2	Performance by gender	83
	6.3	Performance by pupil age	94
	6.4	Performance by ethnic group and English as an Additional Language (EAL)	96
	6.5	Performance by socioeconomic background	97
	6.6	Contextualisation: Changes in the magnitude of the gender gap over time	98
7	Rea	ding performance by pupils' motivations	101
	7.1	Scales for pupils' motivations towards reading	102
	7.2	Pupils' confidence in reading	102
	7.3	Pupils' liking of reading	108
	7.4	Pupils' engagement in reading lessons	114
	7.5	Time spent reading outside of school	121
	7.6 achiev	Contextualisation: The relationship between reading enjoyment and reading rement in Norway	123
8	Sch	ool environment and teacher characteristics	127
	8.1	Teacher characteristics and teaching practices	128
	8.2	School characteristics	136
	8.3	Parental involvement	141
9	Imp	act of the COVID-19 pandemic	146
	9.1	Introduction	147
	9.2	Impact of COVID-19 on data collection	147
	9.3	Impact of COVID-19 on school operations	148
	9.4 2016-2	Expected performance trends 2016-2021 vs observed performance trends 2021	151

10 PIRLS and England's educational policy context		
10.1 Current and past policy developments in England	156	
10.2 Concluding remarks on policy changes in England and PIRLS	162	
References	163	
List of figures	169	
List of tables		
Appendices		
Appendix A: Methodology	176	
Appendix B: Missing data from the NPD	181	
Appendix C: Test languages in PIRLS 2021	182	
Appendix D: Teachers' perspectives on parental involvement	185	
Appendix E: Corrections included in this version of the report	187	

Executive Summary

About PIRLS

The Progress in International Reading Literacy Study (PIRLS) is an international comparative study directed by the International Association for the Evaluation of Educational Achievement (IEA). The aim of PIRLS is to assess and compare the reading performance of pupils at approximately 10 years of age – that is, in what is internationally considered 'fourth grade', or year 5 in England. A total of 57 education systems took part in PIRLS 2021.

England has taken part in PIRLS cycles every 5 years since 2001, though for PIRLS 2021 this involved a delay to data collection until 2022 due to the impact of the COVID-19 pandemic. In 2021, England's sample consisted of 4,150 year 5 pupils from 162 primary schools.

PIRLS focuses on 3 different aspects of reading literacy – how pupils read different types of texts, what reading comprehension processes pupils use to understand those texts, and what attitudes pupils have towards reading. The first 2 aspects are assessed through a reading literacy test that bears many similarities to the types of comprehension tests that pupils in England sit at school, such as the key stage 2 reading test. Attitudes towards reading are assessed through the use of a questionnaire completed by pupils after finishing their test. Questionnaires are also completed by these pupils' teachers and headteachers, providing additional information about their reading lessons and wider school environments. Most education systems also administer a questionnaire to the parents/guardians of participating pupils for further contextual information on reading-related activities at home, but the home questionnaire was not included in England for PIRLS 2021.

Impact of COVID-19 on PIRLS 2021

The global COVID-19 pandemic had an impact on data collection for PIRLS as well as on normal school operations in England and other education systems internationally. The available data do not allow us to measure that impact in a precise way, which complicates both international comparisons and trends over time for PIRLS in the 2021 cycle.

England and 5 other education systems delayed data collection for 12 months as a result of COVID-19. The majority of participating education systems continued with data collection as originally planned, and others delayed by 6 months but tested older pupils at the beginning of their fifth year of formal schooling. This means that direct statistical comparisons to some education systems are not possible to make in this cycle of PIRLS, because of the complexities introduced by comparing across different ages of pupils and varying impact of COVID-19. Following the approach used in the PIRLS 2021 International Report, this report will not focus on direct comparisons between England and education systems that delayed participation by 6 months and tested an older cohort of pupils.

Key highlights from PIRLS 2021

England's average score in PIRLS 2021 was 558. This is statistically significantly¹ higher than the International Median score of 520, though not significantly different to England's scores in most previous PIRLS cycles. England's average overall reading performance has not changed significantly compared to most previous PIRLS cycles, including PIRLS 2016. By contrast, most education systems in PIRLS 2021 experienced significant drops in overall performance since 2016, but this may reflect the fact that many systems collected data for PIRLS during the COVID-19 pandemic.

PIRLS uses 4 International Benchmarks (Low, Intermediate, High and Advanced) to describe the reading skills associated with PIRLS scores. Almost all pupils in England (97%) reached at least the Low International Benchmark, compared to an International Median of 94%, while 86% of pupils in England reached the Intermediate Benchmark level compared to an International Median of 75%. More than half of pupils in England (57%) reached the High International Benchmark in PIRLS 2021, compared to an International Benchmark, and 18% of pupils in England reached the Advanced International Benchmark, compared to the International Median of 7%. Between 2011 and 2016 there was a small increase in the percentage of pupils achieving the Low, Intermediate and High Benchmarks in England. Since 2016, performance at each of the International Benchmarks in England has seen no statistically significant changes.

The gender gap in reading performance has reduced over time in England. Girls still outperform boys by 10 points on average in PIRLS 2021, but this is lower than in previous cycles. Between 2001 and 2011 the gender gap remained relatively consistent with a 22-point difference in 2001, 19 points in 2006 and 23 points in 2011. The gender gap then narrowed to a 15-point difference in 2016 and narrowed further to a 10-point difference in 2021.

The gap between the lowest-scoring and highest-scoring pupils in England has also reduced over time. In the long term, looking across all cycles of PIRLS, this seems

¹ Throughout the report, 'significantly' is used to mean 'statistically significantly at least at the level of p<0.05'.

to be the result of increases in performance amongst the lowest-attainers, while highattainers' average score has remained relatively stable.

Fewer pupils in England reported that they enjoy reading than in previous cycles. Across all education systems pupils who like reading achieve higher on average in PIRLS than pupils who do not like reading. Some attitudes towards reading vary by gender in England; a higher proportion of boys do not like reading compared to girls,

however, both boys and girls report similar levels of confidence in reading. In England, most pupils attend schools where teachers and headteachers report high levels of emphasis on academic success; a safe and orderly school environment; and "hardly any" problems with school discipline. Across all participating education systems, high levels of academic emphasis at schools are associated with higher average performance for pupils in PIRLS 2021. There is also a

positive association between higher achievement in PIRLS and a safe school climate, and low reports of discipline-related problems.

Almost half of pupils who participated in PIRLS 2021 in England report experiencing bullying at least monthly at school. Across all participating education systems, pupils who report experiencing bullying more frequently had lower average performance in PIRLS 2021.

There is a positive correlation between performance in the year 1 phonics screening check and performance in PIRLS 2021. The overall correlation between year 1 phonics check and PIRLS 2021 was 0.43, indicating a moderate, statistically significant relationship between performance in the 2 assessments. This is similar to the relationship seen between PIRLS and phonics scores in 2016.

Several pupil characteristics significantly predict PIRLS 2021 performance in England based on a multiple linear regression analysis. **The strongest predictor of PIRLS performance was the year 1 phonics check mark**, for which a 1-point increase was associated with nearly a 4-point gain in PIRLS 2021 overall reading performance. Number of books at home was the second most powerful predictor of overall reading score, with higher numbers of books associated with higher PIRLS scores. This was followed by eligibility for free school meals (FSM).

Overall reading performance

Pupils in England achieved an overall average of 558 for PIRLS 2021. Average PIRLS 2021 achievement for England was higher than the International Median of 520. It is important to note that due to disruptions related to the COVID-19 pandemic, 14 participating education systems delayed their data collection timeframe and assessed an older cohort of pupils at the beginning of fifth grade (these education systems are not

included in the calculation of the International Median). Reading achievement in England was higher than 39 of 42 other education systems that assessed pupils at the end of fourth grade (equivalent to year 5 in England). This means that reading achievement for year 5 pupils in England is higher than the international average, and higher than achievement in a majority of participating education systems.

England has participated in PIRLS since 2001 and the average performance of pupils across each cycle of PIRLS has fluctuated somewhat over time. After a decline in performance between 2001 and 2006, average achievement increased steadily until 2016. Between 2016 and 2021, the average reading achievement of year 5 pupils in England has remained stable. The stability of the trend in England between 2016 and 2021 is different to the overall trend internationally over this period. The International Median dropped 19 points between 2016 and 2021, and most education systems scored significantly lower in PIRLS 2021 than they did in PIRLS 2016. It is important to note that England is 1 of 6 other education systems that delayed data collection by a year due to COVID-19 disruptions (these education systems are included in the calculation of the International Median).

Alongside a measure of overall reading achievement, PIRLS also provides information about reading performance for pupils reading for 2 different purposes: Literary and Informational. PIRLS 2021 results reveal that pupils in England perform equally well when reading for Literary purposes and reading for Informational purposes. This is a change from previous cycles where pupils in England typically scored significantly higher on the Literary Scale.

The attainment gap in reading performance

The attainment gap between higher-performing pupils and lower-performing pupils has historically been wider in England than in many other education systems. In 2016 the attainment gap for PIRLS reduced substantially. PIRLS 2021 results show that the gap between the lowest-performing and highest-performing pupils in England has continued to narrow. Long-term trends for England reveal that the narrowing of the attainment gap is due to an increase in the scores of the lower achievers rather than a decrease in the scores of the higher achievers.

Reading performance and prior attainment

The relationship between prior attainment on the year 1 phonics screening check and PIRLS results were first explored in 2016. Similar to what was seen in 2016, there is a moderate correlation of 0.43 between pupils' performance on their year 1 phonic check and their performance in PIRLS 2021. This means that, generally, pupils that performed well in their year 1 phonics check also performed well in PIRLS 2021. Pupils who

achieved 100% for the year 1 phonics check had a median PIRLS score of 606, considerably higher than the average PIRLS score for England of 558. Pupils who just met the expected standard for the year 1 phonics check (i.e., achieved 80%, or 32/40) had a median PIRLS score of at least 521, lower than the average PIRLS score for England but in line with the International Median for PIRLS 2021. Pupils who did not meet the expected standard for the year 1 phonics check (achieved below 80%) had below average achievement in PIRLS 2021.

Pupils who do not meet the expected standard in the year 1 phonic check participate in the year 2 phonics screening check in the next academic year. A similar relationship was found between the year 2 phonics check and PIRLS 2021 scores, where stronger performance in the year 2 phonics screening check was generally associated with stronger performance in PIRLS 2021. Although there was a wide range of performance for pupils who participated in the year 2 phonics check, on average, pupils who just met the expected standard in the year 2 phonics check (achieved 80% or 32/40) still had below-average PIRLS 2021 achievement.

Pupils' attainment in key stage 1 reading assessments also showed a relationship with their PIRLS 2021 achievement. Pupils who were able to read at a 'greater depth' according to their key stage 1 assessments had high achievement in PIRLS 2021, with a median PIRLS score of 610. By contrast, for pupils working at the expected reading standard in the key stage 1 assessments, their median PIRLS score was 558, in line with England's average score (558) in PIRLS 2021.

Reading performance differences by pupil characteristics

To understand the impact of pupil characteristics on the overall reading performance of pupils in England, information from the National Pupil Database (NPD) was matched with data from PIRLS 2021. We conducted multiple linear regression analysis to understand which of the selected pupil characteristics predicted pupils' reading performance in PIRLS 2021. Of the 8 characteristics considered, pupils' score in the year 1 phonics check was the most powerful predictor of PIRLS 2021 reading scores. Every 1-point increase on the year 1 phonics screening check is associated with almost a 4-point gain in PIRLS. The number of books in the home also had a positive impact on PIRLS 2021 reading scores; pupils who reported having over 200 books at home scored approximately 56 points higher in PIRLS than those who reported having 10 or fewer books at home. Eligibility for FSM within the last 6 years was a strong negative predictor of PIRLS achievement. FSM-eligible pupils scored about 23 points lower than their peers who were not FSM eligible, after accounting for other pupil characteristics and school attainment band.

Being in a mid-high or high performing school, being in the 'Mixed' ethnic group and being born earlier within the school year were also significant positive predictors of PIRLS scores, but not powerful predictors based on their effect sizes (standardized coefficients). After accounting for all other variables mentioned above, gender and English as an Additional Language (EAL) did not significantly predict PIRLS score.

The gap between reading performance of boys and girls in PIRLS is historically wide. In PIRLS 2021, the gender gap is still evident; in all participating education systems girls achieve higher average scores than boys, and in the vast majority of education systems the difference was significant. In England, girls scored, on average, 10 points higher than boys overall. In PIRLS 2011, England had one of the largest gender gaps and the largest gap of any participating European education system. In 2016, this gap narrowed and in 2021 the gap narrowed further. In 2016, the reduction was largely attributable to an average improvement in the performance of boys, up 11 points from 2011. However, in 2021 the narrowing of the gap reflects a decrease of 4 points in the average achievement of girls and a 2-point improvement in boys scores.

Pupils' attitudes towards reading

Pupils that participated in PIRLS 2021 complete both the reading assessment as well as a questionnaire that asks about their attitudes towards reading. Findings from PIRLS 2021 internationally reveal that pupils who enjoy reading have higher average reading achievement. In England, pupils who like reading scored an average of 34 points higher in PIRLS 2021 than those who do not like reading. Internationally 46% of pupils said that they very much like reading, this compares to 29% of pupils in England who very much like reading. The proportion of pupils in England reporting that they enjoy reading is lower than in previous cycles. Confidence in reading has one of the strongest associations with reading achievement in PIRLS. In England, there is a 90-point difference between the average PIRLS 2021 scores for pupils who feel very confident in reading and those who do not feel confident in reading. In 2016, just over half (53%) of pupils in England felt very confident in reading, this has decreased slightly in 2021 to 45%.

Gender differences are seen in attitudes towards reading in England, and across most education systems. In England, a similar proportion of girls (46%) and boys (44%) report feeling confident in reading, while the proportion of girls that like reading (32%) is higher than the proportion of boys that like reading (25%).

The relationship between reading attitudes and pupil characteristics such as having more books at home, FSM-eligibility, ethnic group and EAL-status was also considered. Pupils with more books at home also show higher levels of confidence in reading, while pupils who had been eligible for FSM in the past 6 years were less likely to be very confident. Within ethnic groups, higher confidence tended to be associated with higher reading

performance where the number of pupils in the sample were high enough to provide reliable estimates. Pupils with and without EAL did not differ greatly in their confidence in reading, on average, nor in the association between confidence and PIRLS performance.

Teacher and school characteristics

Teachers and headteachers of pupils who participated in PIRLS 2021 completed questionnaires related to their formal training and experience teaching, as well as questions related to the school climate. In England, pupils' teachers report having an average of 11.5 years teaching experience, lower than the international average of 18 years. Internationally, a higher proportion (56%) of pupils' teachers reported being very satisfied in their careers than in England (44%). There is no clear relationship between years of teaching experience or career satisfaction and pupils' PIRLS performance, in England or internationally.

Several factors related to school climate showed a strong relationship with pupils' overall achievement in PIRLS 2021. Pupils in schools where teachers report a very high emphasis on academic success achieve higher in PIRLS 2021. In England, pupils' average performance in PIRLS was also positively associated with the school's historic level of performance, with the average pupil at a high-performing school scoring approximately 32 points more than the average pupil at a low-performing school.

Pupils in schools where teachers report very high levels of safety and orderliness achieve higher PIRLS scores than those where lower levels of safety are reported. In England, 67% of pupils' teachers said that their schools were very safe and orderly; this is higher than the International Median of 59%. Headteachers' reports of school discipline also show an association with PIRLS 2021 achievement. Internationally, pupils in schools with hardly any problems with discipline score 26 points higher than those in schools with hardly any problems. In England this range is higher; pupils in schools with hardly any problems have an average PIRLS score 50 points higher than those in schools with schools with moderate to severe problems. A majority of pupils in England (76%) attend schools where headteachers report hardly any problems with school discipline; this is higher than the International Median of 62%.

The frequency of bullying experienced by pupils is also associated with average achievement in PIRLS. Generally, pupils who experience bullying more frequently at school have lower average achievement in PIRLS. In PIRLS 2021, the International Median score for pupils who experience bullying weekly is 53 points lower than those who almost never experience bullying. In England, 11% of pupils experience bullying weekly, and their average performance in PIRLS is 50 points lower than those who almost never experience bullying.

Teachers and headteachers also reported on levels of parental commitment and parental support of pupil achievement. Generally, pupils in schools where reports of parental commitment and support are high have higher average achievement in PIRLS 2021. In England, reports of parental commitment and support are lower than international levels.

Educational policy in England and PIRLS

Despite the complexity of drawing relationships between education policy and international large-scale assessments such as PIRLS, there are indications that, overall, the lessons that can be learned from PIRLS 2021 are positive. Evidence from both PIRLS 2016 and PIRLS 2021 suggests that the phonics screening check can be helpful in predicting later reading ability in primary school. Furthermore, the narrowing of the attainment gap in England suggests that efforts to improve the performance of lowerperforming pupils and schools are having a positive impact.

1 Introduction to PIRLS

Chapter overview

Chapter 1 provides an overview of PIRLS, with information on what PIRLS assesses, and who participates in PIRLS. It also briefly outlines England's past participation in PIRLS and discusses some of the main findings from previous cycles of the study about how well pupils in England could read. The chapter also includes some more technical discussions about the assessment design used in PIRLS and how the study was administered in schools, with a particular emphasis on the impacts of the COVID-19 pandemic on data collection in England as well as around the world. Next, the chapter moves on to discussing the scoring system used in PIRLS, and the main considerations that readers of this report should keep in mind when interpreting those scores. The chapter concludes by introducing the main comparators whose results are compared with England's, before outlining the content of the remaining chapters of this report.

1.1 What is PIRLS?

The Progress in International Reading Literacy Study (PIRLS) is a large-scale study that provides participating education systems with internationally comparable data on their pupils' performance and attitudes towards reading. PIRLS assesses pupils around the world who have received approximately 4 years of formal primary schooling and are typically around 10 years of age² (referred to as 'fourth grade' internationally). The International Association for the Evaluation of Educational Achievement (IEA), the organisation responsible for directing PIRLS, notes that PIRLS assesses pupils of this age group because they are typically transitioning from learning to read towards reading independently for further learning as well as for their own enjoyment (Mullis et al., 2023).

PIRLS was first conducted in 2001 and then every 5 years since. PIRLS 2021 is the fifth cycle of the study, although the data collection period for PIRLS 2021 was substantially longer than planned as a result of COVID-19-related complications (see section **1.2.3** for further details). The assessment design of PIRLS allows for the comparison of results across cycles, allowing education systems to track how their pupils' performance and

² In England participating pupils are in year 5, as they start formal schooling earlier than in some other education systems. The average age of pupils in the PIRLS 2021 England sample was 10.3 years. The international average across participating education systems that assessed the originally-planned age group was 10.2 years, and across these education systems average ages ranged from 9.8 to 10.9 years. In education systems that delayed assessment to the beginning of the next school year, the international average age was 10.8 and the average ages across education systems ranged from 10.1 to 11.3 (for more information on the effects of COVID-19 on assessment, including age groups assessed, see section **0**).

attitudes have changed over time as well as relative to other education systems. Many education systems, including England, use PIRLS data to inform and reflect on educational policies and practices in their own education systems, as well as internationally.

1.1.1 What does PIRLS assess?

PIRLS focuses on 3 different aspects of reading literacy – how pupils read different types of texts, what reading comprehension processes pupils use to understand those texts, and what attitudes pupils have towards reading. The first 2 aspects are assessed through a reading literacy test that bears many similarities to the types of comprehension tests that pupils in England sit at school, such as the key stage 2 reading test. The third aspect is assessed through a questionnaire completed by pupils after finishing their test. Questionnaires are also completed by participating pupils' teachers and headteachers, providing additional contextual information about their reading lessons and wider school environments. Most education systems also administer a questionnaire to the parents/guardians of these pupils for further contextual information on reading-related activities at home. However, the home questionnaire was not administered in England in PIRLS 2021.

Pupils participating in PIRLS answer questions on 2 different texts, each assessing one of the 2 'purposes' for reading. One of the texts is a short fictional story, and the questions assess how well the pupil can read texts with the purpose of providing readers with a 'literary experience'. This text always takes the form of a narrative fiction, rather than other literary forms such as poetry which can be difficult to translate across languages, or forms such as scripts of plays, which are not widely taught in many education systems. The other text is a short non-fiction piece that covers a relatively simple scientific/geographic/historical topic, e.g., "Green Sea Turtles", and assesses how well the pupil can read texts to 'acquire and use information'. Information can be displayed in textual formats but also across tables, charts, and other diagrams.

Some of the questions used in PIRLS are multiple-choice, and others require the pupil to provide a short, written response, though nothing more than a few sentences at most. Additionally, each question in the PIRLS assessment is also categorised as assessing one of 4 main reading comprehension processes. Ordered in increasing complexity, these 4 comprehension processes are:

 Focusing on, and retrieving explicitly stated information – pupils need to be able to locate and understand simple information in the text. Usually, the focus is on understanding words or sentences, and questions assessing this comprehension process might ask the pupil to identify the setting of a story, provide a definition of a term, or determine the main idea of the text, if it has already been clearly stated.

- Making straightforward inferences pupils need to be able to establish links between explicitly stated ideas to understand what else the text is communicating. This often relies on a more global understanding of the text, rather than simply understanding the text at the sentence level. Questions in PIRLS assessing this comprehension process might ask pupils to describe the relationships between characters, identify how one event in the story led to another, or conclude the main argument being made.
- Interpreting and integrating ideas and information pupils need to be able to incorporate their own knowledge of the world to form more complete understandings of text. They are also increasingly expected to combine information from different parts of the text to form a more comprehensive overview of the stories and/or arguments being made. Questions assessing this comprehension process might ask pupils to identify the real-world applications of the information presented, to consider the impacts of character's decisions in stories, or to identify the overall tone or mood of the text.
- Examining and evaluating content, language and textual elements pupils need to consider the merits and weaknesses of texts, including the strength of arguments presented, as well as be able to identify the author's perspective from the language and structure of texts. Questions assessing this comprehension process might ask pupils to explain why the author made certain language choices, or what the author's position is on more complex debates. Pupils may also be asked to provide opinions, backing them with reference to the text.

The IEA have published a series of videos showing examples of some of the texts and their questions used in the PIRLS 2021 assessment. "The Amazing Octopus" (<u>https://www.youtube.com/watch?v=5NM3ge6VPko</u>) is an example of an Informational Purpose text with an easy difficulty level, while "The Empty Pot" (<u>https://www.youtube.com/watch?v=eLWzWy1q3rl</u>) is an example of a Literary Purpose text with a medium difficulty level. Note that these videos focus on the digital administration of these texts and their questions, but pupils in England, and in many other participating education systems, administered the tests in paper-based formats. More information on the assessment design used in PIRLS 2021 is provided in section **1.2.1**.

1.1.2 Who participates in PIRLS?

A total of 57 education systems participated in PIRLS 2021, more than in any previous cycle. Table 1 summarises the education systems participating in PIRLS 2021 according to their first year of participation.

Year of first participation	Participating education systems
2001	Bulgaria, Cyprus, Czech Republic, England, France, Germany, Hong Kong SAR (China), Hungary, Iran, Israel, Italy, Latvia, Lithuania, Morocco, Netherlands, North Macedonia, Norway, Russia, Slovak Republic, Slovenia, Sweden, Türkiye, United States
2006	Austria, Belgium (Flemish), Belgium (French), Taiwan, Denmark, Georgia, Poland, Qatar, South Africa, Spain
2011	Australia, Azerbaijan, Croatia, Finland, Ireland, Malta, Northern Ireland, Oman, Portugal, Saudi Arabia, United Arab Emirates
2016	Bahrain, Egypt, Kazakhstan, Macao SAR (China)
2021	Albania, Brazil, Jordan, Kosovo, Montenegro, Serbia, Uzbekistan
	Source: IEA's PIRLS 2021

Table 1: Education systems participating in PIRLS 2001-2021

Additionally, PIRLS 2021 includes 8 benchmarking systems. These are typically territories within participating education systems, or cohorts that are older or younger than the main national samples. These include Moscow City in the Russian Federation, both Abu Dhabi and Dubai in the United Arab Emirates, a sixth-grade cohort in South Africa, and 4 Canadian provinces – Alberta, British Columbia, Newfoundland and Labrador, and Quebec.

1.1.3 England's historical participation in PIRLS

England has historically performed well in PIRLS. England's performance in PIRLS 2016 was significantly higher than it had been in both the 2006 and 2011 cycles, and higher (though not statistically significantly³ so) than in 2001. Only 7 education systems in PIRLS 2016 had average scores significantly higher than England's, with England performing well above the International Median score.

England's reading achievement in early cycles of PIRLS was marked by larger-thanaverage disparities between the highest-achieving and lowest-achieving pupils compared to some other education systems, particularly in high-performing European nations. Additionally, while female pupils outperform male pupils in the vast majority of education systems and cycles, the gender gap was historically larger in England than in most other education systems, particularly in 2011. These achievement gaps were smaller in the 2016 cycle, and the increase in overall score from 2011 was primarily driven by statistically significant improvements in the performance of lower-achieving pupils, particularly lower-achieving boys. The performance of girls and higher-achieving pupils remained relatively stable over this period.

³ Throughout this report, 'significantly' is used to mean 'statistically significantly at least at the 95% confidence level.

1.2 Interpreting data from PIRLS: a reader's guide

PIRLS, like similar international assessments such as PIRLS' sister-study, TIMSS (Trends in International Mathematics and Science Study), as well as the OECD's PISA (Programme for International Student Assessment), include highly technical components in the collection of pupil data, the design and administration of assessments, and the ways in which pupils' performance data is used to calculate estimates of their abilities. These differ from more conventional assessments that pupils sit in England, such as key stage 1 and key stage 2 assessments, where all pupils are administered the same test consisting of the same questions, and where average achievement is straightforwardly aggregated from individual achievement. This section briefly outlines the administrative and scoring procedures used in PIRLS. The PIRLS 2021 Methods and Procedures Report provides a more thorough discussion of these procedures, particularly with regards to the statistical procedures used to compare paper-based and digital assessment formats (Von Davier et al., 2023).

1.2.1 PIRLS assessment design

PIRLS 2021 utilised 18 separate assessment passages, with each pupil answering questions on one of the 9 'literary experience' passages, and one of the 9 'acquire and use information' passages. Within these breakdowns, 3 passages were considered 'Difficult', 3 passages were considered 'Medium Difficulty', and 3 passages were considered 'Easy'. While some of the passages are new, the majority of the passages have been used in previous PIRLS cycles and have not been publicly released. This approach of withholding and reusing test questions allows for the linking and calibrating of performance data from PIRLS 2021, including for performance on entirely new passages, to historical data, and allows scores to be calculated on the exact same scale as in previous PIRLS cycles (see section **1.2.4**). This method has been used across all cycles, and this means that despite there being no overlap with questions from the original PIRLS 2001 study, each cycle can link its data back to at least one prior cycle, forming a linkage chain that allows for valid comparisons to the original 2001 performance data.

The 18 assessment passages were distributed across 18 different booklets that paired one 'literary experience' passage with one 'acquire and use information'. Nine of these booklets were classified as 'More Difficult' and contained either 2 'Difficult' passages or one Medium and one Difficult passage. Conversely, the other 9 booklets were classified as 'Less Difficult' because they contained either 2 'Easy' passages or one Medium and one Easy passage. No booklets included one Difficult and one Easy passage.

The proportion of 'More Difficult' and 'Less Difficult' booklets administered in each education system varied based on historic performance in PIRLS. Education systems that have historically done well in PIRLS could choose to randomly assign 70% of the

participating pupils in their education system with a 'More Difficult' booklet, and 30% would receive a 'Less Difficult' booklet. Conversely, education systems which have typically performed less well in PIRLS could administer 70% of their pupils a 'Less Difficult' booklet. The purpose of offering education systems this approach is to ensure that the majority of pupils in that education system are sitting an assessment that is relatively well matched to their ability level. It is important to note here that the booklet design and overlapping use of 'Medium' difficulty passages allows for all pupils' performances to be calculated on the same scale (see section **1.2.4**). As England's performance in PIRLS 2016 was only slightly above the suggested threshold for administering a higher proportion of 'More Difficult' booklets, the decision was made to administer a 50:50 ratio of booklets for this cycle.

In England, all assessments and questionnaires were administered in traditional paperbased formats, as has been the case in all previous PIRLS cycles. However, for the first time, PIRLS 2021 provided participating education systems with the option to administer assessments digitally, with 24 education systems choosing to do so. Statistical procedures have been implemented by IEA to ensure that the data from education systems participating in paper-based and digital assessment formats can be compared on the same performance scale (Bezirhan, Foy & von Davier, 2023).

1.2.2 PIRLS in England: sampling and analysis

Unlike many national assessments in England which are focused on calculating the performance of individuals, the main goal of PIRLS is to understand reading performance at the national level. Given the burden of testing every year 5 pupil, in PIRLS a nationally representative sample of pupils is selected in each education system as is standard practice in international large-scale assessments. The IEA requires that each participating education system has a large enough sample of pupils to provide sufficient statistical power, and to help meet the requirement of national representativeness. However, certain sub-populations may still be over-represented or under-represented in the data after this sampling. To address this, 'weights' can be applied to the data at either the pupil, teacher, or school level, to modify how much that data point contributes to the national performance estimates. This method helps to correct for under- or over-representation in the sample.

When determining which primary schools in England would participate in PIRLS 2021, 2 main criteria were used to ensure that the chosen schools were as nationally representative as possible. Firstly, the proportion of different school-types (state-funded, independent schools, and academies) selected matched the proportion of these school-types nationally. Secondly, the selected schools were nationally representative of historical academic performance. This was done by using key stage 2 performance data in the 2018/2019 academic year and ensuring that there was an even distribution of

higher-performing and lower-performing state-schools, again in line with the national distribution of school performance. Additionally, any schools with fewer than 9 pupils in year 5, special schools, schools with alternative provisions, and Pupil Referral Units were all excluded in the school-level sampling. Statistics Canada were responsible for the random selection of schools following these criteria. In total, 169 schools and approximately 4,500 pupils in England were selected for participation in PIRLS 2021. Seven of these schools did not participate in the study for a variety of reasons, leaving 162 schools and a total of 4,150 pupils in England's final PIRLS sample. Appendix A provides further methodological details relevant to PIRLS 2021 in England.

Table 2 shows the numbers and percentages of the pupils in England's PIRLS 2021 sample with respect to the gender, major ethnicity group⁴, EAL status and current/historic free school meal (FSM) eligibility status recorded within England's NPD. It also includes information on the number and percentage of pupils in England's PIRLS 2021 sample who attend independent schools, and consequently do not have data within the NPD. It should however be noted that there are also a small number of pupils who do not have complete data within the NPD that attend state-maintained schools or academies (see Appendix B for more information about missing NPD data in the PIRLS sample). The table reports the percentage of pupils with that characteristic both before and after the application of weights, and provides a comparison to the national statistics available for the 2021/2022 year.

Pupil characteristic	Number of pupils in PIRLS sample (N)	Unweighted % of sample	Weighted % of sample	National figures (%)
Gender – Female	2135	51.5	51.4	* 48.8
Gender – Male	2008	48.5	48.6	* 51.2
Major ethnicity – White	2682	71.7	75.2	** 72.6
Major ethnicity – Black	203	5.4	4.9	** 5.9
Major ethnicity – Asian	551	14.7	12.2	** 12.6
Major ethnicity – Mixed	230	6.2	5.9	** 6.7
Major ethnicity – Other	73	2.0	1.9	** 2.2

Table 2: England's PIRLS 2021 sample characteristics relative to national data

⁴ Please note that the categories given here for gender and ethnic group reflect what is recorded in the National Pupil Database (NPD). These categories are in keeping with commonly-used categorisations in UK administrative data, and reflect the information that was available and appropriate to analyse (e.g. not breaking ethnic group into smaller categories that might be disclosive or provide unreliable estimates due to very small sample size), they do not reflect the views of the authors regarding how gender and ethnic identities can or should be represented.

Pupil characteristic	Number of pupils in PIRLS sample (N)	Unweighted % of sample	Weighted % of sample	National figures (%)
Studies English as an Additional Language (EAL)	856	22.7	20.0	** 19.5
Eligible for free school meals (FSM)	919	24.3	24.0	* 26.0
Has been eligible for free- school-meals at some point in the past 6 years (Ever6FSM)	1005	26.6	26.4	* 29.1
Attends an Independent school	162	3.9	4.3	** 6.5

Figures may appear inconsistent due to rounding and/or because of missing data.

* National figures based on 2021/2022 school census data for year 5 pupils.

** National figures based on 2021/2022 school census data for all year groups. Specific data for year 5 pupils not available.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Table 2 shows that, relative to national figures, England's PIRLS 2021 cohort was highly representative, with most percentages after the application of weights being roughly in line with national averages. Both before and after the weighting of the sample, England's PIRLS 2021 cohort had a slightly lower proportion of pupils who were eligible for FSM than the national average, and slightly fewer pupils coming from independent schools. England's sample for PIRLS 2021 also had a slightly higher proportion of pupils in the 'White' ethnic group than the national average, after the application of weights, as well as a slight over-representation of girls in the sample.

1.2.3 Impacts of COVID-19 on the administration of PIRLS 2021

The disruption to schools that resulted from the COVID-19 pandemic affected the collection of data for the PIRLS 2021 project internationally. Many schools closed for different periods of time during lockdowns or changed how frequently in-person instruction took place at school. This meant that data collection for PIRLS 2021 could not be conducted in some education systems as originally planned. **Table 3** provides a summary of the data collection periods for each education system in PIRLS 2021.

Typically, data collection for a PIRLS cycle takes place between October and November for education systems in the southern hemisphere, and between March and June for education systems in the northern hemisphere. In this way, pupils in their fourth year of

schooling are assessed towards the end of the school year in each hemisphere. For PIRLS 2021, 37 education systems were able to participate in the original timeframe and assessed their fourth-grade cohort at the end of the school year. However, COVID-19 related disruptions led 14 education systems to delay their participation by approximately 6 months meaning they assessed the cohorts at the beginning of the fifth grade, between August and December 2021. Throughout this report, we refer to these education systems as the 'delayed assessment' education systems, and in an approach that is consistent with the PIRLS 2021 International Report (Mullis et al., 2023), we generally report the results of these delayed assessment education systems separately to the other participating education systems. The main reason for this exclusion is because the participating pupils in these education systems were older that the target cohort for the study, and older than previous national samples for those education systems. The testing of older pupils therefore introduces challenges in reporting longitudinal trends in the results for these education systems, and raises questions about the validity of comparisons between the performance of older pupils and those roughly 6 months younger than them.

Where we indicate that results should be interpreted with caution, this often relates to the complications introduced by these complex circumstances due to the impact of the COVID-19 pandemic, as well as to instances in which the data may be less reliable, for example due to small numbers of pupils underlying a particular result. In these instances the results may suggest an indicative pattern at best.

Six education systems, including England, opted to collect their data a year later than their originally planned data collection periods. These education systems, herein described as the 'one year later' education systems, are included in the main comparison tables and figures in both this report as well as the International Report for PIRLS 2021, because the pupils assessed are of the target age of the study. The caveats of this approach are discussed in section **1.2.5**, as well as in relevant sections of this report.

The COVID-19 pandemic also affected the way in which schools were able to administer the PIRLS 2021 assessment. In previous cycles, external administrators visited all participating schools. To reduce the potential for the spread of COVID-19, schools in participating education systems were offered the option of administering the test using internal staff. In England, approximately two-thirds of schools opted to self-administer in this cycle, while the rest opted to invite a test administrator. If schools selected the self-administering option, teachers and/or school leaders attended remote training sessions to train as administrators for PIRLS 2021 for their schools. The quality of the administration of the PIRLS tests was evaluated by both a national and international quality control monitor, each of whom attended 10% of the data collection days in schools. Both self-administering and external-administering schools were included in these quality control evaluations.

Data collection period	Cohort	Participating education systems
Original assessment period	2020-21 fourth-grade cohort assessed at end of 2020- 21 school year	Albania, Austria, Azerbaijan, Belgium (French), Belgium (Flemish), Bulgaria, Cyprus, Czech Republic, Denmark, Egypt, Finland, France, Germany, Hong Kong SAR (China), Italy, Jordan, Kosovo, Macao SAR (China), Malta, Montenegro, Netherlands, New Zealand, North Macedonia, Norway, Oman, Poland, Portugal, Russia, Serbia, Singapore, Slovak Republic, Slovenia, Spain, Sweden, Taiwan, Türkiye, and Uzbekistan. Benchmarking participants Alberta, Canada British Columbia, Canada Newfoundland & Labrador, Canada Moscow City, Russian Federation
Delayed assessment (beginning of fifth grade)	2020-21 fourth-grade cohort assessed at beginning 2021-22 school year	Bahrain, Croatia, Georgia, Hungary, Ireland, Kazakhstan, Latvia, Lithuania, Morocco, Northern Ireland, Qatar, Saudi Arabia, United Arab Emirates, United States. Benchmarking participants Quebec, Canada Abu Dhabi, UAE Dubai, UAE
Assessed one year later	2021-22 fourth-grade cohort assessed at end of 2021- 22 school year	Australia, Brazil, England, Iran, Israel, South Africa Benchmarking participants South Africa (6 th grade)

Table 3: Comparison of data collection periods in PIRLS 2021

Source: IEA's PIRLS 2021

Despite the numerous challenges associated with the COVID-19 pandemic, 162 schools in England participated in this cycle and recruitment rates were not substantially affected. Both internationally and nationally, every effort was made to ensure that the quality and comparability of PIRLS data was maintained throughout the 2021 cycle.

1.2.4 PIRLS scale

The PIRLS scale was developed to score education systems that participated in the original PIRLS study in 2001 and set a score of 500 to reflect the mean-level performance of these education systems, with a standard deviation of 100 points. This

scaling has been maintained across cycles to allow for the calculations of trends in performance. As discussed in section **1.2.1**, the assessment design of PIRLS involves the retention of some assessment passages and their associated test-items across cycles. This, combined with the booklet structure that combines old passages with new passages, allows for every pupil's performance, regardless of the exact combination of questions they answered, to be equated on the PIRLS scale.

PIRLS also provides 4 International Benchmarks (Low, Intermediate, High, and Advanced), set at corresponding scores on the PIRLS scale, to describe the different reading comprehension skills and competencies associated with those scores. These benchmarks are outlined in greater detail in section **2.3**.

The IEA calculates individual pupils' PIRLS scale scores by employing approaches from a branch of statistics known as 'Item Response Theory'. This includes a method that looks at a pupil's pattern of correct and incorrect responses on each of the questions that they answered to estimate a probability that the pupil would get each of the questions in the other 16 assessment passages correct. This method, known as 'imputation' is used to produce a collective body of both real and estimated performance data for every question used in PIRLS 2021, and from that, an overall score known as a 'plausible value' is calculated. This approach is repeated for each pupil 5 times, which produces slightly different patterns of correct and incorrect responses to the imputed test questions, and as a result, this method calculates 5 different plausible values for each pupils' overall performance.

This approach is argued to produce much more accurate estimates of population-level performance than assessments where examinees only provide responses to a relatively small set of questions. However, this approach is not designed to produce accurate estimates of the performance of individual pupils. This also means that estimates of the performance of small sub-populations can also have high levels of statistical error.

1.2.5 Factors to consider when interpreting PIRLS data

The complex nature of PIRLS always introduces caveats when interpreting the data. The additional considerations raised by delays in data collection arising from the COVID-19 pandemic, as well as changes in how the PIRLS assessments were administered in some education systems, mean there are additional complications when interpreting the results. Some of the main factors to consider when interpreting the results in this report are as follows:

• Changes in score cannot be directly attributed to the COVID-19 pandemic. While it is reasonable to assume that the COVID-19 pandemic impacted schools and pupils, and may have had specific impacts on pupils' reading education, it is important to note that any significant changes in PIRLS scores cannot be causally attributed to this, a specific reading policy or practice, or any other factor.

- England's data was collected 12 months after most other education systems. Consistent with the approach used in the International Report, England's data is compared to that of other participating education systems, despite England's data being collected 12 months later than most other education systems. Every education system's data was collected during periods of substantial educational disruption, but it should not be assumed that this disruption was equal across education systems, or across time. England opted to collect data 12 months later than originally planned because of this disruption and the additional burdens that would be placed on schools had they collected data at the originally planned time. We cannot determine how England may have performed had they collected their data at the originally planned time, nor how other education systems may have performed had they also opted to wait 12 months. As such, while England's data is presented with and compared against that of other education systems, the different timing of data collection as well as differences across contexts in the extent of disruption caused by the COVID-19 pandemic should be remembered when interpreting findings and trends in results between PIRLS 2021 and those of previous PIRLS cycles.
- Comparisons of England's scores with international mean results may not be as informative as comparisons with the International Median. Consistent with the approach used in England's National Report for PIRLS 2016 (McGrane et al., 2017), we have opted to refer to the International Median as opposed to the approach used in the IEA's International Report, which typically focuses on mean results and the PIRLS Scale Centrepoint, the mean result of participating education systems in PIRLS 2001. This is because the distribution of performance scores in PIRLS are not symmetrical around the mean / PIRLS Scale Centrepoint, primarily due to a number of education systems performing at the lower end of the PIRLS scale that have a disproportionately negative effect on the mean. The International Median score is less affected by the skew produced by these education systems and is more representative of the average performance of education systems that are similar to England. When determining the International Median, the 14 delayed assessment education systems (as reported in **Table 3**) as well as all benchmarking participants, are not included.
- It is important to consider sources of error and statistical significance when comparing 2 or more estimates. The term 'significant' is used throughout this report to indicate where the difference between 2 results represents a statistically significant difference. PIRLS calculates estimates of performance, and these estimates have a level of uncertainty in them that we refer to as 'error'. This includes many different potential sources of error. For example, as described in

section **1.2.4**, there is a lot of uncertainty in individual pupils' plausible value estimates, and this is a potential source of 'measurement error'. Another source of error can be in the sampling of schools – it is possible that the randomly selected pupils are not actually representative of the education system's population of pupils, and this might mean that the estimate of the education system's score is higher or lower than what the 'real' score for that education system would be if every pupil had been assessed – this can be considered 'sampling error'. When the difference between 2 results is reported to be statistically significant, it is accounting for the estimated level of error in those results, and these error estimates can be quite large when looking at smaller groups. All statistically significant results in this report are significant at the 95% confidence level.

- Small sub-populations of pupils may be over or under-represented in the data. Section 1.2.2 outlined England's sampling design for PIRLS 2021. However, it is important to remember that small groups within the population, particularly relating to different major ethnicity groups as reported in the NPD, can still be over or under-represented even after using appropriate statistical methods to account for the design. It is also important to remember that the procedures for imputing data (discussed in section 1.2.4) are designed to produce accurate estimates of performance of large groups (e.g., education systems' overall results), but are not well suited for estimating the performance of individuals or small groups (e.g., by ethnicity).
- Passages and questions in the assessments may not be equally difficult for pupils from different socio-cultural or language backgrounds. While the IEA conduct thorough piloting of potential test-items to ensure that questions are fair and appropriate for pupils from different cultural backgrounds, some critics of PIRLS have suggested that many of the test questions are not equally challenging for pupils across different education systems and languages, even when these pupils are estimated to have very similar levels of overall reading comprehension (e.g., Grisay, Gonzalez & Monseur, 2009; Sandilands, Oliveri, Zumbo & Ercikan, 2013; Goodrich & Ercikan, 2019). These concerns can be further exacerbated by the fact that many pupils sit translated versions of these test-items, and these can make questions inadvertently more or less difficult than they are in English. It is therefore advised that readers adopt a healthy level of caution in comparing across different curricula, cultures and languages. Appendix C provides details about the language of the test in all participating education systems for PIRLS 2021.
- Trend results across PIRLS cycles assume that the difficulty of questions remains the same over time. We have previously discussed the use of passages from previous cycles that allows the results from PIRLS 2021 to be 'equated', which allows for comparability with results from all previous cycles. However, the

statistical methods that enable scores from every cycle to be compared relies on the assumption that the questions that are re-used (otherwise known as the 'common items') remain equally difficult across time. For similar reasons to the point above about relative difficulty across socio-cultural and language backgrounds, it may be that the content of passages becomes more or less accessible or familiar over time and this affects how pupils are able to engage with the questions about them. While the impact of this is probably small, this potential issue should be kept in mind when drawing comparisons over time.

1.2.6 Comparator selection

This report focuses on England's performance in PIRLS 2021 and compares this to England's performance in previous PIRLS cycles, as well as to other education systems around the world. A group of 'comparators' has been selected as the main focus of these international comparisons and is referred to throughout the report. The 4 main considerations for selecting comparator systems were as follows:

- Administered PIRLS to a similarly aged-cohort: as discussed in section 1.2.3, the COVID-19 pandemic caused a number of delays in the collection of data for PIRLS 2021, and this resulted in some education systems opting to assess pupils in the beginning of the fifth grade instead. These delayed assessment education systems have therefore assessed pupils who are approximately 6 months older than planned. Following the approach used in the PIRLS 2021 International Report, this report will not focus on direct comparisons between England and these education systems.
- English is the first language for most people in the education system: These education systems completed identical tests to England and tend to be culturally similar to England. However, many of these education systems, including Ireland, Northern Ireland, and the United States, are among the 14 delayed assessment education systems, and are consequently not included because they tested older cohorts.
- Among the top-performing education systems in PIRLS 2021: We include a selection of high-performing education systems on the basis that there may be lessons learned through these comparisons.
- Has participated in at least 2 other comparable PIRLS cycles: Being able to compare trends over time was a priority in the selection of focal comparators, so only those education systems that had participated in PIRLS for at least 2 previous comparable cycles (2011, 2016) were included.

With these considerations in mind, 3 main comparators were selected for use throughout the report, though comparisons to other education systems' results may be made where appropriate.

- **Singapore:** has been a consistently high performer in PIRLS since the 2006 cycle, and administers the study in English. Singapore was the top-performing education system in PIRLS 2011, and has improved on their overall performance in every successive cycle of the study. Singapore has also had a wide distribution of performance, similar to England.
- Hong Kong SAR (China): has also been selected because of its history of strong performance in PIRLS, including being the top-performing education system in PIRLS 2006. Hong Kong administers the test in Chinese, and compared to Singapore, has had a narrower range of performance, with fewer pupils at the very top of the end of the PIRLS scale, but also fewer pupils at the lower end.
- Australia: has been selected because of its cultural and linguistic similarities to England. Australia joined PIRLS in 2011 and had a statistically significant increase in performance between 2011 and 2016. Australia performed significantly lower than England in PIRLS 2016, but higher than the International Median score and significantly higher than New Zealand, the other most culturally and linguistically similar education system to England with comparable data. Australia also collected data on a similar timeline to England for PIRLS 2021 (i.e., both delayed data collection by 12 months due to the impact of the global COVID-19 pandemic).

1.3 Overview of the report structure

The rest of this report is divided into 9 main chapters. Chapters 2-8 focus on performance in PIRLS 2021 with respect to a variety of factors of interest. Chapters 9 and 10 provide deeper discussion of the changing educational context in England since the last PIRLS cycle with respect to the COVID-19 pandemic and educational policy more broadly.

The chapter-by-chapter summary is as follows:

- **Chapter 2** focuses **on overall reading performance**. This includes a comparison of performance across every education system that participated in PIRLS 2021. The chapter then assesses England's performance relative to the results in previous cycles as well as to the selected comparator systems, before discussing England's performance with respect to the 4 International Benchmarks.
- Chapter 3 looks at performance on the different subscales for reading purposes and comprehension processes. First, it presents performance in PIRLS 2021 on the Literary and Informational Purpose Scales, and on the 2 main Comprehension Scales – Retrieving and Straightforward Inferencing, and Interpreting, Integrating and Evaluating. England's performance on these scales is assessed in relation to those of the comparators and to England's previous performance on these scales.
- **Chapter 4** examines the **performance of higher- and lower-performing pupils**. Comparisons are drawn between the highest-performing (90th percentile) and

lowest-performing (10th percentile) pupils in England and in comparator systems, and with a particular focus on how any changes in England's overall performance since 2016 is reflected in the results for these higher- and lower-performing groups.

- Chapter 5 conducts a deeper dive into England's performance in PIRLS 2021 using prior attainment data. In particular, this chapter focuses on relationships between pupils' performance in PIRLS and their previous performances in their year 1 and year 2 phonics screening checks, as well as in how their reading was assessed by their teachers at the end of key stage 1.
- Chapter 6 focuses on performance by pupils' characteristics, including both analyses using data from the PIRLS 2021 pupil questionnaires, and for pupils in England, using data from the NPD. The chapter identifies whether and which pupil background characteristics (e.g., gender, ethnicity, EAL-status, free school meal eligibility) are independently and significantly associated with PIRLS performance. Additional contextual factors are also considered in this analysis, including the type of school a pupil attends, home possessions, and access to different resources at home (from PIRLS pupil questionnaire data). Each significant factor is then examined more closely.
- Chapter 7 focuses on pupils' attitudes and motivations towards reading, with a particular emphasis on their self-perceived engagement in reading lessons as well as their confidence in and enjoyment of reading. The chapter also explores how these relate to performance in PIRLS, including breakdowns by pupil gender and for higher-performing and lower-performing pupils.
- Chapter 8 looks at how characteristics of schools, teachers, and their teaching practices relate to PIRLS performance. The chapter covers a wide range of topics based on data from teacher and headteacher questionnaires, including teachers' qualifications and career satisfaction as well as schools' emphasis on academic success, safety and discipline. The chapter concludes with a look at both teachers' and headteachers' perceptions of parental involvement in the school and in their child's reading education.
- Chapter 9 provides a deeper discussion of the impacts of the COVID-19 pandemic, both on reading education within primary schools in England and on the collection of data for PIRLS 2021. The chapter also looks at questionnaire data from headteachers on how their schools operated during the pandemic. Finally, the chapter discusses global trends in performance in PIRLS 2021 with respect to the previous trajectories of results and how these differ from what might have been expected prior to the pandemic.
- **Chapter 10** concludes with a discussion of **educational policy in England**, including how specific knowledge and skill topics within the key stage 1 and key

stage 2 curricula map on to the PIRLS Benchmarks. The chapter also discusses national assessments of reading in England, including the year 1 phonics screening check, reading initiatives in England, and policies aimed at reducing attainment gaps between pupils from advantaged and disadvantaged backgrounds as well as between boys and girls.

2 Reading performance

Chapter overview

Chapter 2 provides an overview of the overall reading performance for pupils in PIRLS 2021. The chapter begins with a description of England's overall performance in PIRLS 2021, then compares England's overall performance with that of comparator systems. It discusses changes in England's reading performance over time, including details of performance trends for both England and the comparator systems across the PIRLS cycles for which data are available. The chapter concludes with a discussion of England's performance – and that of comparator systems – relative to the 4 International Benchmarks.

Key findings

- England's average score of 558 is significantly above the International Median of 520. This score is not significantly different to England's scores in most previous PIRLS cycles, including PIRLS 2016. By contrast, most education systems in PIRLS 2021 experienced significant drops in overall performance since 2016, but this may reflect the fact that many systems collected data for PIRLS during the COVID-19 pandemic.
- There have been no statistically significant changes since 2016 in the percentages of pupils in England performing at each of the International Benchmarks.

2.1 Overall reading performance

In PIRLS 2021, England's overall score is 558. This compares to an International Median score of 520. This International Median score was calculated as the median score of the 43 education systems that tested similarly-aged pupils to previous cycles, but does not include the 14 'delayed assessment' education systems that tested pupils that were roughly 6 months older than the pupils they would normally assess. A more detailed discussion of the delayed assessment systems and why their results are interpreted separately can be found in section **1.2.3**. England's score of 558 is significantly higher than the International Median score, and significantly higher than 39 of the 42 other systems that assessed pupils at the same age. The remaining 3 education systems all scored significantly higher than England.

Table 4 shows the performance of every education system relative to England, brokendown into 3 groups:

- Education systems that scored significantly higher than England in PIRLS 2021.
- Education systems that scored similarly to or higher than England in the previous cycle of PIRLS, but significantly lower than England in PIRLS 2021.
- Every other education system that scored significantly below England but above the International Median score of 520.

Table 4: Performance of education systems in PIRLS 2021 relative to England

Performance	Education system
Secred significantly higher than	Singapore (587)
Scored significantly higher than England in PIRLS 2021	Hong Kong SAR (573)
	Russia (567)
	Finland (549)
	Poland (549)
Scored similarly or significantly higher than England in PIRLS 2016 but	Taiwan (544)
significantly lower in PIRLS 2021	Sweden (544)
	Bulgaria (540)
	Norway (539)
	Australia (540)
	Czech Republic (540)
	Denmark (539)
	Italy (537)
	Macao SAR (536)
	Austria (530)
Scored significantly lower than	Slovak Republic (529)
England in PIRLS 2021	Netherlands (527)
	Germany (524)
	New Zealand (521)
	Spain (521)
	Portugal (520)
	Slovenia (520)
	+20 other education systems below the International Median of 520

Overall scores in parentheses.

Source: IEA's PIRLS 2021

It is important to reiterate that England's data was collected 12 months later than originally planned, later than most participating systems (see **Table 3** in section **1.2.3**) for more details on the effect of the COVID-19 pandemic on PIRLS 2021 data collection. While PIRLS cannot attribute increases or drops in score to any individual factor(s), it is important to consider that most systems collected their data during periods of substantial education disruption caused by the COVID-19 pandemic, and that comparisons between England and systems that collected their data earlier must therefore be interpreted with caution.

England did test pupils of the same age as originally planned (an average age of 10.3 years), and consequently the PIRLS 2021 International Report includes England's results in the main tables of comparisons with other systems that did the same (Mullis et al., 2023). For consistency with the International Report, this National Report for England includes such comparisons, but we strongly encourage readers to bear the differences in data collection periods in mind when interpreting any differences in performance between England and other education systems in PIRLS 2021.

Singapore was the highest performing system in PIRLS 2021 with an average score of 587. This is significantly higher than all other participating systems. Hong Kong and Russia both scored significantly higher than England, but not significantly differently to one another. While trend results are elaborated on in greater detail in section **2.2**, Finland and Poland (both of which collected data when originally planned) both scored significantly higher than England in PIRLS 2016 but significantly lower in PIRLS 2021. Similarly, Taiwan, Sweden, Bulgaria and Norway (all of which collected data according to the original timeline) did not score significantly differently to England in PIRLS 2016, but scored significantly lower than England in PIRLS 2021. Every other system that participated in PIRLS 2021 scored significantly lower than England.

As discussed previously, 14 education systems opted to delay the assessment of pupils in their schools by approximately 6 months because of challenges to data collection brought on by the COVID-19 pandemic. In each of these education systems, the pupils included in the assessment were therefore older than the usual samples for those systems in PIRLS, and this has raised challenges in comparing across systems, as well as longitudinally within those systems. Consistent with the approach used in the PIRLS 2021 International Report, the results of these education systems are not included in the calculation of International Medians (Mullis et al., 2023), and we typically report the results of these education systems separately. **Table 5** shows the performance of these education systems relative to England, but it is important to note that these comparisons do not report whether the differences in scores are statistically significant, because direct statistical comparisons would not be appropriate considering the complexity of differences in circumstances between delayed-assessment systems and England: Different age groups, different timing of data collection, and potentially different impact of the COVID-19 pandemic on normal school operations which cannot be measured and accounted for statistically.

Table 5: Performance of delayed-assessment education systems in PIRLS 2021relative to England

Performance	Education system
Education systems with scores higher than England in PIRLS 2021	Ireland (577) Northern Ireland (566)
Education systems with lower scores than England in PIRLS 2021	Croatia (557) Lithuania (552) United States (548) Hungary (539) Latvia (528) +7 other education systems below the International Median of 520

Statistical significance is not reported for education systems with delayed assessment. Overall scores in parentheses.

Source: IEA's PIRLS 2021

Table 6 shows the performance of the 15 highest-scoring education systems in PIRLS 2021, including the higher-performing systems with delayed assessment. For contextual information, the average age of the participating pupils in each system are listed.

Table 6: Overall scores of top 15 performing systems in PIRLS 2021, includingsystems with delayed assessment

Education system	PIRLS Score	Average pupil age	Data collection period
Singapore	587	10.4	Original
Ireland	577	11.0	Delayed
Hong Kong SAR	573	10.1	Original
Russia	567	10.8	Original
Northern Ireland	566	10.8	Delayed
England	558	10.3	One year later
Croatia	557	11.2	Delayed
Lithuania	552	11.3	Delayed
Finland	549	10.8	Original
Poland	549	10.9	Original
United States	548	10.7	Delayed
Taiwan	544	10.1	Original
Sweden	544	10.7	Original
Australia	540	10.0	One year later
Bulgaria	540	10.7	Original

Original = assessed pupils at the originally planned time. **Delayed** = delayed assessment by approximately 6 months (tested older pupils). **One year later** = assessed pupils approximately one year later than planned (same aged pupils assessed as originally planned).

Source: IEA's PIRLS 2021

2.2 Trends in overall reading performance

England's score of 558 in PIRLS 2021 is not significantly different from England's score of 559 in PIRLS 2016. However, most education systems that participated in both PIRLS 2016 and PIRLS 2021 have experienced statistically significant drops in overall reading performance. This is reflected in a 19 point reduction in the International Median score from PIRLS 2016 to PIRLS 2021.

Table 7 summarises each system's change in performance in PIRLS 2021 relative to 2016. Only 3 systems – Egypt, Oman, and the highest performing system in PIRLS 2021, Singapore – have seen a statistically significant increase in performance since the previous cycle. By contrast, 21 systems experienced statistically significant drops in performance since 2016. England was one of 8 education systems without any statistically significant changes in overall performance.

Table 7: Statistically significant changes in overall PIRLS score from 2016	to 2021
---	---------

Performance	Education system
Scored significantly higher in PIRLS 2021 than in PIRLS 2016	Egypt (+48)
	Oman (+11)
	Singapore (+11)
No statistically significant change between PIRLS 2016 and PIRLS 2021	Australia
	Belgium (French)
	Czech Republic
	England
	France
	Hong Kong SAR
	New Zealand
	Slovak Republic
Scored significantly lower in PIRLS 2021 than in PIRLS 2016	Austria (-11)
	Azerbaijan (-32)
	Belgium (Flemish) (-14)
	Bulgaria (-12)
	Denmark (-8)
	Finland (-17)
	Germany (-13)
	Iran (-15)
	Israel (-20)
	Italy (-11)
	Macao SAR (-10)
	Netherlands (-18)
	Norway (-20)
	Poland (-16)
	Portugal (-8)
	Russia (-14)
	Slovenia (-23)
	South Africa (-31)
	Spain (-7)
	Sweden (-12)
	Taiwan (-15)

Significant changes in score from PIRLS 2016 to PIRLS 2021 shown in parentheses.

Source: IEA's PIRLS 2021

Table 8 outlines the statistically significant changes in overall performance of thirteen of the fourteen systems with delayed assessment. Far fewer of these education systems saw statistically significant drops in performance, though it should be reiterated that these systems tested older pupil cohorts than usual, and as such, more caution should be applied when interpreting longitudinal trends. Croatia, the remaining education system
with delayed assessment, did not participate in PIRLS 2016. Croatia's overall score, however, was not significantly different to their score from PIRLS 2011.

Performance	Education system
	Bahrain (+12)
Scored significantly higher in PIRLS 2021 than in PIRLS 2016	Ireland (+11)
	Morocco (+15)
	Qatar (+42)
	Saudi Arabia (+18)
	United Arab Emirates (+33)
	Georgia
No statistically significant change	Lithuania
between PIRLS 2016 and PIRLS 2021	Northern Ireland
	United States
	Hungary (-15)
Scored significantly lower in PIRLS 2021 than in PIRLS 2016	Kazakhstan (-32)
	Latvia (-30)

Table 8: Statistically significant changes in overall PIRLS score from PIRLS 2016 toPIRLS 2021 for education systems with delayed assessment

Source: IEA's PIRLS 2021

Comparing across all 5 cycles of PIRLS, England's overall score of 558 in PIRLS 2021 is significantly higher than England's score of 539 in PIRLS 2006, but not significantly different from performance in the 2001, 2011, and 2016 cycles. This means that any differences in England's overall performance scores between PIRLS 2021 and these 3 cycles should be interpreted with caution. **Figure 1** shows England's trend in overall PIRLS performance across PIRLS cycles relative to the comparator systems (Singapore, Hong Kong, and Australia) and the International Median.

Both Singapore and Hong Kong scored similarly to the International Median in 2001, but experienced sharp increases in performance between 2001 and 2006. Singapore, the top-performing education system in PIRLS 2021, scored significantly higher than every other system, as well as significantly higher than its own scores across all previous cycles. Hong Kong, the top performing education system in the 2011 cycle of the study, recorded their highest score to date in PIRLS 2021, but this does not represent a statistically significant improvement from 2011 or 2016.

Australia, who joined PIRLS in the 2011 cycle, saw statistically significant improvement from 2011 to 2016. The 4 point drop between PIRLS 2016 and PIRLS 2021 is not statistically significant. Note that Australia, like England, opted to collect their data 12 months later than originally planned for PIRLS 2021, and as such, the trend between the 2016 and 2021 cycles represents a 6 year trend rather than 5.



Figure 1: Performance of England and comparator systems across PIRLS cycles Mean Overall Reading Score

Asterisks (*) indicate that the score shown is significantly different to that system's score for PIRLS 2021. Source: IEA's PIRLS 2021

The International Median score in PIRLS 2021 is 520, lower than in any previous PIRLS cycle. The 14 education systems with delayed assessment were not included in this calculation. However, given that 7 of these systems scored above 520 and the other 7 scored below 520, their inclusion would not have changed the International Median score. However, it should be noted that all 7 of the systems that debuted in PIRLS 2021 (see section **1.1.2**) performed below the International Median, and this does have a small negative effect on the Median. Nonetheless, this alone does not account for the 19 point drop in the International Median, which reflects the overall trend of most systems scoring significantly lower in PIRLS 2021 than in PIRLS 2016.

2.3 Overall performance according to International Benchmarks

As discussed in section **1.2.4**, PIRLS uses 4 International Benchmarks to describe the reading skills associated with scores on the PIRLS scale. These benchmarks and their associated scores are outlined in **Table 9**.

Benchmark	Score	Description
Low International Benchmark	400	Pupils can locate and retrieve explicitly stated information, actions and ideas from text, charts and diagrams. They can also make straightforward inferences about events and reasons for actions.
Intermediate International Benchmark	475	Pupils can locate, recognise, and reproduce explicitly stated actions, events and feelings, as well as make straightforward inferences about the attributes, feelings, and motivations of main characters in stories. They can also begin to integrate and compare information about central ideas and events across the text, and make basic inferences about language choices.
High International Benchmark	550	Pupils can locate and distinguish significant actions and details embedded across the text, and make inferences to explain relationships between intentions, actions, events, and feelings. They can also connect ideas and sequence events across different elements of a text. Pupils will also be able to interpret and integrate story events and character traits and recognise figurative language features such as metaphor and imagery.
Advanced International Benchmark	625	Pupils can interpret and integrate story events and character actions to describe character's feelings, motivations, and character development, and evaluate the intended effects of author's language and style choices. They can also evaluate how the author's selection of visual elements such as charts and diagrams convey the author's point of view.

Table 9: Overview of PIRLS 2021 International Benchmark criteria and scale scores

Source: IEA's PIRLS 2021

Figure 2 shows the proportion of pupils in England and both comparator groups that performed at each International Benchmark in PIRLS 2021. More than half of pupils in England (57%) reached the High International Benchmark. This is consistent with England's overall score of 558, which is slightly above the High International Benchmark of 550. Although Australia has a lower percentage of pupils reaching each of the Benchmarks than England, this difference is most pronounced at the High Benchmark, met by 48% of pupils in Australia. Singapore has nearly twice the proportion of pupils reaching the Advanced Benchmark (35%) compared to England (18%), but in both England and Singapore the same proportion of pupils (97%) performed at a level above

the Low International Benchmark score of 400. Hong Kong has a lower percentage of pupils reaching the Advanced and High Benchmarks than Singapore, but also fewer pupils performing at levels below the Intermediate and Low Benchmarks, indicating that in Hong Kong there was a narrower range of performance overall.



Figure 2: Percentage of pupils in England and comparator systems reaching each International Benchmark in PIRLS 2021

Source: IEA's PIRLS 2021

Figure 3 shows England's performance at each of the International Benchmarks across all 5 cycles of PIRLS, in comparison to the International Median. Consistent with the international trend of lower overall performance in PIRLS 2021, internationally there has been a reduction in the percentage of pupils meeting each of the benchmarks, with the largest reductions at the High and Intermediate Benchmarks. By contrast, there has been relatively little change in England since 2016 in the proportions of pupils reaching the Low, Intermediate and High Benchmarks, though a slightly lower percentage of pupils in England reached the Advanced Benchmark than in PIRLS 2016.





Education system and cycle	Advanced	High	Intermediate	Low
England 2021	18%	57%	86%	97%
England 2016	20%	57%	86%	97%
England 2011	18%	54%	83%	95%
England 2006	15%	48%	78%	93%
England 2001	20%	54%	82%	94%
Int. Median 2021	7%	36%	75%	94%
Int. Median 2016	10%	47%	82%	96%
Int. Median 2011	8%	45%	80%	95%
Int. Median 2006	7%	43%	80%	95%
Int. Median 2001	7%	38%	75%	92%

Source: IEA's PIRLS 2021

2.4 Contextualisation: Average PIRLS performance in European education systems

European education systems typically perform well in large-scale international assessments such as PIRLS and PISA. It is common for European education systems to feature near the top of rankings in these assessments, and typically, they show positive progress across cycles. The most recent cycle of results from PISA are not yet available, however, when considering the performance of participating European education systems in PIRLS 2021, a decline from previous cycles can be seen. The International Median for PIRLS 2021 has also shown a decline from previous cycles. The potential reasons for this are manifold, and although the impact of the COVID-19 pandemic and school closures in Europe may have had an impact on PIRLS 2021 performance, the absence of a control group means it is impossible to determine the extent of this impact (Mullis et al., 2023). It is also important to reiterate that the differing timeframe and circumstances under which different education systems' data was collected means that direct comparisons should be made with caution, as should inferences about the implications of a general decline, which may relate to the impact of COVID-19 on data collection as well as school operations.

Despite this, some insights can be gained by looking into the performance of some of the participating education systems from Europe to understand variations in performance. **Box 2.1** considers the average performance of European education systems, with specific reference to 3 high-performing systems Finland, Sweden, and Bulgaria.

Box 2.1 PIRLS performance in Europe

The European Median score, which includes the results of 27 participating education systems from Europe, has been slightly higher than the overall International Median score in all five cycles, including in 2021. Between 2001 and 2016, the European median climbed steadily from 528 in 2001, to 547 in 2016. However, the European Median in 2021 was 524, a drop of more than 20 points from 2016. This has been primarily driven by the majority of European education systems experiencing large drops in overall average scores.

Consistent with the trend across most European participating systems, Finland, Sweden, and Bulgaria all scored significantly lower in PIRLS 2021 than in 2016. Finland scored significantly higher in 2011 (568) and 2016 (566) than in 2021 (549), a 17-point decline. Sweden, the top-performing education system in the original PIRLS 2001 cycle, and Bulgaria both performed significantly lower in 2021 than they did in 2001. Sweden showed a decline of 11 points between 2001 (561) and 2016 (555), and the declined a further 9 points between 2016 and 2021 (544). Bulgaria had their highest achievement in 2016 (552), a 20-point improvement from their 2011 score (532), however, in 2021 they declined again achieving an average overall PIRLS score of 540.

Achievement at International Benchmarks is consistently strong across Europe. All of the European education systems have much higher proportions of pupils meeting the Advanced and High Benchmarks than the International Median percentages. Of these 3 European education systems, Bulgaria has the highest proportion of pupils reaching the Advanced Benchmark, but also the highest proportion of pupils that perform below the Low Benchmark. By contrast, Finland has the lowest proportion meeting the Advanced Benchmark, but also the lowest proportion failing to meet the Low Benchmark.

Although there is large variation in performance across European education systems, consistently higher performance from European education systems when compared to international averages has also been seen on other large-scale assessments such as PISA (OECD, 2018; Volante & Klinger, 2021).

3 Performance in reading purpose and process scales

Chapter overview

Chapter 3 provides an overview of pupils' performance in on the reading purpose and process scales used in PIRLS. The chapter begins with an overview of pupils' performance in England and the comparator systems on the 2 reading purpose scales – the 'Literary Purpose' scale, and the 'Informational Purpose' scale. This is followed by an account of trends over time in England and the comparator systems on these scales, as well as how these education systems' relative strengths on these scales has (or has not) changed over time. A similar approach is used to present the performance of England and the comparator systems on the 2 reading process scales – 'Retrieving and Straightforward Inferencing', and 'Interpreting, Integrating and Evaluating'. The chapter first focuses on performance on these scales in 2021 before looking at trends across all five previous cycles of PIRLS, and concluding with a look at relative strengths on one of the 2 reading purpose scales across time.

Key findings

- Contrary to all previous PIRLS cycles, pupils in England performed equally well on the Literary and Informational Purpose Scales. England, and most other culturallysimilar English-speaking education systems have historically scored significantly higher on the Literary Scale.
- England's score on the Literary Purpose Scale has decreased slightly from 2016, while the score on the Informational Purpose Scale has slightly increased, though neither of these represent statistically significant changes.

England has continued to score significantly higher on the 'Interpreting, Integrating and Evaluating' Process Scale than the 'Retrieving and Straightforward Inferencing' Process Scale. This remains a common finding among culturally-similar Englishspeaking education systems.

3.1 Performance in reading purposes

As discussed in section **1.1.1**, each pupil that participates in PIRLS answers questions on 2 different texts. One of the texts is a short fiction piece that assesses how well pupils read texts with the purpose of providing readers with a literary experience. The other text is a short non-fiction piece with the purpose of allowing readers to 'acquire and use information'. Using pupils' responses to the so-called *Literary Purpose* and *Informational Purpose* test questions, PIRLS not only calculates overall scores of reading

comprehension, but also calculates separate scores specifically for *Literary Purpose* and *Informational Purpose* reading.

Table 10 shows the overall performance of England and the comparator systems relative to their performances on the individual Literary Purpose and Informational Purpose Scales. In PIRLS 2021, England scored similarly on the Literary Purpose and Informational Purpose Scales, with a score of 558 on the Literary Purpose Scale, and 559 on the Informational Purpose Scale.

Table 10: Performance of England and comparator systems on the Literary Purpose and Informational Purpose Scales relative to overall performance (2021)

Education system	Overall PIRLS Score	Literary Scale Score	Informational Scale Score	Difference
England	558	558	559	-1
Singapore	587	591	586	5
Hong Kong SAR	573	564	582	-18
Australia	540	543	539	4
International Median	520	520	520	0

Statistical significance is not reported for the magnitude of the differences in performance between scales. Difference is calculated as Literary Scale Score – Informational Scale Score. The International Median difference is calculated as a difference of the Literary and Informational Scale Score International Medians. Source: IEA's PIRLS 2021.

It is important to note that the scaling process takes into account question difficulty. This means that, for example, the pupils in an education system that scored significantly higher on the Informational Purpose Scale did not necessarily gain more marks on Informational Purpose questions than Literary Purpose questions, if over all participating education systems, pupils found the questions on the Literary Purpose texts easier. Rather, the pupils in that system were *relatively* strong on the Informational Purpose questions compared to the Literary Purpose questions, relative to how every participating system fared on these scales.

In the case of England, who scored a single point higher on the Informational Purpose Scale than the Literary Purpose Scale in 2021, it is important to remember that this does not necessarily mean that pupils in England scored more marks on Informational Purpose questions, but rather, pupils in England did not demonstrate much of a relative strength on either the Literary or Informational Purpose texts, relative to the pupils in other education systems.

England's score of 558 on the Literary Purpose Scale was significantly lower than Singapore (591) but not significantly different to Russia (566) or Hong Kong (564). England scored significantly higher than every other education system, other than Poland (552). England's score of 559 on the Informational Purpose Scale meanwhile was significantly lower than Singapore (586), Hong Kong (582), and Russia (568), and significantly higher than every other system.

There were no statistically significant differences between Hong Kong and Singapore's performances on the Informational Purpose Scale, or between Hong Kong and England's performances on the Literary Purpose Scale. Both Singapore and Australia had slightly higher scores on the Literary Purpose Scale than the Informational Purpose Scale, and there was no difference in the International Median scores of the 2 Purpose Scales.

Of the comparator systems, Hong Kong showed the greatest difference in performance across the 2 Purpose Scales, scoring 18 points higher on the Informational Purpose Scale than the Literary Purpose Scale. This trend was common across education systems that administered the test in Chinese, with Macao SAR (22 points higher) and Taiwan (16 points higher) joining Hong Kong as the 3 education systems in PIRLS 2021 with the largest differences in performance between the scales in favour of the Informational Purpose Scale.

There are also patterns in the relative strengths on these scales among culturally-similar English speaking systems when including the results from delayed assessment systems such as Ireland, Northern Ireland, and the United States. These 3 systems all scored higher on the Literary Purpose Scale than the Informational Purpose Scale, each having differences of more than 10 points. In the case of the United States, the difference of 18 points was greater than for any other system that participated in PIRLS 2021, and as is touched upon in section **3.1.1**, stronger performance on the Literary Scale has been typical of culturally similar English-speaking education systems in previous cycles of PIRLS. As was shown in **Table 10** however, this was not the case for England in PIRLS 2021.

3.1.1 Trends in performance on Literary and Informational texts

Figure 4 shows the performance of England and the comparator systems on the Literary Purpose Scale across previous PIRLS cycles, as well as in comparison to the International Median score. England's score of 558 in PIRLS 2021 is significantly higher than their score on the Literary Purpose Scale in 2006, but not significantly different to their scores on this scale in 2001, 2011, or 2016. The small differences in points between these cycles, including the 5 point drop from 2016 to 2021, should therefore be interpreted cautiously.

Singapore's score of 591 is significantly greater than their performance on the Literary Purpose Scale in all 4 previous cycles, and represents a 16 point increase from 2016. Hong Kong's score meanwhile is a significant improvement from 2001, but not significantly different to their score in the previous 3 cycles. Australia's performance is roughly in line with their performance on this scale in 2016, and a statistically significant improvement on their performance in 2011.

Figure 4: Performance of England and comparator systems on the reading for Literary Purpose Scale across PIRLS cycles



Mean Literary Scale Score

Education system	2001	2006	2011	2016	2021
England	561	*540	553	563	558
Singapore	*531	*554	*567	*575	591
Hong Kong SAR	*520	559	565	562	564
Australia	n/a	n/a	*527	547	543
International Median	522	532	533	540	520

Asterisks (*) indicate that the score shown is significantly different to that system's score for PIRLS 2021. Source: IEA's PIRLS 2021

Figure 5 shows the performance of England and the comparator systems on the Informational Purpose Scale across PIRLS cycles, and with respect to the International Median score in each cycle. England's score of 559 on the Informational Purpose Scale is higher than all 4 previous cycles of the study, and represents a statistically significant

improvement from the 2001, 2006, and 2011 cycles. The difference between the 2016 and 2021 scores on these scales does not represent a statistically significant improvement in score. Singapore and Hong Kong also both recorded their highest scores on this scale across all 5 cycles, but the improvements from 2016 are not statistically significant. Australia's score on this scale has decreased since 2016, but this change is also not statistically significant.

Figure 5: Performance of England and comparator systems on the reading for Informational Purpose Scale across PIRLS cycles



Mean Informational Scale Score

Asterisks (*) indicate that the score shown is significantly different to that system's score for PIRLS 2021. Source: IEA's PIRLS 2021

Figure 6 shows the differences in Literary Purpose and Informational Purpose Scale scores for England and the comparator systems across all 5 cycles. Statistically significant differences in performances across the 2 scales are also highlighted.





Differences calculated as Literary Scale Score – Informational Scale Score

-17

n/a

Hong Kong SAR

Australia

Source: IEA's PIRLS 2021

-18

4

-14

5

In every previous cycle of PIRLS, England has scored significantly higher on the Literary Purpose Scale than the Informational Purpose Scale. This has previously been observed as a relatively consistent finding across culturally-similar English-speaking education systems. However, this was not the case for England in PIRLS 2021, with a slightly higher

-11

n/a

-13

-1

score on the Informational Purpose Scale for the first time. It should be noted that the difference in performance on the 2 scales is not statistically significant, and neither are the individual changes in score on either Purpose Scale.

Singapore's results in 2021 have also bucked a historical trend of typically performing more strongly on the Informational Purpose Scale, with a score 5 points higher on the Literary Purpose Scale than the Informational Purpose in PIRLS 2021. This difference shows that pupils in Singapore performed significantly better on Literary Purpose texts than Informational Purpose texts in PIRLS 2021, relative to the performance on these types of texts in other participating education systems. Hong Kong has consistently scored significantly higher on the Informational Purpose Scale, while Australia has performed significantly better on the Literary Purpose Scale in the last 2 cycles of PIRLS.

3.2 Performance in reading comprehension processes

Together with 2 reading purposes, PIRLS also assesses 4 comprehension processes within the literary and informational passages:

- focus on and retrieve explicitly stated information;
- make straightforward inferences;
- interpret and integrate ideas and information; and
- evaluate and critique content and textual elements.

Each of the questions associated with the reading passages in PIRLS aim to assess one of these comprehension processes and the responses are then psychometrically modelled into scores for 2 process scales; the *Retrieving and Straightforward Inferencing* (RSI) process scale, and the *Interpreting, Integrating and Evaluating* (IIE) process scale.

In the PIRLS 2021 assessment half of the questions aim to assess comprehension processes related to the RSI Scale. The RSI questions are usually related to a small portion of text in the passages where 20% of the total questions in PIRLS 2021 ask pupils to 'focus on or retrieve explicitly stated information' from the relevant portion of text, and 30% require pupils to 'make straightforward inferences' from the information in the text. The other half of questions in PIRLS 2021 aim to assess comprehension processes related to IIE Scale. IIE questions usually relate to the entire passage or larger portions of the text in a passage where 30% of the questions ask pupils to 'interpret and integrate ideas and information', while 20% ask pupils to 'evaluate and critique content and textual elements'.

Table 11 shows the overall performance of England and the comparator systems inPIRLS 2021 relative to their performances on the RSI Scale and the IIE Scale.

Table 11: Performance of England and comparator systems on the reading comprehension process scales relative to overall performance (2021)

Education system	Overall PIRLS Score	RSI Scale Score	IIE Scale Score	Difference
England	558	554	561	-7
Singapore	587	584	591	-7
Hong Kong SAR	573	577	572	5
Australia	540	534	547	-13
International Median	520	520	520	0

Statistical significance is not reported for the magnitude of the differences in performance between scales. Difference is calculated as RSI – IIE Scale Score. The International Median difference is calculated as a difference of the RSI and IIE Scale Score International Medians.

Source: IEA's PIRLS 2021.

Pupils in England performed slightly above their overall PIRLS 2021 score on the IIE Scale (561) and had slightly lower performance on the RSI Scale (554). This means that pupils in England were relatively stronger on questions requiring higher-level interpretive and evaluative comprehension skills than they were on questions requiring more straightforward retrieval and inferencing, relative to the pupils in other education systems.

England was one of 15 education systems participating in PIRLS 2021 that showed strength on the IIE Scale relative to their overall average. Internationally, the trend of higher scores on the IIE Scale is common among English-speaking systems. Among the comparators, higher performance on the IIE Scale was also seen in Singapore and Australia who both had IIE scores higher than their RSI Scale score. Internationally, Australia was one of the education systems with the highest magnitude difference between their overall score and their scores on the individual process scale scores, with an RSI score 6 points lower than their overall average (540), and an IIE score 7 points higher.

Hong Kong, on the other hand, were among 10 other education systems that showed relative strength in the RSI Scale compared to their overall average. For Hong Kong, there was almost no difference between their overall average (572) and their score on the IIE Scale (573), however, their score on the RSI Scale was 5 points higher (577) than their overall average in PIRLS 2021.

3.2.1 Trends in performance on comprehension process scales

Figure 7 shows the performance of England and the comparator systems on the RSI Scale across previous PIRLS cycles, as well as in comparison to the International Median score on this scale.

Figure 7: Performance of England and comparator systems on the Retrieving and Straightforward Inferencing Process Scale across PIRLS cycles



Mean Retrieving and Straightforward Inferencing Scale Score

Asterisks (*) indicate that the score shown is significantly different to that system's score for PIRLS 2021. Source: IEA's PIRLS 2021

England's score on the RSI Scale in PIRLS 2021 is significantly higher than the score on the same process scale for 2006 and 2011, but not significantly different from their score in the more recent 2016 cycle. Singapore and Hong Kong both show an upward trend on the RSI Scale from 2001 to 2021. Singapore has shown an increase of 11 points from 2016 and has significantly higher performance in 2021 compared to all previous cycles. Hong Kong has shown a 9-point increase since 2016 and has significantly higher performance in 2021 than all previous cycles. Although England, Singapore and Hong Kong show either improvement or stability from 2016 to 2021, Australia and the International Median both show a downward trend across this same period. Australia's

performance on the RSI Scale in 2021 is significantly better than their performance in 2011, but significantly lower than their 2016 performance on this scale.

Figure 8 shows the performance of England and the comparator systems on the IIE Scale across previous PIRLS cycles, as well as in comparison to the International Median score on the IIE Scale. England's performance on the IIE Scale in 2021 is significantly higher than their score on this scale from 2006, but is not significantly different to their score in any other cycle. Similar to the trend seen on the RSI Scale, the International Median for the IIE Scale has decreased from 536 in 2016 to 520 in 2021.

Figure 8: Performance of England and comparator systems on the Interpreting, Integrating and Evaluating Process Scale across PIRLS cycles



Mean Interpreting, Integrating and Evaluating Scale Score

Education system	2001	2006	2011	2016	2021
England	556	*542	555	561	561
Singapore	*526	*557	*570	*579	591
Hong Kong SAR	*530	566	578	568	572
Australia	n/a	n/a	*529	549	547
International Median	526	537	532	536	520

Asterisks (*) indicate that the score shown is significantly different to that system's score for PIRLS 2021. Source: IEA's PIRLS 2021 Singapore's performance on the IIE Scale has shown an upward trend since 2001, with their performance on this scale in 2021 being significantly higher than all 4 previous cycles. Although the IIE Scale score for Hong Kong has shown slight changes from cycle to cycle, only the improvement from the original 2001 cycle has been statistically significant. Australia's IIE Scale score for 2021 is significantly better than their 2011 score, but unlike their performance on the RSI Scale, the decrease in score from 2016 on the IIE Scale is relatively small and does not represent a statistically significant drop.

Figure 9 shows the trends in the differences between RSI and IIE Scale scores for England and the comparators across all 5 cycles. Statistically significant differences in performances across the 2 process scales are also highlighted. Both England and Australia have scored significantly higher on the IIE Scale than the RSI scale in all of their previous participations in PIRLS. This is a common finding across English-speaking education systems, as well as the majority of high-performing European education systems participating in PIRLS (Mullis et al., 2023). Singapore's results show a change from significantly higher performance on the RSI Scale in 2001 and 2006 to significantly higher performance on the IIE Scale since 2011. Hong Kong, on the other hand, reversed their pattern from previous cycles in 2021, scoring significantly higher on the RSI Scale than the IIE Scale for the first time.



Figure 9: Differences in comprehension processes scales scores across PIRLS cycles for England and comparator systems

Education system	2001	2006	2011	2016	2021
England	-8	-6	-10	-6	-7
Singapore	8	6	-5	-6	-8
Hong Kong SAR	-6	-5	-16	-1	5
Australia	n/a	n/a	-2	-8	-13

Differences calculated as RSI Scale Score – IIE Scale Score

Source: IEA's PIRLS 2021

3.3 Contextualisation: Reading for Literary Purposes and curricula in Ireland and Northern Ireland

The extent of relative differences in performance on the Literary and Informational Purpose Scales varied across participating education systems; 23 systems, including England, did not perform substantially differently across the 2 reading purposes assessed by PIRLS 2021. A strength on the Informational Purpose Scale was seen in 3 East Asian systems; Macao, Hong Kong and Taiwan. A relative strength in reading for Literary Purpose was evident in the United States, Georgia and Croatia; as well as other English-speaking systems such as Northern Ireland and Ireland, and benchmarking regions of Canada. **Box 3.1** focuses on the strength in reading for Literary Purpose in Ireland and Northern Ireland and its relationship to reading curricula in both education systems.

Box 3.1 Northern Ireland and Ireland: Literary reading and national curricula

Northern Ireland and Ireland are among seven of the education systems with the highest relative strength in reading for Literary Purpose compared to Informational Purpose in PIRLS 2021. In Ireland, the average Literary Scale score was 6 points higher than their overall average in PIRLS 2021 (577); while in Northern Ireland the average Literary Scale score was 7 points higher than their overall average in PIRLS 2021 (566). In both cases, there was approximately 10 points difference between the 2 scale scores, and performance on the Informational Purpose Scale was 4 points below their overall average for PIRLS 2021. The same trend was evident in PIRLS 2016, where both education systems showed a significantly higher scale score for reading for Literary Purpose when compared to their overall average.

Higher relative performance on the Literary Scale is more commonly seen in education systems where the reading curriculum begins with a focus on stories in the early grades and transitions to focusing on reading to learn subject content in upper grades (Reynolds et al., 2022). Reading for specific purposes is central to the PIRLS reading framework, and similar areas of focus can be seen across the curricula of participating systems. The National Research Coordinators (NRCs) in both Ireland and Northern Ireland reported that there was a major emphasis on Reading for Literary Experience as well as Reading to Acquire Information in their respective national language/reading curricula.

The emphasis on using different types of texts in the primary school is highlighted in the new Primary Language Curriculum (PLC) for Ireland (NCCA, 2019). This curriculum supports teaching and learning in both English and Irish. One of the key aims of the PLC is to prioritise developing pupils' ability to explore and use language for a wide range of purposes. Teachers are encouraged to create engaging learning experiences for pupils using texts across a diverse range of purposes, genres and types.

Continues on next page

Box 3.1 (continued)

There are also specialist reading initiatives in Ireland that aim to foster reading for enjoyment such as the Drop Everything and Read (DEAR) programme, a popular reading initiative that is frequently implemented to promote silent reading, pupil choice and reading for enjoyment (Delaney et al., 2022).

In Northern Ireland, the curriculum for language/reading has been in place since 2007 (CCEA, 2007). Similar to the diversity of text type encouraged in Ireland, the domain of reading in the curriculum for Northern Ireland also encourages reading, exploring, using and understanding a wide range of texts. Pupils in Northern Ireland are also encouraged to engage in independent reading activities for enjoyment as well as to acquire information (Boyd, 2022).

The reading curricula in both Ireland and Northern Ireland promote diversity in the type of text used in the classroom and do not explicitly focus on reading for literary experiences above other reading purposes. Despite this, the performance of pupils from both education systems on the PIRLS Literary Purpose Scale is higher than the Informational Purpose Scale. Potential influences in the results across these scales are wide ranging and it is difficult to know precisely why pupils in Northern Ireland and Ireland perform better on literary texts.

Teachers of pupils from Northern Ireland who participated in PIRLS 2021 report using a wide range of text types for reading activities. However, there are differences between the reports of the frequency at which different text types are used in the classroom; 93% of pupils' teachers report using longer fiction books at least once a week, while only 38% report using longer non-fiction books at the same frequency. Internationally, an average of 39% of participating pupils' teachers report that they use longer fiction books at least once a week in reading lessons, and 26% report using longer non-fiction books at the same frequency at the same frequency. Teacher reports from Ireland are not available for pupils that participated in PIRLS 2021.

Classroom policy and practice are not the only factors likely to influence differences in reading achievement across purpose scales in these education systems. The broader influences on reading development both at home and in the early stages of language development are widely acknowledged, and differential exposure to more fiction materials both inside and outside of the classroom, as well as in home contexts, and variations in early literary experiences are all likely to impact pupil performance on PIRLS.

4 Reading performance of higher and lower performing pupils

Chapter overview

This chapter discusses the distribution of reading performance in England and comparator education systems, focusing on the spread between the lowest-scoring (10th percentile) and highest-scoring (90th percentile) pupils. First, the chapter gives an overview of the spread of performance (at the 10th, 50th, 90th percentiles) across England and comparator systems included elsewhere in this report. It then presents trends across PIRLS cycles at these different levels of performance as well as across subscales in England.

Key findings

- There has been a continued narrowing of the gap between the lowest-scoring and highest-scoring pupils in England.
- Considering the long-term trend, the narrowing of the gap between lowest- and highest-performing pupils is due to an increase in the scores of the lower achievers rather than a decrease in the scores of the higher achievers.
- England and comparators mostly showed fairly stable recent trends in the gap between the 10th and 90th percentiles (2016-2021) while the International Median showed a marked drop.

4.1 Distribution of reading performance

So far in this report, we have focused on mean-level performance within education systems, either in terms of overall performance (chapter 2), or on the Reading Purpose (section **3.1**) or Comprehension Process (section **3.2**) subscales. However, these averages do not provide much information regarding the ranges of performance within education systems.

In this chapter, we focus primarily on performance at the 10th and 90th percentiles. Scores at the 10th percentile represent the highest score achieved by pupils in the bottom 10% of pupils in that education system. Scores at the 90th percentile meanwhile represent the highest score achieved by pupils in the lowest 90% of pupils in that system. This can also be considered as the highest score achieved by a pupil that was not among the top 10% of pupils in that education system. By focusing on the 10th and 90th percentiles, we are able to provide a better picture of the distribution of reading performance within England,

and assess how wide the gaps are between the highest-performing and lowest-performing pupils, and how this has changed over time.

We also often refer to the 50th percentile, which represents the median score within an education system; this is the score which separates the bottom 50% of pupils' scores from the top 50%. Note that these scores often differ slightly to the mean scores reported throughout the rest of the report, because the distribution of scores in education systems are not necessarily symmetrical around the mean.

Table 12 summarises overall reading scores at the 10th, 50th, and 90th percentiles for England and the comparator systems in PIRLS 2021. The table also shows the range between the 10th and 90th percentiles.

Table 12: Average scores at the 10th, 50th and 90th percentile for England andcomparator systems (2021)

Education system	10 th Percentile	50 th Percentile	90 th Percentile	Range
England	458	562	651	192
Singapore	473	597	686	213
Hong Kong SAR	488	579	651	163
Australia	432	546	639	207
International Median	422	525	610	188

Ranges may appear inconsistent with percentiles due to rounding. Range calculated as 90th percentile – 10th percentile. International Median range is the range between the 10th percentile median and the 90th percentile median.

Source: IEA's PIRLS 2021

The range between the 10th and 90th percentiles in England is 192 points – slightly above the International Median range of 188 points. It should however be noted that the ranges in scores were generally larger in lower-performing systems, and smaller in many of the higher-performing systems. The median range between the 10th and 90th percentiles was 187 points for systems performing above the overall reading International Median score of 520, slightly less than the range in England. By contrast, for systems with overall reading scores lower than 520, the range between the 10th and 90th percentiles was 216 points.

Hong Kong had the smallest range in performance of all participating systems, followed by the Netherlands and Italy. Singapore and Australia meanwhile had some of the largest ranges in performance among the higher-performing systems. Of the education systems performing above the International Median, New Zealand had the largest range in performance between the 10th and 90th percentiles, and larger-than-average ranges in performance have been typical for education systems that administer PIRLS tests in English in previous cycles. This was also the case in Northern Ireland, Ireland and the United States in PIRLS 2021.

England's score of 458 at the 10th percentile was significantly lower than the 10th percentile scores of Hong Kong (488) and Singapore (473), while England's score of 651 at the 90th percentile was significantly lower than Singapore (686) and not significantly different to Hong Kong (651). England's scores at both the 10th and 90th percentiles were significantly higher than Australia's scores at these percentiles.

4.2 Trends in performance of lower and higher performing pupils

Figure 10 shows the trends in England's scores at the 10th percentiles across all 5 PIRLS cycles relative to these trends in the comparator systems. England's score of 458 at the 10th percentile is significantly higher than England's score at this percentile in every other PIRLS cycle other than PIRLS 2016. Singapore's performance at the 10th percentile in 2021 was significantly higher than in 2001 and 2006 but not in other PIRLS cycles, and similarly Hong Kong had higher 10th-percentile pupil performance relative to 2001 but not relative to other cycles. Australia's performance at the 10th percentile was not significantly different to their performance in either of their previous participations.

While the trend among the comparator systems was relatively stability in performance at the 10th percentile, particularly between PIRLS 2016 and PIRLS 2021, the International Median dropped by 21 points over this period. As has been previously reported, this finding is in keeping with the general trend of the comparator systems, who did not experience the same drops in performance experienced by most participating education systems in PIRLS 2021.

Figure 10: Performance at the 10th percentile for England and comparator systems across PIRLS cycles

500-475 450 425 400 2001 2006 2011 2016 2021 \diamond - - - -England Singapore Hong Kong Australia Int. Median Education system 2016 2001 2006 2011 2021 England *423 *437 *440 455 458 *402 *456 459 469 473 Singapore *445 488 Hong Kong SAR 486 492 487 Australia n/a n/a 418 432 432 416 433 435 443 International Median 422

Overall Reading Score (10th percentile)

Asterisks (*) indicate that the score shown is significantly different to that education system's score for PIRLS 2021.

Source: IEA's PIRLS 2021

Figure 11 shows the trends in overall performance at the 90th percentile across PIRLS cycles from 2001, 2006, 2011, 2016, 2021. In England, performance at the 90th percentile was quite stable over time; 2021 results were not significantly different from results in any earlier PIRLS cycle. The International Median has been similarly stable across cycles, with a slight decrease between 2016 and 2021. Singapore, on the other hand, has shown a consistently increasing pattern of performance over time at the 90th percentile. Hong Kong and Australia each show marked increases in 90th-percentile reading performance following the first cycles in which they participated (2001 for Hong Kong and 2011 for

Australia), but then level off somewhat thereafter. In Hong Kong, the PIRLS 2021 90th-percentile score is significantly higher than in 2001, 2006 and 2011 but not 2016, and in Australia the 2021 score at the 90th percentile is significantly higher than in 2011 but not 2016.

Figure 11: Performance at the 90th percentile for England and comparator systems across PIRLS cycles



Overall Reading Score (90th percentile)

Asterisks (*) indicate that the score shown is significantly different to that education system's score for PIRLS 2021.

Source: IEA's PIRLS 2021

Together, **Figure 10** and **Figure 11** show that the overall distribution of performance in England has not changed much since 2016; there has been a small, not statistically significant narrowing of the gap between the highest-performing and lowest-performing pupils.

Figure 12 provides a different approach to understanding trends over time within England, showing the spread of reading performance between the 10th and 90th percentiles across all of the PIRLS cycles since 2001. The figure shows the narrower overall range between these percentiles in the 2021 cycle compared to previous cycles, as well as the fact that (except in 2006, when the overall distributions shifted downward notably) performance at the 90th percentile has tended to remain roughly similar, as shown by England's data in **Figure 11**. It seems the decrease in overall range between 10th and 90th percentiles is a consequence of the increase in 10th-percentile reading performance, a promising finding in light of the fact that efforts to narrow achievement gaps generally aim to boost outcomes for lower performers without unduly compromising learning and outcomes for higher performers.



Figure 12: Distribution of England's performance between the 10th and 90th percentiles across PIRLS cycles

PIRLS Cycle	10 th Percentile	50 th Percentile	90 th Percentile	Range
2021	458	562	651	192
2016	455	564	655	200
2011	440	558	652	211
2006	423	546	645	222
2001	437	559	658	222

Range calculated as 90th percentile – 10th percentile.

Ranges may appear inconsistent with percentiles due to rounding.

Source: IEA's PIRLS 2021

Finally, **Figure 13** provides insight into how England's reading performance was distributed between 10th and 90th percentiles for each subscale (Literary Purpose, Informational Purpose, RSI Process and IIE Process), focusing on how this has changed in the 20 years since England's participation in the original cycle of PIRLS in 2001. On the whole, the range between 10th and 90th percentiles has narrowed from the 2001 cycle across all of the subscales. Median-level performance across each of the subscales have remained stable with the exception of a more substantial increase from 2001 to 2021 in the Informational Purpose Median. To summarise, again there is some stability in terms of average and high performance although there is slightly more change in 90th percentile literary purpose performance, but apparent improvement in England at the lower end of the distribution.

Figure 13: Comparison of the distributions of England's performance on the Reading Purpose and Comprehension Process subscales in PIRLS 2021 and PIRLS 2001



10th - 50th 50th - 90th

Subscale	Year	10 th Percentile	50 th Percentile	90 th Percentile	Range
Literary Purpose	2021	457	563	653	196
Literary Purpose	2001	433	567	679	246
Informational Purpose	2021	453	564	657	204
Informational Purpose	2001	434	554	654	220
RSI Process	2021	452	558	651	198
RSI Process	2001	430	554	659	229
IIE Process	2021	463	565	653	190
IIE Process	2001	442	563	659	217

Range calculated as 90th percentile – 10th percentile.

Ranges may appear inconsistent with percentiles due to rounding.

Source: IEA's PIRLS 2021

4.3 Contextualisation: Achievement of higher and lower performing pupils in Singapore

The gap in performance between higher- and lower-achieving pupils, often referred to as the attainment gap, is an area of focus for a number of education systems around the world. In many education systems, substantial resources and effort is put into reducing the attainment gap. However, when looking at reducing this gap, the direction of change is an important consideration. It is not enough to say that the gap has narrowed if the performance shifts towards the lower end of the range of achievement as this would mean that lower-performing pupils are not improving, and higher-performing pupils are getting worse. In most education systems, including England, reducing the attainment gap aims to improve the performance of lower-performing pupils, while ensuring higher-performing pupils continue to show positive progress. **Box 4.1** describes the situation in Singapore, where there have been consistently positive gains at the higher end of achievement.

Box 4.1 Singapore: Improving performance and reducing attainment gaps

Singapore has a large gap in performance of pupils at the 10th and 90th percentiles, one of the largest of all participating education systems in PIRLS. Despite this, it is also the only participating system where achievement across the higher PIRLS International Benchmarks has shown consistent growth across all cycles. Furthermore, the proportion of pupils not reaching the Low Benchmark has reduced, from 10% in 2001 to 3% in 2021, and the percentage of pupils achieving each International Benchmark has increased across each cycle.

Approaches to ensuring all pupils become skilled readers is at the forefront of policy and curriculum in Singapore. The reading and language curriculum focuses on building a strong foundation of basic literacy skills in order to enable high performance throughout schooling (Ministry of Education, 2022a). To ensure this is achieved for all pupils, the Ministry of Education in Singapore has a number of resources and interventions focused on identifying and providing additional support for pupils who have special educational needs, and/or may be at risk of lower literacy levels (Ministry of Education, 2021). Early screening is conducted in first grade to identify pupils for a learning support programme. These pupils receive additional instructional support for 2 years to build their basic literacy skills. Teachers are encouraged to use differentiated instruction and provide extra support for pupils that need it. Additionally, pupils who are identified as having special educational needs are supported by a core group of teachers and allied professionals who receive specific training in supporting special needs (Ministry of Education, 2021).

In 2021, more than a third of pupils in Singapore (35%) reached the Advanced Benchmark, the highest proportion of pupils of all participating systems (Mullis et al., 2023). There are a number of education systems that have been making progress at the Advanced Benchmark level across cycles, however, between 2016 and 2021, Singapore was also the only participating education system to show growth at the Advanced Benchmark level. There are several potential reasons for the growth in performance at higher end of achievement. Although it is not possible to directly measure the impact of COVID-19 interruptions and closures on PIRLS performance over this period, there are some insights that can be gained from Singapore's approach to learning before, during and after the pandemic.

Continues on next page

Box 4.1 (continued)

The shift toward system-wide technology-enabled teaching and learning began in Singapore in 1997 and implementing educational technology in evolving and responsive ways continues to be a national priority (Ministry of Education, 2022b). This means that when in-person teaching and learning was interrupted by the pandemic in 2020, educational-technology systems in Singapore were well placed to ensure a smooth transition to online teaching and learning. Considerable efforts were made to ensure that there was little to no interruption to teaching and learning as a result of the pandemic (Ministry of Education, 2022a). These efforts included ongoing check-ins with schools to address their emerging needs and provide the necessary support in terms of resources and planning, providing digital resources and internet at home to pupils who needed them, and a implementing a curriculum continuity plan. Blended learning approaches that were implemented during the pandemic have become a regular feature of teaching and learning in Singapore; these aim to encourage self-directed learning and develop passionate, lifelong learners (Ministry of Education, 2020).

5 Reading performance by prior attainment

Chapter overview

Chapter 5 focuses on associations between pupils' performance in PIRLS and their prior reading attainment. In particular, this chapter looks at the relationship between pupils' results in PIRLS 2021 relative to their performance on their year 1 and year 2 phonics screening checks, as well as their performance on key stage 1 assessments of reading. In addition, the relationships between these 2 prior assessments of reading and pupils' performance in PIRLS are examined across 2 cycles (2016 and 2021) to understand if the strengths of relationships have changed over time. The chapter concludes with a discussion of relevant results from a comparator education system.

Key findings

- There is a moderate, positive correlation between performance in the year 1 phonics screening check and performance in PIRLS 2021.
- Higher performance on the year 2 phonics screening check was generally associated with higher performance in PIRLS 2021. This finding relates to a smaller group of pupils, as only pupils who did not meet expected standards on or did not participate in the year 1 phonics screening check took part in the year 2 phonics screening check.
- Nearly 90% of pupils classified as reading at a greater depth within the expected standard on the key stage 1 reading assessment scored above the 'High' International Benchmark score of 550 in PIRLS 2021.
- The median score of pupils who were classified as working at the expected standard on the key stage 1 reading assessment was 558, in line with England's average score in PIRLS 2021.
- Higher key stage 1 reading assessment outcomes were generally associated with higher performance in PIRLS 2021 on average. Pupils classified as working towards the expected standard on the key stage 1 reading assessment, and those classified as pre-key stage, had wider ranges of performance in PIRLS 2021.

5.1 PIRLS performance by prior performance in phonics screening checks

Year 1 pupils in England who attend state-maintained schools or academies are expected to participate in an assessment called the phonics screening check⁵. The aim of the check is to identify pupils who are not yet fluent in decoding the sounds of written words and blending them together to read them aloud. Pupils who take the phonics screening check are asked to read aloud 40 words presented to them. Twenty of these words are real words, with most being a single syllable in length (e.g., "print", "clouds"), and a few having 2 syllables (e.g., "fabric", "trapeze"). The other 20 words are 'pseudowords'. Pseudo-words are not real words but can be read aloud phonetically (e.g., "barp", "chell"). Pupils taking the check are presented with 20 alien-like creatures and asked to read out their pseudo-word names presented next to their pictures. This part of the check ensures that pupils taking the check cannot rely on word recognition alone, and must be able to decode the phonetic sounds of unfamiliar words. Since the introduction of the phonics screening check in the 2011/2012 academic year, an 'expected standard' has been set at the correct reading of 32 of the 40 words across the check.

Of the 4,150 pupils in England who participated in PIRLS 2021, year 1 phonics screening check performance data was available for 3,619 pupils. These pupils took the phonics screening check near the end of the 2017/2018 academic year. Of these 3,619 pupils, 87.3% scored at or above the 'expected' standard threshold of 32 marks out of a possible 40. This figure is above the national average of 82% of pupils who met this threshold in the 2018 assessment. The average (mean) score of these pupils in PIRLS 2021 was 557, not statistically significantly different to the average score of England's full PIRLS 2021 cohort (558).

Figure 14 shows how pupils' performance in their year 1 phonics screening check relates to their performance in PIRLS. Data for pupils who scored 32 or higher and therefore met the expected standard are presented in blue at the top of the figure, while the data for pupils who scored less than 32 and so did not meet this expected standard are presented in red, making up the bottom 3 rows of the figure. Because only a small proportion of pupils in England's sample did not meet the expected standard, their results have been grouped into 3 categories (0-15 marks, 16-24 marks, and 25-31 marks).

⁵ Pupils who use British Sign Language or other sign-supported communications, or who are non-verbal / selectively-mute, have recently moved to the country and unable understand English letters/sounds, and other pupils who show no understanding of grapheme/phoneme correspondences are not required to take the phonics screening check.





■ 50th - 90th (Met Expected Standard)

■ 10th - 50th (Below Expected Standard) ■ 50th - 90th (Below Expected Standard)

Phonics Screening Check Mark	Percentage of pupils	10 th percentile	50 th percentile	90 th percentile	Range
40/40	13.8%	531	606	680	149
39/40	14.0%	510	587	665	155
38/40	13.8%	485	568	651	165
37/40	12.8%	480	563	649	169
36/40	10.3%	468	561	643	175
35/40	8.7%	459	552	638	178
34/40	6.7%	448	535	619	171
33/40	4.4%	452	535	609	157
32/40	2.7%	418	521	630	212
25–31	4.4%	420	512	597	176
16–24	3.9%	384	496	585	201
0–15	4.3%	350	467	568	218

Pupils scoring less than 32/40 did not meet the expected standard.

Range calculated as 90th – 10th percentile. Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021
Pupils who answered all 40 questions correctly in the year 1 phonics screening check had a median PIRLS score of 606, with a relatively narrow distribution of performance between the 10th and 90th percentiles. By contrast, there was a much wider range of performance between the 10th and 90th percentiles among pupils that scored around, or below the expected standard threshold of 32/40. The overall correlation between phonics marks and PIRLS scores was 0.43, indicating a moderate, statistically significant relationship between performance in the 2 assessments.

Pupils who do not meet the expected standard in year 1, or who did not participate in the phonics screening check in year 1 are normally expected to participate in the screening check in year 2. In the PIRLS cohort, a total of 519 pupils participated in the year 2 phonics screening check near the end of the 2018/2019 academic year.

Figure 15 shows how pupils' performance in the year 2 phonics screening check relates to their performance in PIRLS. Due to the low number of pupils who participated in a year 2 phonics screening check, data has been combined into 4 groups; pupils scoring between 38 and 40, pupils scoring between 35 and 37, pupils scoring between 32 and 34, and the remaining pupils who scored less than 32, and therefore did not meet the expected standard. As with the year 1 phonics screening check data shown in **Figure 14**, there is generally a wider range in performance for pupils who scored below, or close to the expected standard of 32 marks out of 40, and narrower for pupils scoring at the upper end of the scale. Stronger performance in the year 2 phonics screening check was generally associated with stronger performance in PIRLS 2021.





10th - 50th (Below Expected Standard)

■ 50th - 90th (Met Expected Standard)

■ 50th - 90th (Below Expected Standard)

Phonics Screening Check Mark	Percentage of pupils	10 th percentile	50 th percentile	90 th percentile	Range
38–40	22.3%	449	543	613	165
35–37	22.0%	409	503	600	191
32–34	20.5%	396	506	596	200
0–31	35.2%	347	467	562	215

Pupils scoring less than 32/40 did not meet the expected standard.

Range calculated as 90th – 10th percentile. Some results may appear inconsistent due to rounding. Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

5.2 PIRLS performance by prior performance in key stage 1 reading

At the end of Year 2, teachers of pupils in state-maintained schools and academies provide a judgement as to how well their pupils are performing in reading, writing, mathematics, and science relative to an 'expected standard'. This expected standard is comprised of a series of statements about what the pupil can do. For example, a pupil who is performing at the expected standard in reading can "accurately read most words" that contain 2 or more syllables, that contain common suffixes, and most common exception words. They can also read aloud accurately without overt sounding and blending, and fluently enough so that their reading of age-appropriate books is focused on understanding rather than the decoding of individual words. Additional statements are provided to allow teachers to judge whether pupils are working at greater depth within the expected standard, or whether they are working towards the expected standard.

Teacher-assessed key stage 1 (KS1) reading assessment data was available for 3,669 out of the 4,150 pupils in England's PIRLS 2021 cohort. These pupils were assessed by their teachers near the end of year 2, in the 2018/2019 academic year. Teachers assessed whether their pupils were working 'towards the expected standard', 'at the expected standard', or at a 'greater depth within the expected standard'. Some pupils were also classified as working at pre- key stage levels – at the time, these pupils would have been following a slightly different curriculum than the typical key stage 1 reading curriculum.

In England's PIRLS 2021 cohort, 22.3% of the pupils were evaluated as working either towards the expected standard, or at pre- key stage levels below this, while 27.5% were evaluated as working at a 'greater depth' within the expected standard.

Figure 16 shows how pupils' teacher assessed grades in KS1 reading related to performance in PIRLS. Data for pupils who were classified as working at the expected standard in reading, or at a greater depth within the expected standard are presented in blue in the top 2 rows of the figure, while the data for pupils who were classified as working below the expected standard are presented in red in the bottom 2 rows.

Pupils who were classified as reading at a greater depth within the expected standard at the end of year 2 had strong levels of performance in PIRLS, with nearly 90% of these pupils scoring above the 'High' International Benchmark score of 550, and there generally being a narrow range of performance between the 10th and 90th percentiles of PIRLS performance. By contrast, there was a much wider range in PIRLS performance for pupils who were classified as 'working towards' the expected standard in reading at the end of year 2. The median score of pupils who were classified as working at the expected standard was 558, in line with England's average score (558) in PIRLS 2021.

Figure 16: PIRLS performance between the 10th and 90th percentiles for pupils in England by their key stage 1 reading outcome



■ 10th - 50th (Met Expected Standard) ■ 50th - 90th (Met Expected Standard)

■ 10th - 50th (Below Expected Standard) ■ 50th - 90th (Below Expected Standard)

Key stage 1 reading outcome	Percentage of pupils	10 th percentile	50 th percentile	90 th percentile	Range
Greater depth	27.5%	542	610	682	139
Expected standard	50.2%	480	558	636	156
Working towards	17.9%	409	498	583	173
Pre- key stage	4.4%	351	465	561	210

Greater depth = greater depth within the expected standard.

Expected standard = working at the expected standard.

Working towards = working towards the expected standard.

Pre- key stage = not currently working towards the expected standard.

Range calculated as 90th – 10th percentile. Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

5.3 Contextualisation: Approaches to reading screening across education systems

The ways in which pupils' reading progress is monitored in the primary school vary across education systems. In almost every education system that participates in PIRLS, pupils' reading progress is monitored at a national level, typically relying on qualitative monitoring by teachers, national tests, standardised diagnostic assessments, or a combination of these approaches (Reynolds et al., 2022). The monitoring of pupils' reading progress across all education systems involves both formative and summative

assessment, and in most cases reports on both pupil-level and system-level results. In a number of participating education systems, additional screening measures are taken early to identify individualised reading needs of pupils. **Box 5.1** discusses examples of different approaches to identifying pupils' reading needs from selected education systems.

Box 5.1 Screening for reading difficulties in the early years

The early identification of pupils who have difficulty reading is understood to be an important step in ensuring appropriate support is provided so that all pupils progress well in reading (January & Klingbeil, 2020). Across participating education systems, a number of different screening methods are applied in order to identify pupils who require additional support and/or interventions related to reading. These typically happen at one of the following stages, before beginning primary school (i.e., school readiness), in the early grades (e.g., year 1-2), or at the end of a foundational phase (e.g., year 3 and/or 6).

School readiness screening typically takes place before children enter primary school, in most cases at 5 or 6 years of age. These screening tests often do not focus on reading, but rather on the oral language skills foundational to reading development. In Bulgaria, all pupils are evaluated for their school readiness at the start of their first year of primary school for teachers to differentiate pupils' individual needs (Mavrodieva & Damyanova, 2022). Similarly, in Denmark, pupils participate in mandatory language screenings at the beginning of grade 0, when they are 6 years old (Fougt et al., 2022). The Flemish community of Belgium has a similar mandatory test when children enter primary school at 5 years old, an oral language screening known as the KOALA test (Denies et al., 2022). In Belgium, schools are encouraged to compare the results of external tests such as KOALA to their classroom-based methods to obtain more objective assessments of pupils' progress.

Continues on next page

Box 5.1 (continued)

Early grade reading screening, such as the phonics screening check conducted in England, usually takes place at the end of the first or second year of primary school. These screenings are more focused on reading rather than oral language skills and typically include evaluations of single word reading, phonics, letter-sound recognition, and/or decoding skills. The phonics screening check in Australia, which takes place in year 1, was implemented to support reading progress following concerns over declining reading progress was suggested by international assessments such as PISA (Knowles & Hillman, 2022). South Africa and North Macedonia both use the Early Grade Reading Assessment (EGRA) to assess pupils' pre-reading and reading skills in the early grades (Mihajlovska, 2022; van Staden & Roux, 2022). In both cases, EGRA is used as a method of identifying system-wide needs for interventions in reading as well as identifying individual pupil needs.

In some education systems, diagnostic screening takes place throughout primary school in order to check reading 'milestones' and ensure that pupils are given adequate support as they progress each year. Oman, Latvia, Qatar and the United Arab Emirates are examples of systems that apply regular screening assessments (e.g., annually) to identify and support pupils' needs (Reynolds et al., 2022). In many cases, these assessments are focused on reading as well as mathematics skills.

There are also contexts that do not implement screening methods at the national level for all pupils. In many cases teachers use classroom-based methods to identify the reading needs of their pupils in order to inform interventions where necessary. In Slovakia, for example, if teachers notice that pupils are having difficulties developing reading skills, they can recommend that pupils then undergo an evaluation at a pedagogical-psychological counselling and prevention centre (Kopas, 2022).

6 Reading performance by pupil characteristics

Chapter overview

This chapter uses linked data from the NPD and data from the PIRLS 2021 pupil questionnaires to investigate the associations between pupil characteristics and pupil performance in PIRLS. Using these linked datasets, we investigate which pupil characteristics with the potential to influence reading achievement (including gender, age, eligibility for FSM within the last six years, ethnic group, whether or not pupils are classified as learning EAL, and books at home) are associated with PIRLS reading performance. These relationships, as well as the influence of prior attainment, are considered all together alongside school context (attainment band, from low to high) using multiple regression. Then relationships are considered individually for a selection of the pupil characteristics to better understand these associations. Particular attention is given to the gender gap and any widening or narrowing of this gap over time. The chapter ends with a discussion of relationships between pupil background characteristics and reading performance from an alternative education system to provide additional insight into the situation in England.

Key findings

- After accounting for pupil characteristics, prior attainment and school attainment band all together, the most powerful predictors of PIRLS 2021 overall reading score were pupils' year 1 phonics screening check mark (with higher marks predicting higher PIRLS scores), books at home (with more books predicting higher PIRLS scores) and FSM eligibility in the past 6 years (with FSM-eligible pupils scoring about 23 points lower than FSM-ineligible peers on average). Being born earlier in the school year, being in the 'Mixed' ethnic group and being in mid-high and high performing schools were all significantly associated with higher overall reading performance, but these were much weaker predictors. No other relationships were significant based on multiple regression results.
- Girls have continued to outperform boys as in previous cycles of PIRLS, with significant differences both in terms of overall performance and all PIRLS subscales except for the Informational Purpose Scale.

Continues on next page

Chapter overview (continued)

- The gender gap has continued to narrow in PIRLS 2021 overall. For high (90th percentile) and low (10th percentile) performers, there was very little change in the gender gap in England since PIRLS 2016.
- Pupils in the 'Mixed' ethnic group scored highest on average, 17 points higher than pupils from the 'White' or 'Black' ethnic groups who scored lowest on average in PIRLS 2021.
- Being born earlier in the school year tended to correspond to higher PIRLS scores on average, with about a 20 point difference between pupils born in September and those born the following August.

6.1 The relationship between PIRLS performance and pupil characteristics

In this chapter, we begin by looking at how several different pupil background characteristics predict the performance of pupils in England in PIRLS 2021. This includes characteristics such as pupil age, gender, major ethnic group, language via an indicator of whether a pupil has EAL, and socioeconomic background via pupils' eligibility for FSM within the past 6 years, as well as pupils' performance in the year 1 phonics screening check. We also include school attainment bands (based on school mean attainment in PIRLS 2021) to account for the school contexts of the pupils in the PIRLS 2021 sample.

While this chapter later includes consideration of each of these pupil characteristics in isolation, we begin by exploring how each of these characteristics independently predicts performance in PIRLS 2021 when we consider them all simultaneously. We do this using a method called multiple linear regression, which helps to take account of the complex relationships between predictors. For example, it is possible that pupils of one ethnic group might score significantly higher on average than pupils of another ethnic group, but a multiple linear regression analysis might find that these apparent differences are actually accounted for by some other pupil characteristic such as eligibility for FSM, and that the ethnic group difference is diminished after including free school meal eligibility together with ethnic group in the same analysis.

The multiple linear regression analysis of data from England's pupils in PIRLS 2021 focuses on 8 pupil characteristics and 1 school context variable. These were:

- Pupil mark (out of 40) in the year 1 phonics screening check
- Pupil age (in years, e.g., 10.25 means 10 years and 3 months) as recorded in PIRLS

- Whether a pupil had been eligible for FSM within the last 6 years as recorded in the NPD
- Pupil gender as recorded in PIRLS ('Female' and 'Male')
- Pupil major ethnic group as recorded in the NPD ('White', 'Black', 'Asian', 'Mixed', and 'Other')
- Whether a pupil has EAL or not as recorded in the NPD
- How many books a pupil reported having at home as recorded in PIRLS ('0-10', '11-25, one bookshelf', '26-100, one bookcase', '101-200, two bookcases', and 'More than 200, three or more bookcases').
- The performance of the school that the pupil attends (corresponding to 5 bands of attainment in PIRLS 2021 from 'Low-Performing' to 'High-Performing'⁶; see section 8.2.1 for more information).

For each of the pupil characteristics that are categorical in nature (e.g., 'Yes/No', 'Female/Male', or 'White/Black/Asian/Mixed/Other', etc.), a reference category must be selected so that all other categories are compared to that one in a multiple regression analysis. For this analysis, almost all of the selected reference categories are, as is quite common practice, the largest groups. This means that the reference categories are pupils who were not eligible for FSM within the last 6 years, female pupils, pupils from the 'White' ethnic group, pupils without EAL. The only exceptions to this were for the 2 characteristics with low to high categories; these were pupil-reported books at home, where the reference category was chosen to be the lowest (ten or fewer books), and school attainment band (where 'Low-performing' was chosen as the reference category). In both cases, the lowest rated category was chosen, to look at how each increase in the number of books at home, or school-performance, related to changes in PIRLS performance.

Table 13 provides a summary of the pupils included in the regression analysis. Pupils who did not have complete data for all of the variables listed above were not included in the multiple regression analysis. In total, this meant that 3,479 pupils were included, and their average PIRLS score was 558, identical to the average score for England overall (558) based on the overall PIRLS 2021 sample of 4,150 pupils. Further details regarding missing data from the NPD are available in Appendix B.

⁶ There were originally 6 school attainment bands in PIRLS, one of which corresponded to independent schools. This independent school band does not appear in the multiple regression model because NPD data is not available for pupils in independent schools, so for the purposes of this chapter there were 5 school attainment bands.

Pupil characteristic	Total N	Weighted %
Year 1 phonics check mark	3,479	100.0
Pupil age (in years)	3,479	100.0
Ever6 FSM – No	2,568	73.9
Ever6 FSM – Yes	911	26.1
Gender – Female	1,795	51.3
Gender – Male	1,684	48.7
Ethnic group – White	2,566	77.1
Ethnic group – Black	168	4.3
Ethnic group – Asian	476	11.3
Ethnic group – Mixed	210	5.7
Ethnic group – Other	59	1.6
EAL – No	2,766	82.0
EAL – Yes	713	18.0
Books at home – 0-10	437	12.0
Books at home – 11-25	861	24.8
Books at home – 26-100	1,160	33.4
Books at home – 101-200	618	17.6
Books at home > 200	403	12.2
School – Low performing	457	12.3
School – Mid-low performing	699	21.2
School – Middle performing	837	25.0
School – Mid-high performing	808	21.7
School – High-performing	678	19.9

Table 13: Description of pupil characteristics variables explored in the regressionanalysis

Please note that these are counts and percentages after removing cases with missing data on any variable, for a total of N=3479 cases (having dropped 671 cases missing data on at least one variable). This differs from the total count for later tables in this chapter, where only cases missing data on the variables of interest were omitted for each specific table.

Because of rounding, some results may appear inconsistent.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Table 14 displays the results of the multiple regression analysis. The unstandardised coefficients (b) indicate how many points' difference corresponds to a one-unit change in a particular variable, or (for categorical variables) how many points' difference there is between a particular category and the reference category. For example, after accounting for other characteristics, the analysis suggests that pupils who were eligible for FSM within the last 6 years scored on average about 23 points lower than those who were not eligible, after accounting for all of the other variables in the model. As another example, an increase of one point on the year 1 phonics screening check was associated with an

almost 4 point gain in PIRLS. In addition to these interpretations of unstandardised coefficients, the standardised beta coefficients (β) provide a measure of 'effect size', which in multiple linear regression allows us to compare across predictor variables and determine which are more or less powerful predictors of the outcome (PIRLS 2021 overall reading score).

Pupil characteristic	b	b – statistically significant at confi- dence level	β	β – statistically significant at confi- dence level			
(Constant)	234.68	99.9%	n/a	n/a			
Year 1 phonics check mark	3.64	99.9%	0.34	99.9%			
Pupil age (in years)	14.94	99.9%	0.06	99.9%			
Ever6 FSM – Yes	-22.72	99.9%	-0.13	99.9%			
Gender – Male	-3.26	n/a	-0.02	n/a			
Ethnic group – Black	10.23	n/a	0.03	n/a			
Ethnic group – Asian	8.19	n/a	0.03	n/a			
Ethnic group – Mixed	14.70	99%	0.05	99%			
Ethnic group – Other	21.93	n/a	0.04	n/a			
EAL – Yes	2.41	n/a	0.01	n/a			
Books at home – 11-25	25.10	99.9%	0.14	99.9%			
Books at home – 26-100	39.40	99.9%	0.25	99.9%			
Books at home – 101-200	48.20	99.9%	0.24	99.9%			
Books at home > 200	55.65	99.9%	0.24	99.9%			
School – Mid-low performing	11.86	n/a	0.06	n/a			
School – Middle performing	8.25	n/a	0.05	n/a			
School – Mid-high performing	18.57	95%	0.10	95%			
School – High-performing	19.51	95%	0.10	95%			

Table 14: Regression analysis of pupil characteristics relative to PIRLS 2021 score
in England

b = unstandardised regression coefficient

 β = standardised regression coefficient

n/a = not statistically significant

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Performance in the year 1 phonics screening was found to be the strongest independent predictor of performance in PIRLS. After accounting for all other pupil characteristics, a difference of 10 points on the year 1 phonics screening check was associated with a difference of about 36 points in PIRLS 2021 overall reading performance.

The second strongest predictor of overall reading performance in PIRLS 2021 was the number of books at home. Pupils who reported having over 200 books at home scored about 56 points higher than those who reported having 10 or fewer books at home, while those who reported having 101 to 200 books, 26 to 100 books, and 11 to 25 books at home scored about 48 points, 39 points and 25 points higher than those who reported having 10 or fewer books at home, respectively.

The third strongest predictor of PIRLS overall reading score was eligibility for FSM in the last 6 years. Pupils who were eligible scored, on average, about 23 points lower than those who were not.

The relationship between school attainment band and individual pupil reading performance was not uniform across bands. Only the mid-high performing and high performing bands of school attainment were significantly different from the low performing band, and these were not strong predictors of individual performance in PIRLS 2021.

Of the major ethnic groups included in the multiple regression analysis, only the 'Mixed' ethnic group performed significantly differently to the 'White' reference group, scoring approximately 15 points higher on average. Ethnic groups in general were not strong predictors of PIRLS reading performance.

Pupil age was a significant predictor of PIRLS reading performance, but not a strong predictor based on the standardised coefficient. Being one month older corresponded to approximately a 1.2 point gain in PIRLS overall reading score.

EAL status and gender were not significant predictors of pupils' overall reading performance in PIRLS 2021 after accounting for all of the other above variables.

Together, these pupil and school characteristics explained approximately 27% of the variation in PIRLS performance.

6.2 Performance by gender

6.2.1 Overall performance by gender

In keeping with previous cycles of PIRLS, girls significantly outperformed boys in the vast majority of education systems participating in PIRLS 2021 (Mullis et al., 2023). This was

the case in England, where girls scored, on average, 10 points higher than boys overall. Girls also scored higher than boys in all comparator education systems, as **Table 15** shows. In Hong Kong the gender gap was slightly lower (8 points) than in England, while in Singapore and Australia the gap was wider (18 and 17 points, respectively).

Gender gaps in overall performance varied substantially in size across the education systems participating in PIRLS 2021. Considering only English-speaking education systems, for example, there was no significant difference in the United States, a significant difference in Ireland very similar to that in England, and a difference in Northern Ireland of more than double that in England (Mullis et al., 2023).

Table 15: Average performance of girls and boys in England and comparatorsystems in PIRLS 2021

Education system	Girls' Average	Boys' Average	Gender Gap
England	*562	553	10
Singapore	*596	578	18
Hong Kong	*577	569	8
Australia	*549	532	17
International Median	*523	512	11

Asterisks (*) indicate that the average score for girls in that education system was significantly higher than the score for boys.

International Median gender-gap is the gap between girls' and boys' international medians.

Source: IEA's PIRLS 2021

6.2.2 Trend performance by gender

As **Figure 17** shows, girls' reading performance in England has remained relatively stable across PIRLS cycles, with the exception of significantly lower performance in the 2006 cycle (compared to PIRLS 2021). The overall performance of girls in England was slightly lower in 2021 than in 2016, but this difference was very slight (4 points) and not statistically significant. The International Median has also remained fairly stable across cycles until PIRLS 2021, in which the median score across education systems was 20 points lower than in the 2016 cycle. This may have been due to the influence of COVID-19 on education and data collection across participating education systems. In Hong Kong, girls' reading performance has been fairly stable except for a significantly lower start in the first (2001) cycle compared to PIRLS 2021. By contrast, in Singapore there has been a pattern of consistent increase in girls' reading performance with each cycle of PIRLS, all significantly different from girls' performance in PIRLS 2021. The overall performance of girls in Australia was significantly lower in the 2011 cycle than in the 2021 cycle, but the difference between 2016 and 2021 cycles was not significant.

Figure 17: Average performance of girls in England and comparator systems across PIRLS cycles



Asterisks (*) indicate that the score shown is significantly different to that system's score for PIRLS 2021. Source: IEA's PIRLS 2021

Boys' reading performance trends over time across PIRLS cycles has been quite similar to girls' trends in some education systems (see **Figure 18**). England was a bit of an exception to this, as boys' performance decreased slightly between 2001 and 2006, increased steadily from 2006 through to the 2011 and 2016 cycles, and then remained similar between 2016 and 2021. In Hong Kong, boys' scores have been fairly stable except for a lower start in the 2001 cycle, significantly so compared to Hong Kong's performance in PIRLS 2021. Like girls, boys in Singapore showed a similar pattern of consistent and significant increase in reading performance across PIRLS cycles. Again like girls, boys in

Australia performed significantly lower in the 2011 cycle than in PIRLS 2021, but the difference between 2016 and 2021 cycles was not significant. Fluctuations in the International Median have been similar for boys and girls despite girls' higher median scores across education systems in each cycle.

Figure 18: Average performance of boys in England and comparator systems across PIRLS cycles

	Score				
600 -					
575-			نەتەرىر. بەرمىرىمى	· · · · · · · · · · · · · · · · · · ·	
550-	▲······				
525-		\diamond			
500 -					
2001	2006	2011		2016	2021
		·C		$\diamond -$ –	
England	Singapo	re Hong	Kong Au	stralia Int	. Median
Education system	2001	2006	2011	2016	2021
England	*541	*530	*540	551	553
Singapore	*516	*550	*559	*568	578
Hong Kong	*519	559	563	564	569
Australia	n/a	n/a	*519	534	532
International Median	516	527	524	532	512
Asterisks (*) indicate that the	e score shown is	significantly dif	ferent to that sv	stem's score for	PIRI S 2021

Mean Overall Reading Score

Asterisks (*) indicate that the score shown is significantly different to that system's score for PIRLS 2021. Source: IEA's PIRLS 2021

Given the relative stability of girls' reading performance in England alongside the general increase for boys across cycles of PIRLS, it should come as no surprise that the gender gap has narrowed over time across recent cycles after some fluctuation across PIRLS 2001, 2006 and 2011 as displayed in **Figure 19**. This pattern is distinct from that of

comparator education systems. Singapore saw a dramatic narrowing of the gender gap after the 2001 cycle, as did Hong Kong except for a noticeable and temporary increase in the gender gap in the 2011 cycle. Given only 3 comparable cycles for Australia, it is difficult to firmly establish a trend in the gender gap, with some fluctuation between 2011, 2016 and 2021 cycles.



Figure 19: Gender gap in England and comparator systems across PIRLS cycles

		0	I X	5	,
Education system	2001	2006	2011	2016	2021
England	22	19	23	15	10
Singapore	24	17	17	17	18
Hong Kong SAR	19	10	16	9	8
Australia	n/a	n/a	17	22	17

Gender gaps calculated as average girls' score – average boys' score in each cycle

Source: IEA's PIRLS 2021

Looking more widely at English-speaking education systems to which direct statistical comparisons are not possible due to different data collection timings, considerable variation in gender-gap trends over time are apparent. In Ireland, the gender gap has remained fairly stable over time, for example, while in Northern Ireland it has widened, and in the United States it has narrowed to the point of statistical insignificance.

6.2.3 Performance by gender on reading purpose scales

As was the case for overall reading performance, girls outperformed boys on the Literary Purpose Scale in PIRLS 2021. This was true in England and comparator education systems, and differences were all significant as **Table 16** shows. The gender gaps on this scale in Singapore and Australia were nearly double those in England and Hong Kong. Girls also tended to outperform boys on the Informational Purpose Scale across England and comparator education systems, as **Table 17** shows. However, unlike the gender gaps on the Literary Purpose Scale, the gender gaps on the Informational Purpose Scale were not significant in England or in Hong Kong.

Table 16: Performance of girls and boys in England and comparator systems onthe Literary Purpose Scale in PIRLS 2021

Education system	Overall Scale Score	Girls' Scale Score	Boys' Scale Score	Gender Gap
England	558	*565	551	14
Singapore	591	*604	580	25
Hong Kong	564	*570	558	12
Australia	543	*557	530	26
International Median	520	528	513	15

Asterisks (*) indicate that the average score for girls in that education system was significantly higher than the score for boys. This is not calculated for the International Median.

International Median gender-gap is the gap between girls' and boys' international medians. Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 17: Performance of girls and boys in England and comparator systems on the Informational Purpose Scale in PIRLS 2021

Education system	Overall Scale Score	Girls' Scale Score	Boys' Scale Score	Gender Gap
England	559	563	555	8
Singapore	586	*594	579	15
Hong Kong	582	585	580	5
Australia	539	*544	534	10
International Median	520	521	513	8

Asterisks (*) indicate that the average score for girls in that education system was significantly higher than the score for boys. This is not calculated for the International Median.

International Median gender-gap is the gap between girls' and boys' international medians. Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Figure 20 displays the differences across all PIRLS cycles between scores on the Literary and Informational Purpose Scales for girls and for boys. For girls, the Literary Purpose Scale scores have remained significantly higher than those on the Informational Purpose Scale over time, but there has been a substantial decrease in that difference from PIRLS 2001 to PIRLS 2021 (from 19 to 2). For boys, performance on the Literary Purpose Scale was significantly higher than on the Informational Purpose Scale in the 2001 cycle, but this difference was only a little over a third of the corresponding difference for girls. In later cycles, there was no significant difference between scores on these 2 scales for boys in 2006 and 2011, a very small difference in 2016, and a reversal in the most recent cycle so that boys scored significantly higher on the Informational Purpose Scale. This helps to explain the lack of difference in scores across these scales in PIRLS 2021 when gender is not taken into consideration (as reported in Chapter 3); that is, with girls scoring slightly though not significantly higher on Literary Purpose, and boys scoring significantly higher on Informational Purpose, there appeared to be very little overall difference in scores.





Gender	2001	2006	2011	2016	2021
Girls	19	5	7	10	2
Boys	7	0	0	3	-4
D'11 1 1 1 1 1			10 1		

Differences calculated as Literary Scale score – Informational Scale score

Source: IEA's PIRLS 2021

6.2.4 Performance by gender on comprehension process scales

Similar to the patterns reported above for overall reading performance and Literary Purpose, girls scored significantly higher than boys on the Retrieving and Straightforward Inferencing Process Scale in PIRLS 2021 in England as well as in comparator education systems (see **Table 18**). Gender gaps were similar on this size to those for overall reading performance, and mirrored the same pattern as for the other scales reported above, with larger differences between girls' and boys' scores in Singapore and Australia and smaller (though still significant) differences in England and Hong Kong. As **Table 19** shows, the situation for gender gaps on the Interpreting, Integrating and Evaluating Process Scale was almost identical to that for the Retrieving and Straightforward Inferencing Process Scale across England and comparator education systems.

Table 18: Performance of girls and boys in England and comparator systems onthe Retrieving and Straightforward Inferencing Process Scale in PIRLS 2021

Education system	Overall Scale Score	Girls' Scale Score	Boys' Scale Score	Gender Gap
England	554	*559	549	11
Singapore	584	*593	575	18
Hong Kong	577	*581	573	8
Australia	534	*542	525	17
International Median	520	526	513	13

Asterisks (*) indicate that the average score for girls in that education system was significantly higher than the score for boys. This is not calculated for the International Median.

International Median gender-gap is the gap between girls' and boys' international medians.

Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 19: Performance of girls and boys in England and comparator systems onthe Interpreting, Integrating and Evaluating Process Scale in PIRLS 2021

Education system	Overall Scale Score	Girls' Scale Score	Boys' Scale Score	Gender Gap
England	561	*566	556	10
Singapore	591	*599	583	16
Hong Kong	572	*576	569	7
Australia	547	*557	537	19
International Median	520	526	512	14

Asterisks (*) indicate that the average score for girls in that education system was significantly higher than the score for boys. This is not calculated for the International Median.

International Median gender-gap is the gap between girls' and boys' international medians. Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Although there has been some fluctuation over time, the differences between scores on the Reading and Straightforward Inferencing Process and the Interpreting, Integrating and Evaluating Process Scales have remained relatively stable over time since the first cycle of PIRLS for both boys and girls, with the exception of a spike in the 2011 cycle for girls. Overall, throughout every cycle including PIRLS 2021, both boys and girls have performed better on the Interpreting, Integrating and Evaluating Process Scale (see **Figure 21**).



Figure 21: Differences in RSI and IIE Process Scale scores across PIRLS cycles for girls and boys in England

Differences calculated as RSI scale score – IIE scale score

Source: IEA's PIRLS 2021

6.2.5 Performance by gender at 10th and 90th percentiles

Figure 22 displays the trends in boys' and girls' scores across PIRLS cycles at the 10th percentile, while **Figure 23** displays the corresponding trends at the 90th percentile. While the overall gap in reading performance between girls and boys was significant. as reported in **Table 15**, the gender gap was not significant when considering just the lowest (10th percentile) and highest (90th percentile) performers. To some extent, this is the result of a narrowing of the gaps between girls and boys amongst both high and low performers. However, it is also a consequence of the larger statistical uncertainty in analysing performance only at the extremes of the range of scores – in other words, looking only at the 10th and 90th percentiles involves analysing information from smaller

groups, and as group size gets smaller we need a larger difference for it to be able to be statistically significant.

Figure 22: Performance trends of girls and boys in England at the 10th percentile across PIRLS cycles



Overall Reading Score (10th Percentile)

Source: IEA's PIRLS 2021

Figure 23: Performance trends of girls and boys in England at the 90th percentile across PIRLS cycles



Overall Reading Score (90th Percentile)

6.3 Performance by pupil age

Figure 24 shows the average PIRLS 2021 performance of pupils in England by their month of birth, sorted in order of the school calendar year in England. The mean overall PIRLS 2021 score by birth month is listed below the figure. In PIRLS 2016, pupils in England born near the end of the calendar year scored approximately 35 points lower than pupils born near the beginning of the school year (McGrane et al., 2017). Figure 1 shows that, although not entirely consistent month-to-month, pupils born near the beginning of the school year (September, October and November) scoring around 20 points higher than those born in the fourth quarter (June, July, and August).



Figure 24: Average PIRLS 2021 scores of pupils in England by their month of birth

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Mean overall PIRLS 2021 score by birth month was:

- September 562
- October 577
- November 562
- December 564
- January 559
- February 559
- March 549
- April 565
- May 542
- June 548
- July 544
- August 543

6.4 Performance by ethnic group and English as an Additional Language (EAL)

6.4.1 Performance by ethnic group

The differences in PIRLS performance between pupils of different ethnic groups were mostly small, as shown in **Table 20**

Table 20. Pupils in the 'Mixed' ethnic group scored highest on average, 17 points higher than pupils from the 'White' or 'Black' ethnic groups who scored lowest on average in PIRLS 2021.

Pupil ethnic group	Percentage of pupils	Average PIRLS score
White	75.2%	554
Black	4.9%	554
Asian	12.2%	559
Mixed	5.9%	571
Other	1.9%	563

Table 20: Performance of England's PIRLS 2021 cohort by ethnic group

Some results may appear inconsistent due to rounding.

Pupil ethnic group is based on major ethnic group recorded in the NPD Pupils listed as being in the 'Chinese' major ethnic group have been combined with those listed as 'Asian' for consistency with how these are reported nationally.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

6.4.2 Performance by English as an Additional Language (EAL) status

The regression analysis earlier in this chapter showed that EAL status was not one of the significant predictors of pupils' overall reading performance in PIRLS 2021. However, even before accounting for the contribution of other pupil characteristics, there was little difference between the PIRLS 2021 performance of pupils with and without EAL. Results comparing these groups are shown in **Table 21**.

Table 21: Performance of England's PIRLS 2021 cohort by English as an AdditionalLanguage (EAL) status in the past 6 years

Has EAL	Percent of pupils	Average PIRLS score
No	80.0%	554
Yes	20.0%	557

Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

6.5 Performance by socioeconomic background

6.5.1 Performance by free school meal (FSM) eligibility

Table 22 shows that pupils who were eligible for FSM within the last 6 years scored 39 points lower in PIRLS 2021, on average, than those who were not eligible for FSM over this period.

Table 22: Performance of England's PIRLS 2021 cohort by their eligibility for freeschool meals (FSM) in the past 6 years

FSM eligible within the last 6 years	Percent of pupils	Average PIRLS score
No	73.6%	567
Yes	26.4%	527

Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

6.5.2 Associations between PIRLS performance and the number of books at home

Table 23 shows the percentages of pupils who reported having different numbers of books at home, as well as the corresponding average scores in PIRLS 2021. In general, higher numbers of books at home tended to be associated with higher performance in PIRLS 2021. Pupils with over 200 books at home scored about 85 points higher than those who had none or very few. The points difference between pupils with no books or very few and those with 11 to 25 books is the largest (35 points), with smaller points differences between each pair of categories as the number of books increases.

Table 23: Average PIRLS 2021 score of pupils in England and comparator systems
by the number of books they reported having at home

Books at home	Percent of pupils	Average PIRLS score
None or very few (0-10)	12.4%	507
Enough to fill one shelf (11-25)	24.4%	542
Enough to fill one bookcase (26-100)	32.9%	565
Enough to fill two bookcases (101-200)	17.4%	579
Enough to fill three or more bookcases (>200)	12.8%	591

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

6.6 Contextualisation: Changes in the magnitude of the gender gap over time

Across all participating education systems, girls consistently outperformed boys in PIRLS 2021, a trend also seen in previous cycles. Results from other international large-scale assessments such as PISA, and national large-scale assessments such as the National Assessment of Educational Progress (NAEP) in the United States, as well as metaanalyses of gender gaps for different educational outcomes also reflect higher achievement for girls than boys in the area of reading (Borgonovi et al., 2018; Kaplan & Jude, 2022; Reilly et al., 2019; Schleicher, 2019). When considering the gender gap in PIRLS performance over time, fluctuations and changes are noticeable, and in some education systems the gap between performance narrows. **Box 6.1** discusses education systems where the gender gap in PIRLS has narrowed between 2016 and 2021.

Box 6.1 Changes in gender gaps: boys improving or girls declining?

The gender gap across PIRLS cycles has remained reasonably consistent with girls outperforming boys to some extent across all contexts. However, 15 education systems, including England, show evidence of a gender gap that narrows somewhat over time. In many cases, particularly between 2016 and 2021, this is due to the performance of boys remaining reasonably stable while the performance of girls decreases. This narrowing of the gender gap due to a decrease in the performance of girls has also been seen in other assessments of reading (Borgonovi, 2022; Schleicher, 2019). The declining performance of girls has understandably raised concern for researchers, governments and policymakers across different education systems. The reasons behind the decline are not well understood, and contributing factors are often hard to disentangle.

In PIRLS, there are 4 education systems where the gender gap reduced between 2016 and 2021 due the average scores for boys increasing more than they did for girls, rather than a decline in girls' scores. In Egypt, Oman, Qatar, and Saudi Arabia, although performing below the International Scale Centrepoint for PIRLS (500), overall average performance for all pupils increased significantly between 2016 and 2021. Egypt and Oman participated in the original data collection period while Qatar and Saudi Arabia delayed their participation by six months due to COVID-19. This means that for Qatar and Saudi Arabia, the average age of participating pupils is higher than for education systems which collected data as originally planned. It is important to consider, therefore, that changes in the magnitude of gender gaps are generally seen as age increases. In reading, girls tend to outperform boys in primary school to a greater extent than in secondary school; the gap narrows further in higher education and by adulthood it is negligible (Solheim & Lundetræ, 2018). (Chapter 1 discusses the data collection periods in more detail, however, it is important to remember that generally where data collection was delayed - overall performance in PIRLS was higher than for education systems where data was collected in the original timeframe.)

Continues on next page

Box 6.1 (continued)

Although in all 4 systems there was a significant increase in overall PIRLS average for all pupils, when considering the trends in boys and girls average scores some differences are evident. In most cases, the girls and boys scores increased but the difference between boys scores across cycles was higher than the difference in average scores for girls. In Egypt, the average score for girls increased by 37 points, while the boys average increased by 58 points. In Oman, for girls the change was smaller just 5 points, while for boys a significant increase of 17 points was seen. In Qatar, a significant increase was seen for both girls (33 points) and boys (52 points). While in Saudi Arabia, the average score for girls did not change between 2016 and 2021, but the average score for boys increased by 36 points.

Of the education systems where the gender gap narrowed as a result of boys scores increasing, Oman is the only case where specific initiatives have been implemented that focus on reducing the gender gap. These include professional development initiatives focused on increasing the interest and motivation to read for boys, as well as differentiated teaching strategies to raise the level of reading comprehension for boys (Al Maskari et al., 2022).

Research into the gender gap in educational performance more broadly reveals some insights into changes in trends of performance for boys and girls. There is evidence to suggest that boys perform better than girls on digital format assessments (Borgonovi, 2016; Siddiq & Scherer, 2019). Possible reasons for this point to higher motivation and interest levels for boys when reading and taking assessments on computers (Cai et al., 2017; Borgonovi, 2016). Studies are also emerging looking at the impact of COVID-19 on educational outcomes for boys and girls which look at the transition to online learning, differences in school attendance across genders, and differences in home-care responsibilities (Blackman, 2022; Damani, 2022; De Paz Nieves, 2021). Considering that in a number of education systems, including England, there is evidence of a slightly declining performance of girls, understanding the potential causes of gender gap trajectories is important for ensuring constructive progress can be made.

7 Reading performance by pupils' motivations

Chapter overview

This chapter explores the pupils' attitudes towards reading, and how these attitudes relate to performance in PIRLS. This includes looking at pupils' confidence in reading, liking of reading, and engagement in reading lessons, and exploring how these differ with respect to pupil characteristics (gender, FSM eligibility within the past 6 years, and EAL status) and circumstances (e.g., access to books at home). The chapter also explores trends from 2011 and 2016 in responses to questionnaire items that we are able to compare over time, and concludes with an exploration of how much time girls and boys in England report that they spend reading outside of school.

Key findings

- Confidence in reading is strongly correlated with performance in PIRLS in England, though this is not the same across all education systems. In England, there is approximately a 90-point difference in overall reading performance between the most- and least-confident readers.
- Having more books at home was associated with higher levels of confidence in reading. Pupils who had been eligible for FSM in the past 6 years were less likely to be very confident. Within ethnic groups, higher confidence tended to be associated with higher reading performance where counts were high enough to provide reliable estimates. Pupils with and without EAL did not differ greatly in their confidence in reading, on average, nor in the association between confidence and PIRLS performance.
- Approximately a quarter of pupils in England 'do not like' reading. In England and in every other participating educational system, a higher proportion of boys reported not liking reading than girls. Pupils who like reading 'very much' scored higher on average than pupils who do not like reading.
- Fewer pupils in England reported that they enjoy reading than in the 2011 or 2016 cycles of PIRLS.
- Pupils in England spent slightly less time reading outside of school each day than pupils in most other education systems, and less time reading outside of school than reported in PIRLS 2011 and PIRLS 2016. Girls' time spent reading outside of schools decreased more than boys' in England, but a slightly higher proportion of girls than boys in England report reading for an hour or more each day outside of school.

7.1 Scales for pupils' motivations towards reading

After pupils completed their 2 test-booklets, they completed one additional questionnaire booklet asking them for information about their home environments, their activities at home (both directly and indirectly related to reading), their opinions about reading, and their own perspectives on their reading abilities. In this chapter, we focus on the results from this questionnaire concerning pupils' motivations, beliefs, and behaviours towards reading.

For many of the questionnaire items, pupils were given a statement relating to their attitudes towards reading. Examples of these statements include, "I would like to have more time for reading", and "I have trouble reading stories with difficult words". Pupils were asked to tick a box indicating how much they agreed with that statement on a four-point scale ("agree a lot", "agree a little", "disagree a little" and "disagree a lot").

Questionnaire items relating to similar aspects of pupils' motivations and attitudes towards reading were then combined into different scales, including a "Confident in Reading" Scale, a "Likes Reading" Scale, and an "Engaged in Reading Lessons" Scale. Each pupil's responses to the relevant items in each scale were combined using psychometric techniques to produce a scale score. These scale scores were then used to classify each pupil into one of 3 categories, such as 'very confident', 'somewhat confident', or 'not confident'.

7.2 Pupils' confidence in reading

The '**Confident in Reading'** Scale is based on pupils' responses to 6 items in the Pupil Questionnaire. Pupils were asked the following:

"How well do you read? Tell how much you agree with each of these statements."

- I usually do well in reading.
- Reading is easy for me.
- I have trouble reading stories with difficult words.
- Reading is harder for me than for many of my classmates.
- Reading is harder for me than any other subject.
- I am just not good at reading.

Table 24 shows the proportions of pupils in England and comparator education systems that were classified as 'very confident', 'somewhat confident' and 'not confident' readers. The table also shows the difference in the average PIRLS scores between the 'very confident' and 'not confident' readers in each educational system.

Table 24: Pupils' confidence in reading in England and comparator systems inPIRLS 2021

Education system	Very confident	Somewhat confident	Not confident	Points Difference
England	45%	34%	21%	90
Singapore	51%	33%	16%	107
Hong Kong	32%	39%	29%	66
Australia	43%	38%	19%	104

*Some results may appear inconsistent due to rounding.

** Points difference calculated as (average score of 'very confident' readers) – (average score of 'not confident' readers).

Source: IEA's PIRLS 2021

Around 45% of pupils in England were classified as being 'very confident' readers. This was similar to the percentage in Australia. It was also quite similar to the corresponding percentages in other high-performing English-speaking education systems including those with delayed assessment such as Ireland, Northern Ireland, and the United States.

By contrast, despite high levels of overall performance in Hong Kong, a considerably lower proportion of pupils reported high confidence in reading, and a greater proportion were classified as 'not confident' readers. This was a trend common to other Chinesespeaking education systems including Macao and Taiwan. Of the comparators, Singapore had the largest proportion of pupils who were classified as 'very confident' readers. Reading confidence was generally high among pupils in among Eastern-European education systems, with the greatest proportions of very confident readers in Bulgaria (61%), Kosovo (59%) and Serbia (58%).

7.2.1 Gender differences in confidence in reading

Table 25 breaks down pupils' confidence in reading by their gender. Girls and boys in England reported similar levels of confidence in reading, and the difference in PIRLS scores between the most and least confident pupils were similarly sized for girls and boys, both over 80 points. Internationally, almost every education system participating in PIRLS 2021 had higher proportions of girls categorised as being 'very confident' readers than boys. The International Median difference in the proportions of girls and boys being classified as 'very confident' readers was 4 percentage points, compared to 2 percentage points in England.

Table 25: Pupils' confidence in reading and their performance in PIRLS 2021 inEngland by gender

Confidence	Proportion of girls	Average score - girls	Proportion of boys	Average score - boys
Very confident	46%	598	44%	589
Somewhat confident	34%	546	34%	543
Not confident	20%	513	22%	496

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.2.2 Confidence in reading and access to books at home

Table 26 shows how pupils' confidence on the 'Confident in Reading' Scale in England is related to the number of books they reported having in their homes. As reported in section **6.5.2**, pupils in England who reported having more books at home had considerably higher scores in PIRLS than those reporting that their homes had fewer books. This is also reflected in their confidence in reading. Around two-thirds of the pupils who reported having more than 200 books at home were classified as 'very confident' readers, compared to just a quarter of pupils who reported having 10 or fewer books at home.

Table 26: Pupils' confidence in reading and their performance in PIRLS 2021 inEngland by the number of books at home

Number of books at home	Very confident	Average score - very confident	Not confident	Average score – not confident
0-10 books	25%	557	37%	477
11-25 books	35%	573	27%	505
26-100 books	46%	595	17%	517
101-200 books	57%	603	15%	525
More than 200 books	66%	617	13%	508

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.2.3 Confidence in reading, FSM eligibility, ethnic group and EAL status

This section describes confidence in reading and motivations to read by pupil characteristics such as eligibility for FSM at school, ethnic group, and EAL status. For further information on these pupil background characteristic variables, please refer to the relevant sections in Chapter **6**. For information on the number of pupils in each category for each of these pupil characteristics, please refer to **Table 2** in section **1.2.3**. The information reported in this section about pupil background characteristics uses data from England's NPD, which has been matched with the PIRLS 2021 data for England. As such, this section does not include information on comparator education systems.

Table 27 reports pupils' confidence in reading and their performance in PIRLS 2021 by their eligibility for FSM within the last 6 years. There is a positive relationship between confidence in reading and reading performance regardless of FSM eligibility. In other words, across both pupils who were and those who were not eligible for FSM in the last 6 years, pupils who are more confident in reading also tend to achieve higher scores in reading performance on average. The average performance of pupils eligible for FSM in the last 6 years who are 'very confident' in reading is lower than 'very confident' pupils who were not FSM-eligible at any point during that period (568 compared to 600). The range of performance across levels of confidence in reading for these groups, however, is similar. Pupils who were FSM-eligible within the last 6 years and who are 'very confident' achieve, on average, 82 points higher than their counterparts who were FSMeligible but 'not very confident'. For pupils who were not FSM eligible, the range is 89 points between those who are 'very confident' compared to 'not very confident'. There is a lower percentage of very confident readers in the FSM-eligible group (36%) than in the FSM-ineligible group (48%). This means that pupils who have been eligible for FSM are less likely to be 'very confident' in reading than those who have not been eligible for FSM.

Confidence in reading	Percent of pupils eligible for FSM in the past 6 years	Average PIRLS score for FSM- eligible pupils	Percent of pupils not eligible for FSM in the past 6 years	Average PIRLS score for FSM-non- eligible pupils
Very confident	36%	568	48%	600
Somewhat confident	35%	519	34%	553
Not very confident	29%	486	18%	511

Table 27: Pupils' confidence in reading and their performance in PIRLS 2021 byeligibility for free school meals (FSM) in the past 6 years

Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Table 28 shows pupils' confidence in reading and their range of performance in PIRLS 2021 by ethnic group. Across all major ethnic groups, the majority of pupils responded that they are either 'somewhat confident' or 'very confident' in reading, with between 13% and 21% of pupils expressing that they do not feel very confident in reading. Pupils from the 'Mixed' and 'Black' ethnic groups have the highest percentages of 'very confident'

readers, 55% and 54% respectively. Pupils from the 'White' ethnic group have the highest proportion of 'not very confident' readers (23%), while pupils from the 'Asian' group have the lowest percentage of 'not very confident' readers (13%). Due to low sample sizes (less than 50) in some ethnic group by confidence level combinations, the relationship between ethnic group, confidence in reading and PIRLS 2021 performance could not be reliably calculated for every group. For pupils from the 'White' and 'Asian' groups, where sample size was sufficient to establish the difference in PIRLS 2021 scores across confidence levels, it is evident that higher confidence levels are associated with higher achievement. For pupils from the 'White' group, 'very confident' readers score 92 points higher on average than their 'not very confident' peers. Meanwhile, pupils from the 'Asian' group who are 'very confident' in reading score 100 points higher on average than their 'not very confident' peers.

Table 28: Pupils' confidence in reading and their performance in PIRLS 2021 byethnic group

Pupil ethnic group	Very confident	Somewhat confident	Not very confident	Points difference
White	43%	34%	23%	92
Black	54%	32%	14%	n/a
Asian	49%	38%	13%	100
Mixed	55%	29%	16%	n/a
Other	39%	42%	19%	n/a

Some results may appear inconsistent due to rounding.

Points difference calculated as average score of 'very confident' readers minus average score of 'not very confident' readers. 'n/a' denotes a result not reported for ethnic groups where the estimate of points difference would be based on fewer than 50 pupils.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Table 29 shows the levels of confidence and average performance in PIRLS 2021 according to pupils' EAL status. The percent of pupils with different levels of confidence in reading is similar across both pupils with EAL and without. In both groups, 45% of pupils are 'very confident' in reading and, as was seen previously for other characteristics, confident readers tended to score higher than those who were 'not very confident'. For pupils both with and without EAL, the average achievement of 'not very confident' readers is just over 90 points lower than that of 'very confident' readers.

Table 29: Pupils' confidence in reading and their performance in PIRLS 2021 byEnglish as an Additional Language (EAL) status

Confidence in reading	Percent of pupils with EAL	Average PIRLS score for pupils with EAL	Percent of pupils without EAL	Average PIRLS score for pupils without EAL
Very confident	45%	594	45%	593
Somewhat confident	38%	534	33%	546
Not very confident	16%	500	22%	502

Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

7.2.4 Trends in confidence in reading

The 'Confidence in Reading' scale has changed slightly across PIRLS cycles, both in terms of the specific statements pupils are asked to respond to and in terms of how each response contributes to the overall classification of reading confidence. However, some statements have been used in multiple cycles, and it is possible to look at trends in responses to these.

Figure 25 shows how girls and boys in England responded to the 'Reading is easy for me' statement in the 2011, 2016 and 2021 cycles of PIRLS. For both girls and boys, a lower percentage of pupils in PIRLS 2021 strongly agreed that 'reading is easy' for them than in the previous 2 cycles. Also for both girls and boys, the combined percentage of pupils disagreeing (a little or a lot) that reading was easy for them increased to 15%. However, a greater proportion (6%) of boys disagreed 'a lot' that 'reading is easy' in PIRLS 2021, up from 3% in PIRLS 2016.




Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.3 Pupils' liking of reading

The '**Likes Reading**' Scale is based on pupils' responses to 8 statements in the Pupil Questionnaire. Pupils were asked the following:

"What do you think of reading? Tell how much you agree with each of these statements."

- I like talking about what I read with other people
- I would be happy if someone gave me a book as a present
- I think reading is boring
- I would like to have more time for reading
- I enjoy reading
- I learn a lot from reading
- I like to read things that make me think
- I like it when a book helps me imagine other worlds

Table 30 shows the proportions of pupils in England and each of the comparator education systems who 'very much like reading', 'somewhat like reading' or 'do not like reading'. The table also shows the difference in average PIRLS scores between pupils who 'very much like reading' and 'do not like reading' in each education system.

Table 30: Pupils' liking of reading in England and comparator systems in PIRLS2021

Education system	Very much likes reading	Somewhat likes reading	Does not like reading	Points Difference
England	29%	48%	24%	34
Singapore	33%	47%	20%	46
Hong Kong	30%	47%	23%	40
Australia	29%	45%	26%	45

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

Between comparators, there were not substantial differences in the proportions of pupils that were classified as 'very much' liking reading, with roughly one third of pupils in each of these education systems reporting the highest levels of enjoyment. Looking beyond the education systems to which direct statistical comparisons were possible, low levels of reading enjoyment were common in higher-performing education systems and culturally-similar English-speaking education systems more broadly. Ireland, Northern Ireland and the United States were all very similar to England and Australia in their proportions of pupils who very much, somewhat, and do not like reading, as well as in their relationships between liking of reading and PIRLS performance overall. Scandinavian education systems had some of the lowest proportions of pupils classified as 'very much' liking reading across all of the participating education systems, with Norway (13%) having the lowest proportion, followed by Denmark (14%). Sweden (18%) and Finland (23%) also had lower than average proportions of pupils who 'very much like' reading than England. By contrast, more than 80% of pupils in Kosovo and Uzbekistan were classified as 'very much' liking reading. The corresponding International Median was 46%, with almost all of

the education systems above this threshold having overall PIRLS scores below the International Median score of 520.

Although the trend *across* systems was that education systems with higher overall performance had lower proportions of pupils that reported 'very much' liking reading, *within* the vast majority of those education systems, pupils that enjoy reading more also scored higher in PIRLS 2021. In England, there was a 34 point average difference between pupils that 'very much' like reading and those that 'do not' like reading, a slightly smaller difference than in comparator education systems.

7.3.1 Gender differences in liking of reading

Table 31 breaks down pupils' liking of reading by their gender. In England, 32% of girls and 25% of boys were classified as 'very much' liking reading. Conversely, 19% of girls and 28% of boys were classified as pupils that 'do not' like reading. In every participating education system in PIRLS 2021, a higher proportion of girls reported the highest levels of reading enjoyment than boys. In England, 32% of girls and 25% of boys (a difference of 7 percentage points) 'very much liked' reading, a difference that was roughly in line with the International Median. There was a similar relationship between liking of reading and PIRLS performance between girls and boys; girls that do not like reading scored an average of 36 points lower than girls that very much like reading, while this difference was 30 points for boys.

Table 31: Pupils' liking of reading and their performance in PIRLS 2021 in Englandby gender

Liking of reading	Proportion of girls	Average score - girls	Proportion of boys	Average score - boys
Very much like reading	32%	574	25%	566
Somewhat like reading	48%	567	47%	556
Do not like reading	19%	538	28%	535

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.3.2 Liking of reading by access to books at home

Table 32 shows how pupils' enjoyment of reading in England is related to the number of books they reported to have in their homes. Perhaps unsurprisingly, pupils in England who reported liking reading the most also tended to report having more books at home. The relationship between the number of books at home, liking of reading, and PIRLS performance was more complex. Having more books at home was positively associated with PIRLS performance, as was liking of reading. Specifically for pupils with 10 or fewer books at home, however, pupils that reported not liking reading outperformed those that

very much like reading (those who somewhat like reading also outperformed those who very much like reading, but scored lower than those who 'did not like' reading). This seemingly counterintuitive finding should be interpreted with caution due to relatively low numbers of pupils responding this way. For pupils reporting 11 or more books at home, those who very much like reading outperform those that do not like reading, with the gap in average scores increasing as the number of books at home increases.

Table 32: Pupils' liking of reading and their performance in PIRLS 2021 in Englandby the number of books at home

Number of books at home	Very much likes reading	Average score - very much likes	Does not like reading	Average score – does not like
0-10 books	16%	479	46%	514
11-25 books	21%	542	29%	534
26-100 books	28%	571	21%	549
101-200 books	35%	589	15%	550
More than 200 books	51%	602	11%	552

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.3.3 Liking of reading by FSM eligibility, ethnic group and EAL status

Table 33 presents pupils' liking of reading and their PIRLS 2021 average scores by their FSM eligibility within the past 6 years. In the group of pupils who were FSM-eligible within that period, 31% reported that they 'do not like reading'. This is slightly higher than the percent of pupils who 'do not like reading' amongst those who were not FSM-eligible within the past 6 years (22%). In both groups, approximately one-third of pupils said that they 'very much like reading', which suggests that FSM-eligibility does not have a substantial influence on the proportion of pupils who 'very much like reading'. However, across all levels of reading enjoyment, the average PIRLS 2021 scores for pupils who were not FSM-eligible within the past 6 years is higher. Additionally, the range of average scores is narrower across levels of reading enjoyment for pupils who have been FSM eligible (13 points) than for those who have not been eligible (36 points).

Table 33: Pupils' liking of reading and their performance in PIRLS 2021 byeligibility for free school meals (FSM) in the past 6 years

Liking of reading	Percent of pupils eligible for FSM in the past 6 years	Average PIRLS score for FSM- eligible pupils	Percent of pupils not eligible for FSM in the past 6 years	Average PIRLS score for pupils not eligible for FSM
Very much likes	27%	533	29%	580
Somewhat likes	42%	530	50%	569
Does not like	31%	520	22%	544

Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Table 34 presents the different levels of reading enjoyment for pupils from each ethnic group. Also reported in the table is the difference in average performance in PIRLS 2021 for pupils who 'very much like reading' and those who 'do not like reading' for each ethnic group. The proportions of pupils in the 'Asian' group who 'very much like reading' is the highest (36%) in any ethnic group. Due to low counts (fewer than 50) in some ethnic group by liking of reading combinations, the relationship between ethnic group, liking of reading and PIRLS 2021 performance is not reported. For the 3 groups that could be reliably calculated based on large enough numbers of pupils, pupils who 'very much like reading.

Table 34: Pupils' liking of reading and their performance in PIRLS 2021 by ethnicgroup

Pupil ethnic group	Very much likes reading	Somewhat likes reading	Does not like reading	Points difference
White	27%	48%	25%	35
Black	29%	51%	19%	n/a
Asian	36%	48%	16%	18
Mixed	28%	44%	28%	39
Other	19%	48%	33%	n/a

Some results may appear inconsistent due to rounding.

Points difference calculated as average score of 'very much likes' readers minus average score of 'does not like' readers. 'n/a' denotes a result not reported for ethnic groups where the estimate of points difference would be based on fewer than 50 pupils.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Table 35 shows pupils' liking of reading and their average performance in PIRLS 2021 by their EAL status. Proportions of pupils who 'very much like reading' are similar across

those with and without EAL, 30% and 28% respectively. As was seen for other characteristics above, and generally in PIRLS 2021, pupils who like reading to some extent score higher, on average, than those who do not like reading. In considering PIRLS performance of pupils who 'very much like reading' across pupils with and without EAL, the pupils with EAL score 15 points lower on average than their counterparts without EAL. For pupils who 'somewhat like reading' as well as pupils who 'do not like reading', the differences in average PIRLS scores between the groups with and without EAL are small.

Table 35: Pupils' liking of reading and their performance in PIRLS 2021 by English
as an Additional Language (EAL) status

Liking of reading	Percent of pupils with EAL	Average PIRLS score for pupils with EAL	Percent of pupils without EAL	Average PIRLS score for pupils without EAL
Very much likes	30%	557	28%	572
Somewhat likes	49%	559	47%	561
Does not like	20%	540	25%	535

Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

7.3.4 Trends in liking of reading

Similar to the 'Confidence in Reading' Scale, changes in questionnaire items and scaling procedures mean that assessing trends on the 'Liking of Reading' Scale overall is not possible, but we can look at responses to individual statements contributing to the 'Liking of Reading' Scale where these have remained the same across several cycles.

Figure 26 shows how girls and boys in England responded to the statement, 'I enjoy reading' in the 2021, 2016, and 2011 cycles of PIRLS. Across all 3 of these cycles, smaller proportions of boys have agreed that they enjoy reading than girls, and around twice the proportion of boys has strongly disagreed. The proportions of both boys and girls who disagreed (a little or a lot) with the statement has increased since 2016 and 2011.



Figure 26: Percentages of girls and boys in England providing responses to the statement 'I enjoy reading' across the last 3 PIRLS cycles

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.4 Pupils' engagement in reading lessons

The '**Engaged in Reading Lessons**' Scale is based on of pupils' responses to 9 statements in the Pupil Questionnaire. Pupils were asked the following: "How much do you agree with these statements about your reading lessons?"

- I like what I read about in school
- My teacher gives me interesting things to read
- I know what my teacher expects me to do
- My teacher is easy to understand
- I am interested in what my teacher says
- My teacher encourages me to say what I think about what I have read
- My teacher lets me show what I have learned
- My teacher does a variety of things to help us learn
- My teacher tells me how to do better when I make a mistake

Table 36 shows the proportions of pupils in England and each of the comparator education systems classified 'very engaged, 'somewhat engaged' or 'less than engaged' in their reading lessons. The table also shows the difference in average PIRLS scores between 'very engaged' pupils and 'less than engaged' pupils in each education system.

 Table 36: Pupils' engagement in reading lessons in England and comparator

 systems relative to their performance in PIRLS 2021

Education system	Very engaged	Somewhat engaged	Less than engaged	Points Difference
England	54%	41%	5%	35
Singapore	48%	45%	7%	36
Hong Kong	39%	50%	11%	30
Australia	52%	42%	7%	35

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

In England, 54% of pupils were classified as being 'very engaged' in reading. This was higher than the other comparator systems, particularly compared to Hong Kong. Additionally, fewer pupils in England were classified as being 'less than engaged' in their reading lessons than the other comparators. However, England's figure of 54% was slightly below the International Median of 57%.

Looking beyond the selected comparator education systems, the 6 education systems with the highest levels of reading lesson engagement were all Eastern European, with 94% of pupils in Albania classified as 'very engaged' in their reading lessons. By contrast, Hong Kong's figure of 39% was only greater than Denmark's (38%). Hong Kong also had the highest proportion of pupils who were classified as 'less than engaged' in their reading across all participating education systems in PIRLS 2021.

Higher engagement in reading lessons was associated with stronger overall reading performance in most education systems, including England. The difference in points in England between the most and least engaged pupils was roughly in line with the international average.

7.4.1 Gender differences in engagement in reading lessons

Table 37 breaks down pupils' engagement in reading lessons by gender. A slightly higher proportion of girls in England were classified as being 'very engaged' in their reading lessons than boys, and a slightly larger proportion of boys were classified as 'less than engaged'. The gender gap in PIRLS performance was relatively stable across pupils with differing levels of reading lesson engagement, and the gap (of about 34 points) between pupils who were 'very engaged' and those who were 'less than engaged' was consistent across boys and girls.

Table 37: Pupils' engagement in reading lessons and their performance in PIRLS2021 in England by gender

Engagement in reading lessons	Proportion of girls	Average score - girls	Proportion of boys	Average score - boys
Very engaged	56%	567	52%	558
Somewhat engaged	40%	561	42%	551
Less than engaged	4%	533	6%	524

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.4.2 Engagement in reading lessons by access to books at home

Table 38 shows the relationship between the number of books pupils reported having at home and their engagement in reading lessons. In general, higher levels of engagement were associated with higher average scores for overall reading performance, but the group reporting 0 to 10 books at home deviated from this pattern slightly. Although 'less than engaged' pupils had lower scores on average than their 'very engaged' counterparts with few or no books at home, their peers who were 'somewhat engaged' were actually the highest-scoring on average. It is difficult to know exactly why this might have been the case.

Table 38: Pupils' engagement in reading lessons and their performance in PIRLS
2021 in England by number of books at home

Number of books at home	Very engaged	Average score - very engaged	Less than engaged	Average score – less than engaged
0-10 books	48%	504	7%	472
11-25 books	50%	545	5%	527
26-100 books	54%	569	5%	532
101-200 books	59%	584	4%	552
More than 200 books	58%	594	5%	574

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.4.3 Engagement in reading lessons by FSM eligibility, ethnic group and EAL status

Table 39 presents the range of pupils' engagement in reading lessons according to their eligibility for FSM within the last 6 years, as well as the average PIRLS 2021 score associated with each category. Across both groups, the vast majority of pupils are engaged in reading lessons to some extent. Furthermore, the proportions of pupils with different levels of engagement are very similar for both groups. However, pupils who had been FSM eligible scored lower on average than those who had not been FSM eligible across all levels of engagement. Pupils who are 'very engaged' in reading lessons and have been FSM eligible have an average score of 532 in PIRLS 2021, while their peers who have not been FSM eligible have an average score of 571. The differences in PIRLS achievement across the levels of engagement is wider for pupils who are FSM eligible (51 points) than for those who are not FSM-eligible (31 points).

Table 39: Pupils' engagement in reading lessons and their performance in PIRLS2021 by eligibility for free school meals (FSM) in the past 6 years

Engagement in reading lessons	Percent of pupils eligible for FSM in the past 6 years	Average PIRLS score for FSM- eligible pupils	Percent of pupils not eligible for FSM in the past 6 years	Average PIRLS score for pupils not eligible for FSM
Very engaged	50%	532	54%	571
Somewhat engaged	44%	530	42%	565
Less than engaged	6%	481	5%	540

Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Table 40 presents the levels of pupil engagement in reading lessons according to ethnic group. As is apparent in the table, the proportions of pupils at each of the 3 levels of engagement are similar across all ethnic groups. This means that there is no strong association between pupils' engagement in reading lessons and their ethnic groups. Due to small sample sizes (fewer than 50) in most groups, the points difference between the PIRLS scores of pupils who were 'very engaged' and 'less than engaged' could only be calculated for pupils in the 'White' ethnic group. Within the 'White' group, 'very engaged pupils scored about 47 points higher than 'less than engaged' pupils.

Table 40: Pupils' engagement in reading lessons and their performance in PIRLS
2021 by ethnic group

Pupil ethnic group	Very engaged	Somewhat engaged	Less than engaged	Points difference
White	53%	42%	5%	47
Black	51%	41%	8%	n/a
Asian	56%	39%	5%	n/a
Mixed	49%	45%	6%	n/a
Other	37%	53%	9%	n/a

Some results may appear inconsistent due to rounding.

Points difference calculated as average score of 'very engaged' pupils minus average score of 'less than engaged' pupils. 'n/a' denotes a result not reported for ethnic groups where the estimate of points difference would be based on fewer than 50 pupils.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Table 41 presents pupils' engagement in their reading lessons and their average PIRLS 2021 scores by EAL status. The proportion of pupils reporting each level of engagement in reading lessons is very similar across pupils with and without EAL. In both groups, only 5% of pupils are 'less than engaged' in reading lessons, indicating that engagement is generally fairly high for the vast majority of pupils. There was not a consistent relationship between average PIRLS score and engagement across pupils with and without EAL. Amongst the pupils without EAL, pupils with higher engagement tended to have higher performance in PIRLS on average, but this pattern was not evident for pupils with EAL.

Table 41: Pupils' engagement in reading lessons and their performance in PIRLS2021 by English as an Additional Language (EAL) status

Engagement in reading lessons	Percent of pupils with EAL	Average PIRLS score for pupils with EAL	Percent of pupils without EAL	Average PIRLS score for pupils without EAL	
Very engaged	53%	555	52%	563	
Somewhat engaged	42%	557	42%	555	
Less than engaged	5%	n/a	5%	519	

Some results may appear inconsistent due to rounding.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

7.4.4 Trends in engagement in reading lessons

For reasons similar to those explained above for the 'Confident in Reading' and 'Liking of Reading' Scales, we focus on the trend over time not for the overall 'Engaged in Reading Lessons' Scale but for one of the items contributing to it, as this item was included for multiple cycles of PIRLS. The item asked pupils to respond to the statement, 'my teacher gives me interesting things to read'. **Figure 27** shows the proportion of girls and boys in England providing each response to this question in PIRLS 2011, PIRLS 2016, and PIRLS 2021.

Responses to this question in England have become slightly more positive over time, particularly for boys. While 26% of boys either slightly or strongly disagreed with this statement in PIRLS 2011, this figure is only 18% in PIRLS 2021, and there has been a 9% increase over the same period in the proportion of boys who 'agree a lot'. Change over time in girls' responses has been more modest, with only a small increase across cycles in the proportion of girls who agree strongly and little difference in the proportions of girls who disagree that their teacher gives them interesting things to read. The proportions of girls and boys who agree that their teacher gives them interesting things to read is roughly the same between girls and boys in England in PIRLS 2021.

Figure 27: Percentages of girls and boys in England providing responses to the statement 'My teacher gives me interesting things to read' across the last 3 PIRLS cycles



Agree a lot Agree a little Disagree a little Disagree a lot						
Gender and Cycle	Agree a lot	Agree a little	Disagree a little	Disagree a lot		
Girls - 2021	46%	37%	13%	4%		
Girls - 2016	45%	38%	13%	3%		
Girls - 2011	41%	42%	12%	4%		
Boys - 2021	44%	37%	11%	7%		
Boys - 2016	40%	39%	15%	7%		
Boys – 2011	35%	39%	17%	9%		

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.5 Time spent reading outside of school

The PIRLS questionnaire completed by pupils also included questions about pupils' reading behaviours. This included asking pupils about how much time they spend reading outside of school, specifically:

"How much time do you spend reading outside of school on a normal school day?"

Pupils were instructed to choose between 4 response categories: "Less than 30 minutes", "30 minutes up to 1 hour", "From 1 hour up to 2 hours" or "2 hours or more".

Table 42 shows the proportions of pupils in England and comparator education systems reporting that they spend different amounts of time each day reading outside of school.

Table 42: Pupils' time spent reading outside of school each day in England andcomparator systems

Education system	system Two hours One hour to or more two hours		30 minutes to an hour	Less than 30 minutes
England	7%	9%	35%	49%
Singapore	13%	15%	35%	37%
Hong Kong	9%	14%	37%	40%
Australia	9%	9%	32%	50%

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

Almost half of pupils in England reported that they spend less than 30 minutes each day reading outside of school. This was slightly higher than the International Median of 44%, but similar to Australia. Outside of the direct comparators included in **Table 42**, Bahrain, Finland, Italy, Macao SAR, Oman, New Zealand, the United States and Portugal all had a similar proportion of pupils who reported spending less than 30 minutes a day reading outside of school. A smaller proportion of pupils in England also reported that they read for two hours or more each day than was true in comparator education systems. Of the comparator systems, Singapore had the highest proportion of pupils reporting that they read for at least two hours outside of school each day, and the lowest proportion reading for less than 30 minutes.

Beyond the education systems included in direct statistical comparisons in this report, other high-performing systems where pupils reported higher-than-average amounts of out-of-school reading included Germany and Taiwan. The 4 education systems with the lowest proportion of pupils reporting spending two or more hours each day reading outside of school were all Scandinavian, including Finland (4%), Sweden (3%), Norway (3%) and Denmark (2%). Amongst other English-speaking education systems, the

proportions of pupils reporting spending two or more hours each day reading outside of school varied somewhat, from fairly low in the United States (6%) and Northern Ireland (6%), slightly higher in Ireland (8%) and Australia (9%), to relatively high in New Zealand (11%).

The International Median score for pupils who read for 2 hours or more each day was 532, compared to 506 for pupils that read for less than 30 minutes each day. This compares to scores of 580 and 536, respectively, for pupils in England.

Focusing on England, **Figure 28** shows how the amount of time spent reading outside of school has changed over the 2011, 2016, and 2021 cycles of PIRLS for girls and boys. Across all 3 cycles, boys in England have reported less time reading spent outside of school than girls. However, this gap has been narrowing because of considerably larger reductions in the time spent reading by girls. While only 36% of girls reported spending less than 30 minutes a day reading in 2011, this figure stands at 47% for PIRLS 2021. Boys showed a more modest increase in the proportion reporting that they read for less than 30 minutes a day, from 49% to 52% over the same time period.



Figure 28: Percentages of girls and boys in England reporting different amounts of time they spend reading outside of school each day

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

7.6 Contextualisation: The relationship between reading enjoyment and reading achievement in Norway

There is well established evidence to suggest that there is a positive relationship between reading enjoyment and reading achievement. In other words, children who like reading typically score higher on tests of reading achievement than those who do not like reading. Findings from PIRLS have added to the body of research exploring this relationship by showing that pupils with positive attitudes towards reading have higher overall achievement. The relationship between reading attitudes and achievement, however, is generally a reciprocal one and not causal (Toste et al., 2020). Added to this, the number of influential factors in what contributes to positive (or negative) attitudes towards reading means that reading enjoyment can vary considerably both across and within contexts (Rogiers et al., 2020). **Box 7.1** explores the relationship between enjoyment of reading and reading achievement in an education system with high achievement, but low percentages of pupils who enjoy reading – Norway.

Box 7.1 Norway: High reading achievement, but low enjoyment

The relationship between reading attitudes and achievement is well recognised across a range of studies and assessments, however, there are large variations of the rate at which pupils from both high and low performing education systems report enjoying reading. In some of the high performing education systems in PIRLS, there are substantial numbers of pupils reporting that they do not enjoy reading. Norway, for instance, has high overall achievement in PIRLS (539), but only 59% of pupils report that they like reading, the lowest percentage across all participating education systems. South Africa, on the other hand, is the lowest performing education system in PIRLS in terms of average achievement (288) but 90% of South African pupils report that they like reading.

It is important to note that in both education systems the positive relationship between achievement and enjoying reading is still clear. In Norway, pupils who say they like reading 'very much' have higher average PIRLS scores (556) than those who report 'somewhat' liking reading (547), and those who 'do not' like reading have the lowest average PIRLS scores (528). Similar findings for Norway are also seen in other international tests of reading such as PISA, as well as national tests of reading (Vestheim et al., 2019).

Continues on next page

Box 7.1 (continued)

Results showing that Norwegian pupils have low motivation to read, such as those from PISA and PIRLS, have led to policy changes that focus on improving reading skills as well as improving motivation in school (Wagner & Støle, 2022). Research focused on understanding the factors that play a role in attitudes towards reading is also prevalent in Norway, and findings indicate that home-related factors such as socio-economic status and parental attitudes towards reading are influential (Çaliskan & Ulas, 2022; Frønes et al., 2020; Støle et al., 2020). The children of parents who report that they enjoy reading typically also have positive attitudes toward reading, while children from homes with lower socio-economic status and fewer books in the home, have less positive attitudes towards reading.

The relationship between reading attitudes and gender has also received considerable attention in Norway. In PIRLS 2021, the gap between boys and girls who 'very much like reading' is less substantial than for the majority of education systems; 15% of girls reported that they very much like reading while 12% of boys reported the same level of enjoyment. Researchers have also been exploring different consequences of reading assessments conducted on screen versus on paper compared to reading on paper (Dahan et al., 2018; Engdal Jensen, 2020; Kong et al., 2018). Findings revealed that although children often report that they prefer reading on screens, their reading comprehension can be negatively affected if the mode of the test is digital, and the consequences are more severe for boys than for girls.

8 School environment and teacher characteristics

Chapter overview

This chapter explores how school environment and teacher characteristics relate to reading performance in PIRLS 2021. Three areas are examined: Characteristics and practices of the teacher, characteristics and climate of the school, and extent of parental involvement in pupils' school experiences. Teacher characteristics include teachers' qualifications, years of experience, and overall career satisfaction as well as teaching practices such as strategies for teaching reading and use of instructional materials in reading lessons. School climate includes factors such as emphasis on academic success, perceptions of safety and discipline at the school, and pupils' reports of bullying at school. Parental involvement is considered in terms of involvement in the school and in children's learning. The chapter concludes with a discussion of the emphasis on academic success and achievement in the United Arab Emirates.

Key findings

- In England, a majority of the pupils who participated in PIRLS 2021 had teachers who reported having a postgraduate degree or higher (including PGCEs). There is no clear relationship between the average performance of pupils in PIRLS 2021 and the formal education levels of their teachers.
- Pupils in England who participated in PIRLS 2021 were taught by teachers with an average of 11.5 years teaching experience. For England and comparators, there is no clear association observed between years of teaching experience and PIRLS performance.
- Just under half of pupils' teachers in England report being 'very satisfied' in their careers. This is lower than the International Median. However, there is no strong relationship between teacher career satisfaction and pupils' overall performance in PIRLS 2021 across England or across the comparator systems.
- In England, most pupils' teachers report that their schools place a high emphasis on academic success. In every education system in PIRLS 2021, the performance of pupils whose teachers reported that their schools place a 'very high' emphasis on academic success tends to be higher on average.

Continues on next page

Chapter overview (continued)

- The association between headteachers reports of school discipline and PIRLS achievement across all participating education systems shows that on average, pupils in schools with hardly any problems with discipline score 26 points higher in PIRLS than those in schools with moderate to severe problems. In England, this range is even higher, pupils in schools with hardly any problems have an average PIRLS score 50 points higher than those in schools with moderate to severe problems.
- Almost half of pupils who participated in PIRLS 2021 in England report experiencing bullying at least monthly at school. Across all participating education systems, pupils who report experiencing bullying more frequently had lower average performance in PIRLS 2021.
- Reports of parental commitment and support of pupil achievement in England are lower than comparators and low internationally. Generally, pupils in schools where reports of parental commitment and support are high, have higher average achievement in PIRLS 2021.

8.1 Teacher characteristics and teaching practices

To provide context for PIRLS 2021 results, teachers of pupils who participated in PIRLS completed a questionnaire related to their backgrounds, their training and professional development, and their teaching practice. In order to present nationally representative information, the data presented throughout this chapter pertains to percentages of pupils whose teachers answered in a particular category on this questionnaire, and not proportions of teachers themselves.

8.1.1 Teacher qualifications and professional development

Table 43 presents the percentages of pupils whose teachers reached or exceeded different levels of education for England and comparators. In England, a majority (62%) of the pupils who participated in PIRLS 2021 had teachers who reported having a postgraduate degree or higher. This is a change from previous cycles where only 8% of pupils' teachers reported having a postgraduate degree. This change is likely to due to an adaptation made to this question on the PIRLS 2021 teacher questionnaire for England. The Post Graduate Certificate in Education (PGCE) is the typical route into teaching for teachers in England and is considered equivalent to a postgraduate degree. Previously, the low percentage of teachers reporting that they had a postgraduate degree in PIRLS questionnaires was thought to be attributable to teachers not interpreting their PGCE qualification as a postgraduate degree. To account for this, 'PGCE' was

specifically added as an option in the postgraduate degree category in the PIRLS 2021 questionnaires for England. This inclusion of PGCE was made for England's questionnaires but not all participating education systems, added to this teacher education routes vary considerably across education systems. This means that comparisons between education levels of teachers across different education systems should be interpreted cautiously.

In England, Hong Kong, and Australia, almost all pupils (more than 98%) are taught by teachers who report having a bachelor's degree or higher. In Singapore this percentage is lower (88%). There is no clear relationship between the average performance of pupils in PIRLS 2021 and the formal education levels of their teachers. It is important to note that the inclusion of PGCE was made for England's questionnaires, but not for the questionnaires of other participating education systems, furthermore teacher education routes vary considerably across contexts. This means that comparisons between education levels of teachers across different education systems should be interpreted cautiously.

 Table 43: Percentage of pupils in England and comparator systems whose teachers have reached or exceeded different education levels (2021)

Education system	e system Postgraduate Degree or higher * degree		Post- secondary or short-cycle tertiary	Upper secondary qualification or lower **
England	62%	36%	1%	<1%
Singapore	15%	73%	11%	0%
Hong Kong SAR	36%	63%	<1%	<1%
Australia	17%	81%	2%	<1%

* Postgraduate degrees or higher include PGCE qualifications in England

** Upper secondary qualifications or lower include A-level qualifications in England Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 44 presents the percentage of pupils whose teachers report that their formal training placed an emphasis on language, teaching of reading, and reading theory. Internationally, the aspect of reading education most emphasised in formal training is language, where the International Median reveals that 78% of pupils' teachers reported it as an area of emphasis. In England, 79% of pupils' teachers report that language is an area of emphasis in their training, a slight increase from 74% in 2016. Of the comparators, Singapore had the highest percentage of pupils whose teachers reported language as an area of emphasis (87%).

Reports of pedagogy and/or teaching reading as an area of emphasis in England have increased from previous cycles. In PIRLS 2011, just under half of pupils' teachers (48%) reported that pedagogy was an area of emphasis in their formal training. This increased to 65% in PIRLS 2016 and now 71% in the 2021 cycle. The International Median for pedagogy and/or teaching reading as an area of emphasis is 66%, equal to Hong Kong's average but slightly lower than all other comparators in this area.

Education system	Average PIRLS Score	Language	Pedagogy / Teaching Reading	Reading Theory
England	558	79%	71%	30%
Singapore	587	87%	80%	31%
Hong Kong SAR	573	80%	65%	23%
Australia	540	72%	78%	39%

Table 44: Percentage of pupils in England and comparator systems whose teachers' formal training emphasised different aspects of reading education (2021)

Source: IEA's PIRLS 2021

Across England and comparators, reading theory is the aspect of reading education that the lowest percentage of pupils' teachers reported as an area of emphasis. In England, 30% of pupils' teachers said that it was emphasised, which is an increase from 16% in PIRLS 2016. Hong Kong had the lowest percentage reporting that reading theory was an area of emphasis (23%) among the comparators. Australia had the highest percentage of reports that reading theory was emphasised (39%). The International Median for reading theory as an area of emphasis in formal training is 35%. In England and internationally, there is no relationship evident between different aspects of reading education emphasised in formal training and pupils' overall performance in PIRLS 2021.

Table 45 shows the years of teaching experience that pupils' teachers have in England and comparators, according to 4 categories: less than 5 years, 5-9 years, 10-19 years, 20 years or more. In England, pupils who participated in PIRLS 2021 are taught by teachers with an average of 11.5 years teaching experience, which is the lowest across all comparator systems. England is among just 5 participating education systems where the average number of years of teaching experience is less than 12 years, the others being the United Arab Emirates, Qatar, Jordan, and Bahrain. Internationally, pupils who participated in PIRLS 2021 are taught by teachers who have an average of 18 years of teaching experience. Consistent with the findings in PIRLS 2016, for many education systems, including England and all of the comparator systems, there is no clear association observed between teaching experience and PIRLS performance; while pupils in England whose teachers have 20 years or more had slightly higher average PIRLS performance than those in other categories, the next highest-performing group were pupils whose teachers have less than 5 years' experience.

Education system	20 years or more	10-19 years	5-9 years	Less than 5 years
England	17%	31%	30%	22%
Singapore	35%	30%	20%	15%
Hong Kong SAR	42%	21%	16%	21%
Australia	27%	27%	33%	13%

Table 45: Percentage of pupils in England and comparator systems whoseteachers have different amounts of teaching experience (2021)

Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 46 presents information about the reading-related professional development (e.g., workshops, seminars) that teachers of pupils who participated in PIRLS have received in the last 2 years. In England, 65% of pupils' teachers have participated in professional development related to teaching reading comprehension skills or strategies. This is lower than the percentage of pupils' teachers across comparator education systems but slightly higher than the International Median (60%) for the area of reading comprehension. Participation in professional development related to integrating literacies across the curriculum was similar but slightly lower than the other 2 areas, the International Median was 48% and just over half of pupils' teachers in England (52%) and Australia (57%) reporting recent professional development related to reading comprehension was higher than professional development related to integrating literacies. There was no relationship observed between pupils' performance in PIRLS 2021 and their teachers' participation in different areas of reading-related professional development.

Table 46: Percentage of pupils in England and comparator systems whoseteachers have participated in different areas of reading-related professionaldevelopment (2021)

Education system	Teaching reading comprehension skills or strategies	Integrating literacies across the curriculum
England	65%	52%
Singapore	71%	29%
Hong Kong SAR	82%	40%
Australia	76%	57%

Source: IEA's PIRLS 2021

8.1.1 Teacher career satisfaction

The PIRLS 2021 scale for teacher job satisfaction is determined based on teachers' responses to the question:

"How often do you feel the following way about being a teacher?"

- I am content with my profession as a teacher
- I find my work full of meaning and purpose
- I am enthusiastic about my job
- My work inspires me
- I am proud of the work I do
- I feel appreciated as a teacher

The teacher job satisfaction scale was modified for the PIRLS 2021 cycle to include how often teachers feel appreciated. This means that it is not possible to consider trends in teacher job satisfaction across cycles.

Table 47 presents the percentages of pupils whose teachers reported different levels of job satisfaction in England and the comparator systems. Teacher job satisfaction is classified into 3 categories: 'very satisfied', 'somewhat satisfied', and 'less than satisfied'.

Education system	Very satisfied	Somewhat satisfied	Less than satisfied
England	44%	46%	10%
Singapore	42%	40%	18%
Hong Kong SAR	37%	45%	18%
Australia	49%	41%	10%

Table 47: Percentage of pupils in England and comparator systems whoseteachers report different levels of career satisfaction (2021)

Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

In England, 44% of teachers reported being 'very satisfied' in their careers. This is lower than the International Median of 56%. Similarly, all comparators have lower proportions of pupils with 'very satisfied' teachers than the International Median, with Hong Kong having the lowest proportion (37%). There is no strong relationship between teacher career satisfaction and pupils' overall performance in PIRLS 2021 in England or in any of the comparators. In other words, pupils of teachers who report different levels of career satisfaction do not have substantially different average achievement in PIRLS 2021.

However, when considering pupils' teachers across all education systems, in education systems in which a very high proportion of pupils' teachers say they are 'very satisfied' in their careers, lower overall average performance in PIRLS 2021 is sometimes seen. For

example, more than 93% of pupils' teachers in Kosovo and Oman are 'very satisfied', but both have an overall average for PIRLS 2021 below 430. By contrast, in Poland, which is among the higher-performing education systems in PIRLS 2021 with a score of 549, the lowest proportion of pupils' teachers (27%) report being 'very satisfied'.

8.1.2 Strategies for teaching reading

PIRLS 2021 asks teachers about the tasks they assign in reading instruction to help pupils develop reading comprehension skills and strategies. Teachers report how frequently they ask pupils to perform each of the following 12 tasks:

- A. Locate information within the text
- B. Identify the main ideas of what they have read
- C. Explain or support their understanding with text evidence
- D. Compare what they have read with experiences they have had
- E. Compare what they have read with other things they have read
- F. Make predictions about what will happen next in the text they are reading
- G. Make generalisations and draw inferences based on what they have read
- H. Evaluate and critique the style or structure of the text they have read
- I. Determine the author's perspective or intention
- J. Self-monitor their reading (e.g., recognise when they don't understand)
- K. Determine if a website is useful for a specific purpose
- L. Evaluate the credibility of a website

The final 3 tasks in this list (J-L), which focus on self-monitoring and digital literacy strategies, are new to PIRLS 2021 and so comparisons with responses from previous cycles are not possible.

Table 48 and Table 49 present the percentages of pupils whose teachers ask them to perform each task at least once a week. In England, almost all pupils are asked to locate information in texts, identify the main ideas of what they have read, and to explain their understanding of what they read at least once a week. These 3 tasks are among the most frequently applied across all comparators as well as internationally. About 64% of pupils in England are asked to perform tasks associated with evaluating the style and structure of texts weekly, which is similar to the International Median (67%) as well as Hong Kong (68%) and Australia (67%). However, in Singapore less than half of pupils (43%) are asked to evaluate style and structure on a weekly basis.

When compared to the International Median (87%), pupils in England are asked to compare their reading to personal experiences slightly less frequently (78%). When considering digital literacy tasks, such as understanding the purpose of a website, it is less common for these to be assigned to pupils in England (29%) when compared to the

International Median (40%). Digital literacy tasks are most frequently assigned to pupils in Kosovo, Uzbekistan, and North Macedonia.

Table 48: Percent of pupils in England and comparator systems whose teachers ask them to complete at least weekly tasks requiring reading skills A-F (2021)

Education system	Α	В	С	D	E	F
England	99%	97%	99%	78%	70%	92%
Singapore	91%	87%	90%	80%	70%	83%
Hong Kong SAR	95%	98%	91%	78%	66%	79%
Australia	97%	95%	92%	90%	84%	94%

A - Locate information within the text.

B - Identify the main ideas of what they have read

C - Explain or support their understanding with text evidence

D - Compare what they have read with experiences they have had

E - Compare what they have read with other things they have read

F - Make predictions about what will happen next in the text they are reading

Source: IEA's PIRLS 2021

Table 49: Percent of pupils in England and comparator systems whose teachersask them to complete at least weekly tasks requiring reading skills G-L (2021)

Education system	G	Н	Ι	J	К	L
England	98%	64%	68%	80%	29%	20%
Singapore	79%	43%	49%	62%	20%	16%
Hong Kong SAR	86%	68%	88%	53%	19%	14%
Australia	93%	67%	78%	88%	42%	34%

G - Make generalisations and draw inferences based on what they have read

H - Evaluate and critique the style or structure of the text they have read

I - Determine the author's perspective or intention

J - Self-monitor their reading (e.g., recognize when they don't understand)

K - Determine if a website is useful for a specific purpose

L - Evaluate the credibility of a website

Source: IEA's PIRLS 2021

8.1.3 Use of instructional materials

Teachers of pupils who participated in PIRLS 2021 were asked about the types of instructional materials they use regularly during reading lessons. The teachers answered questions about the frequency with which they assigned the following 8 text types related to both literary and informational texts:

Literary Reading Materials

- A. Short stories (e.g., fables, fairy tales, action stories, science fiction, detective stories)
- B. Longer fiction books with chapters

- C. Plays
- D. Poems/poetry

Informational Reading Materials

- E. Nonfiction subject area books or textbooks
- F. Longer nonfiction books with chapters
- G. Nonfiction articles that describe and explain about things, people, events, or how things work (e.g., newspaper articles, brochures)
- H. Non-continuous texts (e.g., diagrams, maps, illustrations, photographs, tables)

Table 50 presents the percentages of pupils whose teachers assign each of these text types at least once a week. In England, longer books (both literary and informational) are slightly more frequently assigned (82% for literary) than the International Median (33% for literary). This is also evident in Australia, where 86% of pupils are assigned longer fiction books by their teachers at least once a week, whereas only 11% of pupils in Hong Kong are assigned longer fiction books at least weekly. Across all comparators, it is relatively uncommon for plays to be assigned to pupils at least weekly, and this is true across all participating education systems with an International Median of 7%.

Notably, Hong Kong has the highest relative performance in the Informational Purpose Scale compared to the Literary Purpose Scale of the comparators. However, the frequency at which pupils' teachers in Hong Kong report assigning informational texts is the lowest of the comparators and lower than the International Median for all informational text types. The International Medians show that across all education systems, the most frequently assigned informational text-type is non-fiction subject area books (e.g., textbooks) (70%), and the least frequently assigned informational text-type is longer non-fiction books (23%).

Education system	Α	В	С	D	Е	F	G	Н
England	62%	82%	4%	10%	58%	32%	37%	38%
Singapore	71%	32%	2%	2%	56%	17%	54%	32%
Hong Kong SAR	55%	11%	1%	7%	41%	10%	28%	23%
Australia	87%	86%	15%	17%	83%	43%	60%	66%

Table 50: Percent of pupils in England and comparator systems whose teachersassign different instructional materials at least weekly for reading (2021)

A - Short stories

B - Longer fiction books with chapters

C – Plays

E - Nonfiction subject area books

F - Longer nonfiction books with chapters

G - Nonfiction articles

H - Non-continuous texts

Source: IEA's PIRLS 2021

D - Poems/poetry

Table 51 presents information from the pupil questionnaire related to pupils' interest in the reading materials assigned by their teachers. Pupils were asked the extent to which they agree with the statement 'My teacher gives me interesting things to read'. Across England and comparators, a majority of pupils agree that their teachers give them interesting things to read. There is a negative relationship between pupils' who disagree a lot with the statement, and their overall performance in PIRLS 2021. In other words, pupils who do not find their assigned reading materials interesting. For more detailed information on the relationship between pupils' motivations and attitudes to reading and their performance in PIRLS, see Chapter 7.

Table 51: Percentage of pupils in England and comparator systems who reportedthat their teacher gives them interesting things to read

Education system	Agree a lot	Agree a little	Disagree a little	Disagree a lot
England	45%	37%	12%	5%
Singapore	49%	35%	12%	5%
Hong Kong SAR	41%	40%	13%	6%
Australia	43%	37%	13%	6%

Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

8.2 School characteristics

8.2.1 Performance by school type

As discussed in section **1.2.2**, when conducting the sampling in England for PIRLS 2021, 2 main criteria were used. One of these was to ensure that there was a representative range of pupils from schools with different levels of historic performance in key stage 2 examinations. Five categories of key stage 2 performance were created using this approach, and this informed the school-level sampling for state-maintained schools and academies. These were labelled as 'low-performing', 'mid-low performing' 'middle performing', 'mid-high performing' and 'high performing'. The remaining schools in England's sample were independent schools. **Table 52** shows the distribution of PIRLS scores for pupils in England's PIRLS 2021 sample by the type of school they attend. The table shows that pupils' average performance in PIRLS was positively associated with the school's historic level of performance, with the average pupil at a low-performing school. However, across all school types, there were relatively wide ranges in PIRLS performance between the 10th and 90th percentiles. Pupils at independent schools scored, on average, higher than pupils in other schools.

Table 52: Performance of England's PIRLS 2021 cohort by the type of school theyattend

School-type	Percentage of pupils	10th percentile	50th percentile	90th percentile	Range
Low performing	14.0	430	542	627	197
Mid-low performing	19.6	449	550	636	187
Middle performing	24.0	460	559	644	184
Mid-high performing	20.4	464	571	658	194
High performing	17.9	473	574	660	187
Independent	4.3	516	606	693	177

Range calculated as 90th percentile – 10th percentile.

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

8.2.2 School climate

Alongside the teacher questionnaire, the contextual framework for PIRLS 2021 also includes a school questionnaire where headteachers respond to questions related to the school climate. Both questionnaires ask headteachers and teachers to respond to questions about the extent to which the school emphasises academic success, levels of safety at school, and problems associated with school discipline. The pupil questionnaire also asks questions related to the school climate such as how frequently pupils experience bullying at school. This section focuses on 4 of the scales related to school climate as reported by teachers, headteachers and pupils.

The first of these scales, school emphasis on academic success, was calculated based on teachers' responses to the following question:

"How would you characterise each of the following within your school?"

- Teachers' understanding of the school's curricular goals
- Teachers' degree of success in implementing the school's curriculum
- Teachers' expectations for pupil achievement
- Teachers' ability to inspire pupils
- Collaboration between school leadership and teachers to plan instruction
- Parental involvement in school activities
- Parental commitment to ensure that pupils are ready to learn
- Parental expectations for pupil achievement
- Parental support for pupil achievement
- Pupils' desire to do well in school

- Pupils' ability to reach school's academic goals
- Pupils' respect for classmates who excel academically

Table 53 presents information related to the extent to which participating pupils' teachers believe that their school emphasises academic success for England and comparators. Across all education systems that participated in PIRLS 2021, the emphasis on academic success reported was typically high and so responses were categorised as either 'very high', 'high', or 'medium'.

In England, most pupils' teachers (61%) report that their school places a 'high' emphasis on academic success. This is more than the other comparators and also higher than the International Median (54%). Across all comparators there are fewer reports of schools that place a 'very high' emphasis on academic success. In England and Australia, 9% of pupils' teachers report that there is 'very high' emphasis, while in Hong Kong just 3% of schools report a 'very high' emphasis. Across participating education systems, the average performance of pupils whose teachers reported their schools have a 'very high' emphasis on academic success tends to be higher than those reported to have a 'high', and the lowest average performance was generally found in schools with a 'medium' emphasis on academic success. In England, pupils in schools with a 'very high' emphasis on academic success scored, on average, 37 points higher in PIRLS 2021 than schools where a 'medium' emphasis was reported. The difference in achievement at the International Median is 26 points in favour of pupils in schools with a 'very high' emphasis on academic success compared to those with a 'medium' emphasis.

Table 53: Percentage of pupils in England and comparator systems whose teachers believe different levels of emphasis are placed on academic success at the school

Education system	Very high emphasis	High emphasis	Medium emphasis
England	9%	61%	31%
Singapore	6%	42%	52%
Hong Kong SAR	3%	49%	48%
Australia	9%	48%	44%

Because of rounding some results may appear inconsistent.

Source: IEA's PIRLS 2021

Teachers were also asked about the levels of safety and orderliness at their schools. The scale for school safety and orderliness was calculated based on teachers' responses to the following question:

"Thinking about your current school, indicate the extent to which you agree or disagree with each of the following statements?"

- This school is located in a safe neighbourhood
- I feel safe at this school
- This school's security policies and practices are sufficient
- The pupils behave in an orderly manner
- The pupils are respectful of the teachers
- The pupils respect school property
- This school has clear rules about pupil conduct
- This school's rules are enforced in a fair and consistent manner
- The pupils are respectful of each other

The reports of school safety and orderliness are categorised: 'very safe and orderly', 'somewhat safe and orderly', and 'less than safe and orderly'. Across all participating education systems, the highest percentage of pupils whose teachers report that their schools are 'very safe and orderly' are in Albania (94%) and Uzbekistan (93%). At the other end of the scale, a relatively high percentage of pupils' teachers said that their schools are 'less than safe and orderly' in South Africa (15%) and Sweden (9%), both well above the International Median of 2% for this category. Table 54 presents the percentage of pupils whose teachers report different levels of safety and orderliness at school for England and comparators. Generally, education systems with a high average performance in PIRLS 2021 often had high frequencies of pupils' teachers reporting that their school was 'very safe and orderly'. In England, 67% of pupils' teachers said that their schools were 'very safe and orderly', and these pupils have an average score 14 points higher than those whose teachers said their schools were 'somewhat safe and orderly'. Australia has a higher-than-average percentage of pupils whose teachers report their schools were 'less than safe and orderly' at 6%, and the average performance of pupils at those schools is 42 points lower than for pupils at 'very safe' schools. In many of the participating education systems, including England, the low frequency of responses in the 'less than safe and orderly' category means that average performance of pupils in those schools cannot be reliably calculated.

Education system	Very safe and orderly	Somewhat safe and orderly	Less than safe and orderly	
England	67%	32%	2%	
Singapore	58%	41%	1%	
Hong Kong SAR	61%	38%	<1%	
Australia	58%	36%	6%	

Table 54: Percentage of pupils in England and comparator systems whose teachers report different levels of safety and orderliness at their school

Because of rounding some results may appear inconsistent.

Source: IEA's PIRLS 2021

The school questionnaire asked headteachers to respond to 10 potential problems associated with school discipline:

- Arriving late at school
- Absenteeism (i.e., unjustified absences)
- Classroom disturbance
- Cheating
- Profanity
- Vandalism
- Theft
- Intimidation or verbal abuse among pupils (including texting, emailing, etc.)
- Physical fights among pupils
- Intimidation or verbal abuse of teachers or staff (including texting, emailing, etc.)

The responses to these problems form the school discipline scale were then classified as 'hardly any problems', 'minor problems' and 'moderate to severe problems'. Table 55 presents the percentage of pupils in schools where headteachers report on the level of problems associated with school discipline in England and comparators. In England and the comparator systems, the percentage of pupils in schools with 'moderate to severe problems' with school discipline is substantially lower than those in schools with 'minor or hardly any problems'. In PIRLS 2016, the percentage of pupils in schools with 'hardly any problems' with discipline in England was higher than reports from PIRLS 2021 (82% vs 76%). Hong Kong had the highest percentage of pupils in schools with 'hardly any problems' with discipline (93%) across all participating education systems, and no pupils' headteachers reported that their school had 'moderate to severe problems' in Hong Kong. The association between school discipline and PIRLS achievement across all participating education systems shows that, on average, pupils in schools with 'hardly any problems' with discipline scored 26 points higher in PIRLS 2021 than those in schools with 'moderate to severe problems'. In England, this range is even higher, pupils in schools with 'hardly any problems' with discipline have an average PIRLS score 50 points higher than those in schools with 'moderate to severe problems'.

Table 55: Percentage of pupils in England and comparator systems whose
headteachers report different levels of problems associated with school discipline

Education system	Hardly any problems	Minor problems	Moderate to severe problems
England	76%	22%	2%
Singapore	81%	19%	<1%
Hong Kong SAR	93%	7%	0%
Australia	69%	29%	2%

Because of rounding some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 56 shows reports from the pupil questionnaire about the frequency at which pupils experience bullying at school in England and comparator systems. Just over half of pupils in England (54%) stated that they 'never or almost never' experience bullying at school. This is lower than the International Median of 62%, and considerably lower than Hong Kong where 81% of pupils say they 'never or almost never' experience bullying. Internationally, Hong Kong has the lowest percentage of pupils who experience bullying on a weekly basis (3%). Reports of weekly experiences of bullying are similar in both England (11%) and Australia (13%), and in both education systems, 35% of pupils report experiencing bullying about once a month.

In England, pupils who experience bullying weekly have lower average achievement in PIRLS (518) than pupils who experience bullying monthly (555), while pupils who almost never experience bullying have the highest average achievement in PIRLS (568). The International Median (480) for pupils who experience bullying weekly is 53 points lower than those who never or almost never experience bullying (533). This negative association between the frequency of bullying experienced and average performance in PIRLS 2021 is evident in all education systems that participated.

Table 56: Percentage of pupils in England and comparator systems and thefrequency of bullying they experience at school

Education system	Never or almost never	About monthly	About weekly
England	54%	35%	11%
Singapore	59%	29%	12%
Hong Kong SAR	81%	16%	3%
Australia	52%	35%	13%

Because of rounding some results may appear inconsistent.

Source: IEA's PIRLS 2021

8.3 Parental involvement

As part of the scale related to school emphasis on academic success, headteachers and teachers responded to items related to parent involvement and support for the school:

- Parental involvement in school activities
- Parental commitment to ensure that pupils are ready to learn
- Parental expectations for pupil achievement
- Parental support for pupil achievement

The section that follows describes the headteachers' perceptions of parent involvement in terms of each of these 4 aspects. In each aspect parental involvement is classified as either 'very high', 'high', 'medium', or 'low/very low'. Overall, teachers' perceptions of parent involvement in terms of these aspects was similar to that of headteachers, further information related to teachers' perspectives on parental involvement is available in Appendix D. **Figure 29** shows the percentage of pupils in England whose headteachers report different levels of parental involvement for each aspect. Across all 4 aspects, parent involvement in England is typically reported to be 'high' or 'medium', with low percentages of pupils' headteachers reporting that parental involvement is 'very high' in any aspect. The tables that follow in this section describe the different levels of parental involvement in more detail for England and its comparators.



Figure 29: Perceptions of the level of parental involvement in different aspects of pupils' educational experiences - headteachers in England (2021)

Aspect of parental involvement	Very High	High	Medium	Low / Very Low
Support for pupil achievement	4%	31%	49%	16%
Expectations of pupil achievement	9%	35%	43%	11%
Commitment to ensuring pupils are ready to learn	5%	30%	50%	16%
Involvement in school activities	4%	36%	40%	21%

Because of rounding some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 57 shows the percentages of pupils whose headteachers feel that parents are involved in school activities and their relative performance in PIRLS 2021. In England, just 4% of pupils' headteachers believe parents are very involved in school activities. This is lower than the corresponding percentages for comparators (10%). The percentage of pupils whose headteachers believe that there is low/very low involvement from parents in school activities is lowest in Singapore (5%), and higher in England (21%) and Australia (25%). Pupils in schools in England where parents are rated as having higher levels of involvement in school activities (i.e., 'very high', 'high' or 'medium') have reasonably consistent average achievement in PIRLS 2021. However, in schools where reports of parental involvement is 'low/very low' in England there is notably lower average PIRLS 2021 achievement.

Table 57: Percentage of pupils in England and comparator systems whoseheadteachers believe that parents are involved in school activities

Education system	Very high involvement	High involvement	Medium involvement	Low involvement
England	4%	36%	40%	21%
Singapore	10%	46%	38%	5%
Hong Kong SAR	10%	40%	34%	15%
Australia	10%	33%	32%	25%

Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 58 shows the percentages of pupils whose headteachers believe that parents are committed to ensuring pupils are ready to learn. The majority of pupils' headteachers in England believe that parental commitment is either 'high' or 'medium', with some reports of 'low/very low' (16%) parental commitment and relatively few reports of 'very high' (5%) parental commitment. A similar trend across categories is seen in Hong Kong, where parental commitment is typically reported as either 'high' or 'medium'. Singapore and Australia both have the same level of 'very high' parental commitment (13%), however, at the other end of the scale they differ. In Singapore just 3% of pupils' headteachers report that parental commitment is 'low/very low', compared to 11% in Australia.

No strong relationship is evident between parental commitment and pupils' average achievement in Hong Kong. In England, Singapore and Australia, however, pupils in schools where headteachers reported that parental commitment is 'high' or 'very high' score higher on average than those in schools reporting lower levels of parental commitment. Where there are a low proportion of responses particularly in the 'very high' and 'low/very low' categories, differences in pupil performance represent a just small sample of pupils and should be interpreted with some caution.

Table 58: Percentage of pupils in England and comparator systems whose headteachers believe that parents are committed to ensuring that pupils are ready to learn

Education system	Very high commitment	High commitment	Medium commitment	Low commitment
England	5%	30%	50%	16%
Singapore	13%	52%	33%	3%
Hong Kong SAR	5%	25%	52%	18%
Australia	13%	42%	35%	11%

Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 59 presents the different levels of parental expectations for pupil achievement reported by pupils' headteachers in England and comparators. In England the percentage of pupils' whose headteachers report parents have 'very high' expectations for pupil achievement is 9%. This is lower than the International Median of 16% and also lower than all comparators. In both Singapore and Australia, almost a quarter (24%) of pupils' headteachers report that parents have 'very high' expectations for pupil achievement. At the other end of the scale, England has the highest percentage of 'low/very low' reports of parental expectations (12%) among comparators. Indeed, the percentage of pupils whose parents have low expectations for pupil achievement in England is among the 4 highest across all participating education systems. The other 3 education systems with high proportions of pupil's headteachers reporting 'low/very low' expectations for pupil achievement are Morocco, South Africa and Jordan. Pupils at schools where headteachers report parents have 'very high' expectations for pupil achievement score higher than those at schools where parents are reported to have lower expectations. This trend is evident in England and internationally.

headteachers believe that parents have high expectations for pupil achieveme					l achievement
	Education system	Very high expectations	High expectations	Medium expectations	Low expectations
	England	9%	35%	43%	12%

53%

41%

41%

24%

15%

24%

Table 59: Percentage of pupils in England and comparator systems whose headteachers believe that parents have high expectations for pupil achievement

Because of rounding, some results may appear inconsistent.

Singapore

Australia

Hong Kong SAR

Source: IEA's PIRLS 2021

1%

3%

3%

22%

41%

32%
A similar pattern in parental commitment to pupil achievement is evident in reports of parental support in ensuring high pupil achievement. **Table 60** shows the different levels of parental support for pupil achievement as reported by pupils' headteachers in England and comparators. In England, a relatively low percentage of pupils' headteachers report 'very high' levels of parental support for pupil achievement (4%). This is the same as the percentage of pupils whose headteachers report 'very high' parental support in Hong Kong, however, the pattern in England shows that there are relatively more reports of 'low/very low' parental support (16%) than in all other comparators. Generally, internationally there is a positive relationship between higher levels of parental support and higher performance in PIRLS 2021. In England, the pupils whose headteachers report that their parents have a 'high' level of support is an average of 43 points higher than those whose parental support is 'low/very low'.

Table 60: Percentage of pupils in England and comparator systems whoseheadteachers believe that parents support the school in ensuring high pupilachievement

Education system	Very high support	High support	Medium support	Low support
England	4%	31%	49%	16%
Singapore	12%	57%	29%	1%
Hong Kong SAR	4%	41%	47%	8%
Australia	13%	38%	42%	7%

Because of rounding, some results may appear inconsistent.

9 Impact of the COVID-19 pandemic

Chapter overview

This chapter explores the impact of the COVID-19 pandemic on PIRLS 2021 in England and comparator education systems. It discusses changes made to data collection procedures, as well as survey responses from school leaders regarding the impact of the pandemic on normal school operations and provision of remote learning resources during periods of disruption. Finally, it discusses how the COVID-19 pandemic may have contributed to differences between observed trends and those that might have been expected between the 2016 and 2021 cycles of PIRLS.

Key findings

- The impact of the COVID-19 pandemic on data collection differed across education systems. Many went ahead with data collection as planned, but 14 education systems delayed data collection by 6 months and tested pupils at the beginning of the fifth year of formal schooling, and 6 systems (including England) delayed data collection by 12 months and tested pupils at the beginning of their fourth year of formal schooling as originally planned.
- There was widespread disruption to normal school operations in the 2020/21 school year across most of the participating education systems, but also substantial variation in the number of weeks affected. For systems that delayed data collection, reported disruptions to school operations referred to a different time period than that in which pupils were tested for this cycle of PIRLS. However, they were also asked to report about school disruptions specifically related to the 2020/21 school year.
- The vast majority of schools offered a variety of resources to support pupils and teachers when normal school operations were disrupted due to the COVID-19 pandemic in the 2020/21 school year. Printed resources and access to digital devices for pupils were less commonly offered than other distance learning resources internationally, but schools in England frequently offered both.
- We cannot exactly predict what results might have been in the absence of the COVID-19 pandemic, but observed trends in reading performance across the 2016 and 2021 cycles suggest that performance decreased in a majority of education systems that did not delay data collection, and either decreased or remained fairly stable in systems that delayed by 12 months and tested the same age group as originally planned. Where older pupils were tested after a 6-month delay, performance increased in a majority of education systems, but it is difficult to disentangle the age effect from any impact of COVID-19.

9.1 Introduction

The COVID-19 pandemic had a substantial impact on school operations in education systems around the world. Although the trends from PIRLS 2016 to 2021 may provide some insight into the potential impact of the pandemic, it is essential to note that PIRLS 2021 was not designed to measure the effects of the COVID-19 pandemic on pupils' reading achievement.

As a result of the pandemic, there were a number of changes to the way in which PIRLS was conducted in England. Data collection was delayed by 12 months, which was one of the choices open to education systems participating in PIRLS. In section **9.2**, we briefly review the impact of the COVID-19 pandemic on data collection for PIRLS 2021 internationally. Items were added to the school and curriculum context questionnaires to collect information related to school closures, COVID-19 impact and remote learning provision. In section **9.3**, we report results based on these items and place these in the broader context of education policy in England in response to the COVID-19 pandemic. In section **9.4**, we discuss differences between observed and expected trends between the 2016 and 2021 cycles of PIRLS, and consider how decisions taken in England and other participating systems about whether and for how long to delay data collection seem to have affected the impact of COVID-19 on PIRLS 2021 performance.

9.2 Impact of COVID-19 on data collection

Disruptions to normal primary school operations impacted data collection considerably. As noted in section **1.2.3**, participating education systems fell into 3 broad groups according to whether and how they made adjustments to the original data collection plans for PIRLS 2021. Two Southern Hemisphere and 35 Northern-Hemisphere education systems chose not to make any adaptations and tested pupils at the end of their fourth year of schooling according to the original timeline. Fourteen Northern Hemisphere systems delayed testing by 6 months, and tested pupils at the beginning of their fifth year of formal schooling. Three Southern Hemisphere and 3 Northern Hemisphere systems (including England) tested pupils in the age group originally intended, i.e., pupils who were at the end of their fourth year of formal schooling (year 5 in England), but tested one year later than originally planned.

In England, another adaptation to data collection made due to the COVID-19 pandemic was that schools were given the option to self-administer PIRLS 2021 if they preferred to do so. Training was provided for internal administrators (i.e., members of school staff) to facilitate this. Remote training was provided for both scorers and test administrators (von Davier et al., 2023), and this training included guidance on additional COVID-19 requirements for test administration (Kayton & McGrane, 2022).

9.3 Impact of COVID-19 on school operations

Headteacher questionnaires in England and in most education systems participating in PIRLS 2021 included a limited number of questions concerning the effect of the pandemic on school operations and the remote learning provision when face-to-face instruction did not take place due to any periods of school closure. For the sake of comparability across education systems, these questions specifically referred to the 2020/21 school year regardless of when individual systems collected their data, but the phrasing of these questions referred to "the current academic year (2020/21)". Inferences drawn from the responses to these items, and comparisons regarding these results across systems, must therefore be cautious at best – headteachers in England, for example, would have had to respond based on their best recollection of the previous year (2020/21) while pupil performance was measured in 2021/22, and some may have mistakenly responded based on the "current academic year" (2021/22). It is also important to note that while Singapore is one of the comparators for which results are reported in other chapters of this report, Singapore is not included in results reported below because COVID-19 pandemic-related information was not collected there.

9.3.1 School closures

There were 2 nationwide school closures in England in 2020/21, one lasting from March to September 2020 and then another from late December 2020 to early March 2021. Some pupils (including children of critical workers and children classified as vulnerable) continued to access at least some in-person learning, but most pupils were affected by these closures (see Kayton & McGrane, 2022 for a more detailed account). In PIRLS 2021, headteachers were asked to indicate the number of weeks in which primary school operations were affected by the COVID-19 pandemic in 2020/21. **Table 61** shows the percent of headteacher responses by ranges of number of weeks affected (from 'Not affected' to 'More than 8 weeks') in England, Hong Kong, and Australia. Note that Singapore, the remaining comparator, did not administer questionnaire items relating to the COVID-19 pandemic.

Education system	Not affected	Less than 2 weeks	2-4 weeks	5-8 weeks	More than 8 weeks
England	22%	12%	10%	15%	41%
Hong Kong SAR	5%	7%	16%	14%	59%
Australia	11%	20%	14%	9%	46%
International Median	11%	10%	12%	12%	39%

Table 61: Number of weeks in which primary school operations were affected by the COVID-19 pandemic in 2020-21 (percent of schools⁷ by education system)

Some results may appear inconsistent due to rounding.

Source: IEA's PIRLS 2021

There was considerable variation in the number of weeks affected across schools in England as well as in comparator education systems. In nearly half of the participating schools in England and Australia, and more than half of the participating schools in Hong Kong SAR, primary school operations were reported to have been affected for more than 8 weeks of the 2020/21 school year.

The apparent variation across education systems in terms of the proportions of schools in which operations were affected for different amounts of time may be related to accuracy of headteacher reporting to some extent, and owing to the phrasing of the question, some may have reported on the current year (e.g., in Hong Kong, where they were reporting weeks affected in the year of data collection, versus in England and Australia where they were reporting weeks affected in the previous year). Still, this variation across systems can be seen as an indication of how the COVID-19 pandemic affected primary school operations differently across contexts, which introduces complexities in interpreting comparative results from PIRLS 2021 internationally.

9.3.2 Remote learning provision

Headteachers were also asked whether their schools provided remote instruction or distance learning resources for primary grades when normal school operations were affected by the COVID-19 pandemic, followed by a question asking whether or not specific types of support for remote learning were offered, including:

- Delivery of printed learning materials to pupils
- Internet-based resources for pupils
- Access to digital devices for pupils

⁷ Please note that results in this report may differ from the results reported by the IEA where school-level responses are analysed. This is because in this report school weights were used to analyse school-level responses, while IEA results analysed school-level responses using student weights. Conclusions are not substantially affected either way, but exact numbers may differ slightly.

- Recommendations for teachers about how to provide online instruction
- Technical support for teachers
- Access to digital devices for teachers

Table 62 shows the percentages of schools that offered remote instruction or distance learning resources in general, as well as the percentages offering each type of support for remote learning in 2020/21 when normal school operations were affected by the pandemic. The vast majority of all participating schools offered all of these resources when normal operations were disrupted by the COVID-19 pandemic, according to headteacher responses, and nearly all offered at least some sort of remote instruction or distance learning resources.

In England, this is hardly surprising, as Government guidance required schools to provide access to remote learning in any instance where school-age children could not attend school as a result of public health advice, and there were statutory expectations for remote learning provision in these circumstances (see Kayton & McGrane, 2022 for further details). International Median results suggest that it was slightly less common internationally for schools to offer delivery of printed learning materials or access to digital devices for pupils, but in England and comparators printed learning materials were somewhat more commonly provided, and in England provision of access to digital devices for pupils was widespread.

Table 62: Resources offered when normal school operations were disrupted by theCOVID-19 pandemic (percent of schools by education system)

Offered by schools during periods of disruption?	England	Hong Kong SAR	Australia	International Median
Remote instruction/ distance	99%	98%	94%	96%
learning resources	5570	3070	5470	3070
Delivery of printed learning	89%	87%	90%	84%
material to pupils	0370	0770	3070	0470
Internet-based resources for	97%	100%	100%	96%
pupils	9170	100 /0	100 /0	9070
Access to digital devices for	95%	88%	79%	77%
pupils	9070	00 /0	1970	1170
Recommendations for				
teachers about how to provide	97%	100%	95%	97%
online instruction				
Technical support for teachers	94%	100%	94%	91%
Access to digital devices for	97%	97%	98%	87%
teachers	5170	5170	3070	0770

Some results may appear inconsistent due to rounding.

9.3.3 Catch-up programmes

Although PIRLS 2021 did not collect specific data on catch-up programmes, a brief account of what was offered to schools in England to mitigate the disruption to children's learning due to the COVID-19 pandemic helps to contextualise insights and findings from PIRLS 2021. Catch-up provision is multifaceted and not all specifically focused on a particular subject area (or focused on Science, Technology and Mathematics rather than on reading). Here, we focus on what has been offered to pupils with direct or close relevance to reading.

The National Tutoring Programme was put in place by the Department for Education as one of the core elements of catch-up provision for pupils. This programme provided and continues to provide funding for primary and secondary schools to spend on targeted academic support, which could take the form of academic mentors with full-time dedicated roles providing intensive and targeted support, external tuition partners brought in to offer additional support, or school-led tutoring provided by school personnel (Department for Education, 2022). Total funding for this programme includes £1bn over 4 years to 2023/24, with £350m provided to schools in 2022/23. The National Tutoring Programme was closely linked to the catch-up premium offered to schools for the 2020/21 school year (£650 million in total across England in 2020/21; Department for Education, 2021).

Free online lessons have been another relevant type of provision on offer to try to catch pupils up after COVID-19 interruptions to learning. For example, Oak National Academy (an independent public body; <u>https://www.thenational.academy/</u>) offers a range of content across subject areas to support pupil learning and enrichment.

9.4 Expected performance trends 2016-2021 vs observed performance trends 2016-2021

As the previous sections indicate, the COVID-19 pandemic impacted data collection for PIRLS 2021 as well as normal school operations internationally. As a result, it also introduced considerable complexity to the interpretation of results and trends over time across participating education systems. It is not possible to disentangle all of this complexity using statistical techniques, so we cannot accurately predict what the results would have been in England or elsewhere in the absence of the COVID-19 pandemic. That said, we can draw some cautious inferences based on observed trends across systems according to the different choices made concerning whether and how much to delay data collection, as well as whether to test the originally intended age group (the end of the fourth year of formal schooling, or year 5 in England) or the original cohort at a later point in time (i.e., the beginning of the fifth year of formal schooling).

Table 63 displays the differences in overall performance since the previous cycle for education systems that participated in both PIRLS 2016 and PIRLS 2021, grouped by their data collection choices (as originally planned, delayed by 6 months and testing at the beginning of the fifth year of formal schooling, or delayed by 12 months and testing at the end of the fourth year of formal schooling). The lack of a consistent pattern across these groups is likely a consequence of the fact that the impact of the COVID-19 pandemic differed across contexts in ways that cannot be described by the available data for this study. Nonetheless, there are some broad trends: A larger proportion of systems that collected data as originally planned showed significant decreases in overall reading performance, and a larger proportion of systems that delayed by 6 months and tested an older age group showed significant increases in overall reading performance since 2016. These results suggest that COVID-19 may have had a greater impact on reading performance outcomes in education systems that continued with data collection as originally planned. The complication of testing an older age group in systems that delayed data collection by 6 months makes it difficult to disentangle the impact of the pandemic from any effect of pupil age.

Of the 9 education systems that saw increases in their scores since previous cycles in PIRLS 2021, 7 were located in North Africa (Egypt, Morocco) or the Gulf states (Bahrain, Oman, Qatar, Saudi Arabia and the United Arab Emirates). Four of these education systems (Oman, Qatar, Morocco and the United Arab Emirates) already appeared to have upward trajectories in their PIRLS scores from 2011 to 2016. While these may be instances in which broad trends within systems continued despite the COVID-19 pandemic, this does not necessarily mean that these education systems were unaffected; we cannot know whether their scores would have been similar or perhaps even higher under different circumstances.

Performance	Education system
Performance Tested at the originally planned time (tested pupils at the end of the fourth grade)	Egypt (+48) Oman (+11) ** Singapore (+11) New Zealand ** Hong Kong SAR Czech Republic France Slovak Republic Belgium (French) Spain (-7) Denmark (-8) Portugal (-8) Macao SAR (-10) Italy (-11) Austria (-11) * Sweden (-12) * Bulgaria (-12) Germany (-13) Belgium (Flemish) (-14) ** Russia (-14) * Taiwan (-15)
	* Bulgaria (-12) Germany (-13) Belgium (Flemish) (-14) ** Russia (-14)
	Netherlands (-18) * Norway (-20) Slovenia (-23) Azerbaijan (-32)

Table 63: Trends across 2016 and 2021 PIRLS cycles according to data collectionadaptations due to the COVID-19 pandemic

Performance	Education system
Delayed testing by 6 months (tested pupils at the beginning of the fifth grade)	Qatar (+42) United Arab Emirates (+33) Morocco (+15) Saudi Arabia (+18) Bahrain (+12) Ireland (+11) Georgia Lithuania Northern Ireland United States Hungary (-15) Kazakhstan (-32) Latvia (-30)
Delayed testing by 12 months (tested pupils at the end of the fourth grade)	Australia England Iran (-15) Israel (-20) South Africa (-31)

Significant changes in score from PIRLS 2016 to PIRLS 2021 shown in parentheses where appropriate. A single asterisk (*) indicates that the education system did not score significantly differently to England in PIRLS 2016. A double asterisk (**) indicates that the education system scored significantly higher than England in PIRLS 2016.

Source: IEA's PIRLS 2021

While PIRLS 2021 was not designed to measure the impact of the COVID-19 pandemic on reading performance and progress, other studies have endeavoured to do so. One study found that pupils were approximately 2 months behind expected progress in reading when they returned to classrooms in Autumn 2020, and under a month behind on average by the end of term in Summer 2021 (Renaissance Learning & Education Policy Institute, 2021). The same study also found that learning losses in reading varied according to individual and local area disadvantage as well as by region. Another study by Blainey and Hannay (2021) found that at least in years 2 to 6 in England, reading results seemed to be close to what would have been expected before the COVID-19 pandemic. Evidence from Ofsted reports has indicated that primary school leaders identify reading as an area of pupil learning that was substantially affected by COVID-19 disruption (Ofsted, 2020), but this did not involve direct measurement of reading performance or progress so must be cautiously interpreted.

Taken together, these different sources paint a mixed picture of the effect of the COVID-19 pandemic on reading in primary school. Tentatively, we might speculate that on average reading may not have been dramatically affected compared to some other school subjects and skills, but that disruptions to teaching and learning may have disproportionately affected some schools and pupils according to the particularities of individual, local, school and regional contexts.

10 PIRLS and England's educational policy context

Chapter overview

This chapter provides a discussion of the educational policy context in England. It describes the relationship between PIRLS International Achievement Benchmarks and national curriculum expectations in England. The chapter also presents a discussion about the PIRLS results in the context of national assessments of reading, as well as reading initiatives in England. Finally, it discusses the relationship between educational policies and PIRLS in other participating education systems.

10.1 Current and past policy developments in England

The results from PIRLS can provide valuable system-level insights regarding reading policy in England. However, educational policy is complex and the relationship between the results of PIRLS and policy structures or interventions should not be considered causal. The evolving nature of educational policy also means that some interventions or initiatives discussed here have not been in place for sufficient periods of time to effectively evaluate their impact. Nevertheless, PIRLS provides valuable data about the reading achievement of year 5 pupils in England that can help to understand and inform policy interventions over time.

10.1.1 National curriculum

PIRLS provides detailed information regarding specific reading competencies associated with different levels of achievement reached by pupils. This information is valuable for understanding the relationship between reading achievement in PIRLS and reading curriculum expectations specific to the context of participating education systems.

A new national curriculum was introduced in England in 2014, and this is the first PIRLS cycle where the participating pupils have been exclusively taught under the new curriculum. **Table 64** presents the core knowledge and skills in reading comprehension according to the national language and reading curriculum for key stage 1 and key stage 2 in England. Also included in this table is an indication of which of the PIRLS International Benchmarks are associated with each of the curriculum expectations. (For a more detailed overview of the International Benchmark competencies see Chapter 2). Pupils participating in PIRLS 2021 were in year 5 and are expected to have achieved the knowledge and skills for reading comprehension in key stage 1 and to be making good progress on the knowledge and skills for key stage 2. It is important to note that they are not yet expected to have achieved all the skills listed in the table.

Table 64: Key stage 1 and 2 reading comprehension skills and knowledge and theirassociation with PIRLS International Benchmark levels

Key stage	Knowledge and skills for reading comprehension	Associated with PIRLS benchmarks
Key stage 1 (years 1-2)	Understand both the books that they can already read accurately and fluently and those that they listen to	Low
	Develop pleasure in reading, motivation to read, vocabulary, and understanding	Low & Intermediate
	Participate in discussions about books, poems, and other works that are read to them and those that they can read for themselves	Intermediate
Key stage 2 (years 3-6)	Develop and maintain positive attitudes to reading and understanding of what they read	Intermediate
	Retrieve, record, and present information from nonfiction	Intermediate & High
	Be able to discuss and evaluate how authors use language, including figurative language, considering the impact on the reader	High
	Distinguish between statements of fact and opinion	High
	Participate in discussions about books that are read to them and those they can read for themselves, building on their own and others' ideas and challenging views courteously	High & Advanced
	Explain and discuss their understanding of what they have read, including through formal presentations and debates	High & Advanced

Key stage 1 expectations can be associated with competencies in Low and Intermediate PIRLS International Benchmarks. There is an overlap of the benchmarks across key stages and a few of the competencies from the Intermediate PIRLS Benchmark are more closely related to expectations for key stage 2. Results from the 2021 cycle of PIRLS show that 97% of pupils in England reached the Low Benchmark, while 86% reached the Intermediate Benchmark. This indicates that a large majority of pupils in year 5 have achieved results that are aligned with the curriculum expectations from key stage 1 for reading comprehension.

The curriculum requirements for key stage 2 are more closely aligned to the High International Benchmark. The results from PIRLS 2021 show that 57% of participating pupils were able to reach the High Benchmark level. Some of the skills in key stage 2 are also associated with the Advanced International Benchmark, and results show that 18% of pupils in England were able to demonstrate reading skills at the Advanced level. Year 5 pupils that participated in PIRLS 2021 have not yet finished key stage 2, and as such are not yet expected to have achieved all the skills associated with key stage 2 for reading comprehension. The results show that good progress is being made, with almost all pupils demonstrating key stage 1 reading skills, and more than half of the pupils demonstrating skills associated with key stage 2 requirements. For a more detailed discussion of the Benchmark achievement for England, see Chapter 2.

10.1.2 Reading initiatives

In recent years, there have been a number of policy interventions in England that have aimed to improve reading achievement and reading instruction in primary school. In 2010, a major drive to improve the standards of reading was launched. This drive promotes Systematic Synthetic Phonics (SSP) instruction through initiatives such as a matched funding scheme, phonics roadshows and phonics partnerships between schools and, more recently, the English Hubs Programme offering school to school support and validation of phonics programmes. The value of SSP instruction for the early stages of reading development has been discussed widely, and a thorough grounding in phonics is understood to be an integral tool for developing strong reading (Castles, Rastle & Nation, 2018; Snowling et al., 2019). The emphasis on reading instruction in England is also evident in the survey responses from PIRLS. The teachers of pupils who participated in PIRLS 2021 reported that they spend, on average, 3 and a half hours per week on instruction or activities specifically related to reading.

The value of a good grounding in phonics for reading achievement in England has been shown by researchers in England investigating the link between early grade phonics performance and later achievement in PIRLS (Double et al., 2019; Stainthorp, 2020). The prioritisation of a good grounding in phonics in the early years was clear from the questionnaire responses in England. Almost all participating pupils in England (98%) attend schools where headteachers stated that knowing letter-sound relationships first receives major emphasis in year 1 or earlier. PIRLS, however, is aimed at pupils who have already transitioned from learning to read into reading to learn and the PIRLS Reading Framework is therefore focused on later phases of reading where readers read "to learn, to participate in communities of readers in school and everyday life, and for enjoyment" (Mullis & Martin, 2021, p. 6).

Findings from PIRLS have shown a positive relationship between pupils' enjoyment of reading and their achievement in PIRLS (see Chapter 7). In England and internationally, pupils who report that they like reading have higher average performance in PIRLS than those who report that they do not like reading.

Several interventions supporting reading for pleasure are in place in England. In 2015, an evidence-based plan for ensuring progress in reading development that aimed to

continue to focus on SSP instruction while promoting the development of mature readers was outlined (Department for Education, 2015). This plan encourages reading for pleasure through the promotion of book clubs and library membership for pupils in key stage 2. Some interventions in England focused on the promotion of reading for pleasure in primary schools – such as the English Hubs Programme (2018) and the new Reading Framework (2021) – have not been in place long enough to discuss their impact yet.

10.1.3 National assessments of reading

Year 5 pupils who participated in PIRLS 2021 have previously participated in national assessments of reading in year 1 (Phonics screening check) and at the end of key stage 1 (key stage 1 Reading). Since its implementation in 2012, the predictive value of the phonics screening check and its link to later reading achievement has been widely discussed (Carter, 2020; Double et al., 2019; Goldstein et al., 2018). The positive relationship seen between performance on the phonics screening check and PIRLS results (discussed in more detail in Chapter 5), provides support for the effectiveness of early monitoring systems.

10.1.4 Teacher professional development

A recent reform of teacher education in England has made changes to both the initial teacher education systems and teacher professional development programmes. The new structures focus on mentorship and early career support opportunities for teachers entering the field. Although not mandated, ongoing professional development for teachers in England is considered a professional duty, and 5 days per year are set aside for professional development. Almost all (>97%) of pupils' teachers in PIRLS 2021 reported that they found professional development activities such as workshops, seminars and 'access to a mentor' helpful for their ongoing professional development.

A new National Professional Qualification for Leading Literacy (NPQLL) launched in October 2022. The NPQLL will train existing teachers and leaders to become literacy experts who will drive up standards of literacy teaching in their schools and improve literacy outcomes for every child. The NPQLL programme is still in the early stages of implementation; future PIRLS cycles will be better placed to assess the potential impact of the training and development programme on improving literacy outcomes for pupils.

10.1.5 Attainment gap

Narrowing the gap in performance between high and low achieving pupils has been an ongoing area of focus in England's education ecosystem. Phonics partnership programmes (2015-2019) linked schools with proven expertise with less successful schools to help improve the quality of phonics instruction. The English Hubs programme, which has been active since 2018, is a school-to-school improvement programme that

focusses on systematic synthetic phonics, early language, and reading for pleasure. These programmes focus on improving reading skills for pupils at the lower end of the performance scale without compromising the performance of pupils at the higher end of the performance scale. Performance at the higher end (90th percentile) has remained stable across all cycles of PIRLS in England, while performance at the lower end (10th percentile) showed consistent improvement from 2001-2016 and has remained stable between 2016 and 2021 (see Chapter 4 for more detail). Overall, the PIRLS 2021 results revealed that the attainment gap in England has consistently narrowed across cycles of PIRLS. This means that outcomes for lower-performing pupils are improving while the higher-performing pupils are not compromised.

A programme of targeted funding support for schools with higher proportions of disadvantaged pupils, known as the Pupil Premium, was introduced in 2011. The purpose of the Pupil Premium is to use evidence-based methods to improve educational outcomes for disadvantaged pupils. Pupils who are eligible for FSM, or have been in the past 6 years, as well as currently or previously looked-after children are eligible for Pupil Premium. Pupils who are eligible for FSM had lower average PIRLS performance in PIRLS 2021 than their peers (see Chapter 6 for a full discussion). This is consistent with findings from the Educational Endowment Foundation (EEF) that help to guide and support the implementation of the Pupil Premium. More recently, an Accelerator Fund for English has been introduced with the aim to invest in improving reading skills as well as assisting with recovery from the pandemic (Department for Education, 2022). The full effect of both the Pupil Premium and Accelerator Fund are only expected to be evident in future cycles of PIRLS.

10.1.6 Educational policies in relation to PIRLS in other participating education systems

There is a wide range of educational policies specific to reading employed by education systems that participate in PIRLS, each with their own objectives and areas of focus. This diversity makes cross-system comparisons particularly complicated in the area of policy. Nevertheless, there are some policies and interventions common to a number of education systems participating in PIRLS that can be explored to better understand the relationship between PIRLS and educational policy more broadly.

As discussed previously in this report, a primary focus area of the reading curriculum in England has been a focus on phonics instruction in the primary school years. Several of the education systems that participate in PIRLS also report in the PIRLS 2021 Encyclopedia that they focus on phonics the early stages of reading instruction. Although there is a positive relationship seen in England between phonics performance and later PIRLS performance (see Chapter 5), this relationship is more difficult to determine across at a cross-system level. As was seen PIRLS 2016, the education systems that apply phonics instruction in their approach include those whose average performance in PIRLS 2021 ranks near the top end of scores (e.g., Singapore) as well as some at the lower end of the rankings (e.g., South Africa).

Similar to England's phonics screening check, comparators Australia and Singapore both have screening checks that take place in the first year of primary school. In all 3 education systems, these checks aim to identify pupils who may require additional learning support in reading at an early age so that they can receive the necessary support. In Australia, their phonics screening check was implemented in an effort to improve reading outcomes (Knowles & Hillman, 2022). In Singapore, pupils that are identified as requiring additional support through their screening check in year 1, then participate in a learning support programme to build basic English literacy skills for up to 2 years (Ministry of Education, 2022). In both Australia and Singapore, the performance of pupils at the lower end of the achievement scale in PIRLS has shown some improvement over time but has mostly remained consistent.

The language and reading curricula of participating education systems are an important feature of PIRLS, which aims to be a forward-thinking and curriculum-aligned assessment. In many of the education systems that participated in PIRLS 2021, language and reading curriculums are reasonably new (introduced within the last 10 years) and/or are currently being revised (Reynolds et al., 2022). In Hong Kong, an updated curriculum for primary school was implemented in 2014, the same year as England introduced its new national curriculum. The updated curriculum in Hong Kong has a special focus in the area of Learning to Learn, and more specifically, Reading to Learn. There is considerable overlap between how PIRLS defines reading and how the focus of reading to learn in Hong Kong, where a primary focus is ensuring that reading as a means of improving lifelong learning and holistic personal development.

Reading interventions to improve reading are common among education systems participating in PIRLS 2021 (Reynolds et al., 2022). Although these vary considerably across education systems, the promotion of reading for enjoyment and increasing time and opportunity to read are frequently noted. Hong Kong, for example, promotes access to libraries through a Library-on-Wheels project that aims to being books to different communities. Programmes to encourage enjoyment from reading are also common across education systems participating in PIRLS. In Singapore, a Read@School programme promotes storytelling, games and book talks aligned with pupils' interests and in Ireland, a Drop Everything and Read (DEAR) programme has been implemented.

10.2 Concluding remarks on policy changes in England and PIRLS

The relationship between national educational policy and international large-scale assessments such as PIRLS is one of the primary areas of interest for participating education systems. The complexity and diversity of approaches both within and across education systems makes drawing substantiative conclusions about the relationship between PIRLS and policy difficult. Nevertheless, there are some indications that, overall, the lessons that can be learned from PIRLS 2021 are positive for policy changes in England. The positive relationship between the results from the Phonics screening check and future PIRLS performance suggest that the screening check can be helpful in predicting later reading ability in primary school. Furthermore, the narrowing of the attainment gap in England suggests that efforts to improve the achievement of lowerperforming pupils and schools are having a positive impact. The progress of more recent changes in policy, as well as the influence of adaptations to teaching and learning policy that resulted from the COVID-19 pandemic, will need more time and research to understand fully. However, despite recent challenges, studies such as PIRLS are able to provide some evidence to support ongoing improvements to educational policy in England.

References

- Al Maskari, Z., Al Mawali, F., & Al Zadjali, S. (2022). PIRLS 2021 Encyclopedia: Oman. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Åse Kari H. Wagner, Å. K. H., & Støle, H. (2022). PIRLS 2021 Encyclopedia: Norway. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Badri, M. (2019). School Emphasis on Academic Success and TIMSS Science/Math Achievements. *International Journal of Research in Education and Science*, 5(1), 176-189.
- Bezirhan, U., Foy, P., & von Davier, M. (2023). Chapter 10: PIRLS 2021 Achievement Scaling Methodology: Item Response Theory, Population Models, and Linking Across Modes in von Davier, M., Mullis, I. V. S., Fishbein, B., & Foy (Eds.). *Methods and Procedures: PIRLS 2021 Technical Report* (pp. 10.1-10.19). Boston College, TIMSS & PIRLS International Study Center. Retrieved from <u>https://doi.org/10.6017/lse.tpisc.tr2103.kb3131</u>
- Blackman, S. N. (2022). The impact of Covid-19 on education equity: A view from Barbados and Jamaica. *Prospects*, 51(4), 611-625.
- Blainey, K., & Hannay, T. (2021). The effects of educational disruption on primary school attainment in summer 2021. RS Assessment & SchoolDash. Retrieved from <u>https://www.risingstars-uk.com/media/Rising-Stars/Assessment/Whitepa-</u> <u>pers/RSA Effects of disruption Summer Aug 2021.pdf</u>
- Borgonovi, F. (2016). Video gaming and gender differences in digital and printed reading performance among 15-year-olds students in 26 countries. *Journal of adolescence*, 48, 45-61.
- Borgonovi, F. (2022). Is the literacy achievement of teenage boys poorer than that of teenage girls, or do estimates of gender gaps depend on the test? A comparison of PISA and PIAAC. *Journal of Educational Psychology*, 114(2), 239–256. <u>https://doi.org/10.1037/edu0000659</u>
- Borgonovi, F., Ferrara, A., & Maghnouj, S. (2018), "The gender gap in educational outcomes in Norway", OECD Education Working Papers, No. 183, OECD Publishing, Paris, <u>https://doi.org/10.1787/f8ef1489-en</u>.
- Borgonovi, F., Choi, A., & Paccagnella, M. (2021). The evolution of gender gaps in numeracy and literacy between childhood and young adulthood. *Economics of Education Review*, 82, 102119.
- Boyd, S. (2022). PIRLS 2021 Encyclopedia: Northern Ireland. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Cai, Z., Fan, X., & Du, J. (2017). Gender and attitudes toward technology use: A metaanalysis. *Computers & Education*, 105, 1-13.
- Çaliskan, E. F., & Ulas, A. H. (2022). The Effect of Parent-Involved Reading Activities on Primary School Students' Reading Comprehension Skills, Reading Motivation, and

Attitudes towards Reading. *International Electronic Journal of Elementary Education*, 14(4), 509-524.

- Carter, J. (2020). The assessment has become the curriculum: Teachers' views on the Phonics Screening Check in England. *British Educational Research Journal*, 46(3), 593–609. <u>https://doi.org/10.1002/BERJ.3598</u>
- Castles, A., Rastle, K., & Nation, K. (2018). Ending the Reading Wars: Reading Acquisition From Novice to Expert. *Psychological Science in the Public Interest*, 19(1), 5–51. <u>https://doi.org/10.1177/1529100618772271</u>
- CCEA. (2007). Primary Curriculum in Northern Ireland. Belfast: CCEA. https://ccea.org.uk/document/924
- Dahan Golan, D., Barzillai, M., & Katzir, T. (2018). The effect of presentation mode on children's reading preferences, performance, and self-evaluations. *Computers & Education*, 126, 346–358.
- Damani, K., Daltry, R., Jordan, K., Hills, L., & Evans, L. (2022). EdTech for Ugandan girls: Affordances of different technologies for girls' secondary education during the Covid-19 pandemic. *Development Policy Review*, 40, e12619.
- De Paz Nieves, C., Gaddis, I., & Muller, M. (2021). Gender and COVID-19. WorldBank.
- Delaney, E., McHugh, G., McAteer, S., & O'Neill, B. (2022). PIRLS 2021 Encyclopedia: Ireland. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Denies, K., Vanbuel, M., Aesaert, K., Van Keer, H., Adriaens, V., Erauw, I., Denys, A., & Vermeire, E. (2022). PIRLS 2021 Encyclopedia: Belgium (Flemish). Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Department for Education. (2015). Reading: The next steps. Supporting higher standards in schools. Retrieved from <u>https://www.gov.uk/government/uploads/system/up-</u> loads/attachment_data/file/409409/Reading_the_next_steps.pdf
- Department for Education. (2021). Catch up premium: Guidance. Retrieved from: <u>https://www.gov.uk/government/publications/catch-up-premium-coronavirus-covid-</u> <u>19/catch-up-premium</u>
- Department for Education. (2021). The Reading Framework: Teaching the foundations of literacy. Retrieved from <u>https://www.gov.uk/government/publications/the-reading-framework-teaching-the-foundations-of-literacy</u>
- Department for Education. (2022). Bumper £24 million to boost children's literacy. Press Release: 9 October 2022
- Department for Education. (2022). English Hubs. Retrieved from <u>http://www.eng-lishhubs.org/</u>
- Department for Education. (2022). National Tutoring Programme: guidance for schools 2022 to 2023. Retrieved from: <u>https://www.gov.uk/government/publications/na-tional-tutoring-programme-guidance-for-schools-2022-to-2023</u>
- Double, K. S., McGrane, J. A., Stiff, J. C., & Hopfenbeck, T. N. (2019). The importance of early phonics improvements for predicting later reading comprehension. *British Educational Research Journal*, 45(6), 1220–1234. <u>https://doi.org/10.1002/berj.3559</u>

- Engdal Jensen, R. (2020). Implications of Changing the Delivery Mode on Reading Tests in Norway—A Gender Perspective. *Equity, equality and diversity in the Nordic model of education*, 337-361.
- Fougt, S. S., Molbæk, L., Neubert, K., Gabrielsson, R. H., & Mejding, J. (2022). PIRLS 2021 Encyclopedia: Denmark. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Frønes, T. S., Pettersen, A., Radišić, J., & Buchholtz, N. (2020). Equity, equality and diversity in the Nordic model of education (p. 412). *Springer Nature*.
- Goldstein, H., Moss, G., Sammons, P., Sinnott, G., & Stobart, G. (2018). A baseline without basis: The validity and utility of the proposed reception baseline assessment in England. Summary, London: British Educational Research Association. Retrieved July, 27, 2018.
- Goodrich, S., & Ercikan, K. (2019, December). Measurement Comparability of Reading in the English and French Canadian Populations: Special Case of the 2011 Progress in International Reading Literacy Study. In *Frontiers in Education* (Vol. 4, p. 120). Frontiers Media SA.
- Grisay, A., Gonzalez, E., & Monseur, C. (2009). Equivalence of item difficulties across national versions of the PIRLS and PISA reading assessements. *IERI*, 2.
- Grissom, J. A., Kalogrides, D., & Loeb, S. (2017). Strategic staffing? How performance pressures affect the distribution of teachers within schools and resulting student achievement. *American Educational Research Journal*, 54(6), 1079-1116.
- Ibrahim, A. (2020). What hurts or helps teacher collaboration? Evidence from UAE schools. *Prospects*, 1-18.
- January, S. A. A., & Klingbeil, D. A. (2020). Universal screening in grades K-2: A systematic review and meta-analysis of early reading curriculum-based measures. Journal of School Psychology, 82, 103-122.
- Jenkinson, R. C., & Alzaabi, S. (2022). PIRLS 2021 Encyclopedia: Oman. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Kaplan, D., & Jude, N. (2022). Trend analysis with international large-scale assessments: Past practice, current issues, and future directions. In *International Handbook of Comparative Large-Scale Studies in Education: Perspectives, Methods and Findings* (pp. 831-844). Cham: Springer International Publishing.
- Kaya, M., & Erdem, C. (2021). Students' well-being and academic achievement: A metaanalysis study. Child Indicators Research, 14(5), 1743-1767.
- Kayton, H. L., & McGrane, J. A. (2022). PIRLS 2021 Encyclopedia: England. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Knowles, S. & Hillman, K. (2022). PIRLS 2021 Encyclopedia: Australia. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Kong, Y., Seo, Y., & Zhai, L. (2018). Comparison of reading performance on screen and on paper: A meta-analysis. *Computers & Education*, 123, 138–149.

- Kopas, G. (2022). PIRLS 2021 Encyclopedia: Slovak Republic. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Matsumoto, A. (2019). Literature review on education reform in the UAE. *International Journal of Educational Reform*, 28(1), 4-23.
- Mavrodieva, M., & Damyanova, A. (2022). PIRLS 2021 Encyclopedia: Bulgaria. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- McGrane, J. A., Stiff, J., Baird, J.-A., Lenkeit, J., & Hopfenbeck, T. (2017). Progress in International Reading Literacy Study (PIRLS): National report for England. Department for Education. Retrieved from <u>https://www.gov.uk/government/publica-</u> <u>tions/pirls-2016-reading-literacy-performance-in-england</u>
- Mihajlovska, B. (2022). PIRLS 2021 Encyclopedia: North Macedonia. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Ministry of Education, Singapore. (2020). Blended learning to enhance schooling experience and further develop students into self-directed learners. Retrieved from <u>https://www.moe.gov.sg/news/press-releases/20201229-blended-learning-to-enhance-schooling-experience-and-further-develop-students-into-self-directed-learners</u>
- Ministry of Education, Singapore. (2021). Special educational needs support at mainstream primary schools. Retrieved from <u>https://www.moe.gov.sg/special-educational-needs/school-support/primary-schools</u>
- Ministry of Education, Singapore. (2022a). PIRLS 2021 Encyclopedia: Singapore. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Ministry of Education, Singapore. (2022b). Educational technology journey. Retrieved from: <u>https://www.moe.gov.sg/education-in-sg/educational-technology-journey</u>
- Ministry of Education, UAE. (2016). United Arab Emirates school inspection framework 2015–2016. Dubai: MoE.
- Ministry of Education. (2022). PIRLS 2021 Encyclopedia: Singapore. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Moussa, N. M., & Ali, W. F. (2022). Exploring the relationship between students' academic success and happiness levels in the higher education settings during the lockdown period of COVID-19. *Psychological Reports*, 125(2), 986-1010.
- Mullis, I. V. S., von Davier, M., Foy, P., Fishbein, B., Reynolds, K. A., & Wry, E. (2023). PIRLS 2021 International Results in Reading. Boston College, TIMSS & PIRLS International Study Center. <u>https://pirls2021.org/results</u>
- Mullis, I. V. S., & Martin, M. O. (Eds.). (2021). PIRLS 2021 Reading Assessment Framework. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://timssandpirls.bc.edu/pirls2021/frameworks/</u>

- National Council for Curriculum and Assessment. (2019). Primary Language Curriculum. Retrieved from <u>https://curriculumonline.ie/getmedia/2a6e5f79-6f29-4d68-b850-</u> <u>379510805656/PLC-Document English.pdf</u>
- Ofsted. (2020). COVID-19 series: briefing on schools, November 2020. Retrieved from https://www.gov.uk/government/collections/ofsted-covid-19-series
- Organisation for Economic Cooperation and Development. (2018). PISA 2018 results (volume II). OECD Publishing. <u>https://doi.org/10.1787/b5fd1b8f-en</u>
- Reilly, D., Neumann, D. L., & Andrews, G. (2019). Gender differences in reading and writing achievement: Evidence from the National Assessment of Educational Progress (NAEP). American Psychologist, 74(4), 445–458. https://doi.org/10.1037/amp0000356
- Renaissance Learning & Education Policy Institute. (2021). Understanding progress in the 2020/21 academic year: Interim findings, January 2021. Department for Education. Retrieved from <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/994364/Understanding_Progress_in_the_2020_21_Academic_Year_Initial_Report_3_.pdf</u>
- Reynolds, K. A., Wry, E., Mullis, I. V. S., & von Davier, M. (2022). PIRLS 2021 Encyclopedia: Education Policy and Curriculum in Reading. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Rogiers, A., Van Keer, H., & Merchie, E. (2020). The profile of the skilled reader: An investigation into the role of reading enjoyment and student characteristics. *International Journal of Educational Research*, 99, 101512.
- Sandilands, D., Oliveri, M. E., Zumbo, B. D., & Ercikan, K. (2013). Investigating sources of differential item functioning in international large-scale assessments using a confirmatory approach. *International Journal of Testing*, *13*(2), 152-174.
- Schleicher, A. (2019). PISA 2018: Insights and Interpretations. OECD Publishing.
- Siddiq, F., & Scherer, R. (2019). Is there a gender gap? A meta-analysis of the gender differences in students' ICT literacy. *Educational research review*, 27, 205-217.
- Snowling, M. J., Lervåg, A., Nash, H. M., & Hulme, C. (2019). Longitudinal relationships between speech perception, phonological skills and reading in children at high-risk of dyslexia. *Developmental Science*, 22(1), e12723. <u>https://doi.org/10.1111/DESC.12723</u>
- Solheim, O. J., & Lundetræ, K. (2018). Can test construction account for varying gender differences in international reading achievement tests of children, adolescents and young adults?—A study based on Nordic results in PIRLS, PISA and PIAAC. Assessment in Education: Principles, Policy & Practice, 25(1), 107–126. https://doi.org/10.1080/0969594X.2016.1239612
- Stainthorp, R. (2020). A national intervention in teaching phonics: A case study from England. *The Educational and Developmental Psychologist*, 37(2), 114-122.
- Støle, H., Wagner, Å. K. H., & Schwippert, K. (2020). The importance of parents' own reading for 10-year old students' reading achievement in the Nordic countries. *Equity, equality and diversity in the Nordic model of education*, 363-384.

- Takriti, R. A., Elhoweris, H., & Atkinson, S. J. (2020). Examining the expectations of early years' teachers in the UAE regarding a successful start to school for children with and without special educational needs. *Early Child Development and Care*, 190(4), 516-525.
- Toste, J. R., Didion, L., Peng, P., Filderman, M. J., & McClelland, A. M. (2020). A metaanalytic review of the relations between motivation and reading achievement for K– 12 students. *Review of Educational Research*, 90(3), 420-456.
- Van Staden, S., Roux, K. (2022). PIRLS 2021 Encyclopedia: South Africa. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://pirls2021.org/encyclopedia</u>
- Vestheim, O. P., Husby, M., Aune, T. K., Bjerkeset, O., & Dalen, T. (2019). A population study of relative age effects on national tests in reading literacy. *Frontiers in Psychology*, 10, 1761.
- Volante, L. & Klinger, D. A. (2021). PISA and Education Reform in Europe: Cases of Policy Inertia, Avoidance, and Refraction, European Education, 53:1, 45-56, DOI: 10.1080/10564934.2021.1987839
- von Davier, M., Mullis, I. V. S., Fishbein, B., & Foy, P. (Eds.). (2023). Methods and Procedures: PIRLS 2021 Technical Report. Retrieved from Boston College, TIMSS & PIRLS International Study Center website: <u>https://tims-</u> <u>sandpirls.bc.edu/pirls2021/methods</u>

List of figures

Figure 1: Performance of England and comparator systems across PIRLS cycles37
Figure 2: Percentage of pupils in England and comparator systems reaching each International Benchmark in PIRLS 2021
Figure 3: Performance of England at PIRLS International Benchmarks across PIRLS cycles with respect to International Medians
Figure 4: Performance of England and comparator systems on the reading for Literary Purpose Scale across PIRLS cycles46
Figure 5: Performance of England and comparator systems on the reading for Informational Purpose Scale across PIRLS cycles47
Figure 6: Differences in Literary and Informational Purpose Scale scores across PIRLS cycles for England and comparator systems
Figure 7: Performance of England and comparator systems on the Retrieving and Straightforward Inferencing Process Scale across PIRLS cycles
Figure 8: Performance of England and comparator systems on the Interpreting, Integrating and Evaluating Process Scale across PIRLS cycles
Figure 9: Differences in comprehension processes scales scores across PIRLS cycles for England and comparator systems
Figure 10: Performance at the 10 th percentile for England and comparator systems across PIRLS cycles61
Figure 11: Performance at the 90th percentile for England and comparator systems across PIRLS cycles
Figure 12: Distribution of England's performance between the 10 th and 90 th percentiles across PIRLS cycles
Figure 13: Comparison of the distributions of England's performance on the Reading Purpose and Comprehension Process subscales in PIRLS 2021 and PIRLS 200165
Figure 14: PIRLS performance between the 10 th and 90 th percentiles for pupils in England by their prior performance in the year 1 phonics screening check
Figure 15: PIRLS performance between the 10th and 90th percentiles for pupils in England by their prior performance in the year 2 phonics screening check

Figure 16: PIRLS performance between the 10 th and 90 th percentiles for pupils in England by their key stage 1 reading outcome
Figure 17: Average performance of girls in England and comparator systems across PIRLS cycles
Figure 18: Average performance of boys in England and comparator systems across PIRLS cycles
Figure 19: Gender gap in England and comparator systems across PIRLS cycles87
Figure 20: Differences in Literary and Informational Purpose Scale scores across PIRLS cycles for girls and boys in England90
Figure 21: Differences in RSI and IIE Process Scale scores across PIRLS cycles for girls and boys in England
Figure 22: Performance trends of girls and boys in England at the 10th percentile across PIRLS cycles
Figure 23: Performance trends of girls and boys in England at the 90th percentile across PIRLS cycles
Figure 24: Average PIRLS 2021 scores of pupils in England by their month of birth95
Figure 25: Percentages of girls and boys in England providing responses to the statement 'Reading is easy for me' across the last 3 PIRLS cycles
Figure 26: Percentages of girls and boys in England providing responses to the statement 'I enjoy reading' across the last 3 PIRLS cycles
Figure 27: Percentages of girls and boys in England providing responses to the statement 'My teacher gives me interesting things to read' across the last 3 PIRLS cycles
Figure 28: Percentages of girls and boys in England reporting different amounts of time they spend reading outside of school each day123
Figure 29: Perceptions of the level of parental involvement in different aspects of pupils' educational experiences - headteachers in England (2021)

List of tables

Table 1: Education systems participating in PIRLS 2001-2021 16
Table 2: England's PIRLS 2021 sample characteristics relative to national data19
Table 3: Comparison of data collection periods in PIRLS 2021 22
Table 4: Performance of education systems in PIRLS 2021 relative to England31
Table 5: Performance of delayed-assessment education systems in PIRLS 2021 relative to England
Table 6: Overall scores of top 15 performing systems in PIRLS 2021, including systemswith delayed assessment
Table 7: Statistically significant changes in overall PIRLS score from 2016 to 202135
Table 8: Statistically significant changes in overall PIRLS score from PIRLS 2016 toPIRLS 2021 for education systems with delayed assessment
Table 9: Overview of PIRLS 2021 International Benchmark criteria and scale scores38
Table 10: Performance of England and comparator systems on the Literary Purpose andInformational Purpose Scales relative to overall performance (2021)
Table 11: Performance of England and comparator systems on the readingcomprehension process scales relative to overall performance (2021)50
Table 12: Average scores at the 10th, 50th and 90th percentile for England andcomparator systems (2021)
Table 13: Description of pupil characteristics variables explored in the regression analysis 81
Table 14: Regression analysis of pupil characteristics relative to PIRLS 2021 score in England
Table 15: Average performance of girls and boys in England and comparator systems inPIRLS 2021
Table 16: Performance of girls and boys in England and comparator systems on theLiterary Purpose Scale in PIRLS 2021
Table 17: Performance of girls and boys in England and comparator systems on theInformational Purpose Scale in PIRLS 2021

Table 18: Performance of girls and boys in England and comparator systems on theRetrieving and Straightforward Inferencing Process Scale in PIRLS 202191
Table 19: Performance of girls and boys in England and comparator systems on theInterpreting, Integrating and Evaluating Process Scale in PIRLS 202191
Table 20: Performance of England's PIRLS 2021 cohort by ethnic group
Table 21: Performance of England's PIRLS 2021 cohort by English as an AdditionalLanguage (EAL) status in the past 6 years
Table 22: Performance of England's PIRLS 2021 cohort by their eligibility for free schoolmeals (FSM) in the past 6 years
Table 23: Average PIRLS 2021 score of pupils in England and comparator systems bythe number of books they reported having at home
Table 24: Pupils' confidence in reading in England and comparator systems in PIRLS2021
Table 25: Pupils' confidence in reading and their performance in PIRLS 2021 in England by gender
Table 26: Pupils' confidence in reading and their performance in PIRLS 2021 in Englandby the number of books at home
Table 27: Pupils' confidence in reading and their performance in PIRLS 2021 by eligibilityfor free school meals (FSM) in the past 6 years
Table 28: Pupils' confidence in reading and their performance in PIRLS 2021 by ethnic
group106
group
Table 29: Pupils' confidence in reading and their performance in PIRLS 2021 by English
Table 29: Pupils' confidence in reading and their performance in PIRLS 2021 by English as an Additional Language (EAL) status
Table 29: Pupils' confidence in reading and their performance in PIRLS 2021 by English as an Additional Language (EAL) status

Table 34: Pupils' liking of reading and their performance in PIRLS 2021 by ethnic group
Table 35: Pupils' liking of reading and their performance in PIRLS 2021 by English as anAdditional Language (EAL) status
Table 36: Pupils' engagement in reading lessons in England and comparator systemsrelative to their performance in PIRLS 2021115
Table 37: Pupils' engagement in reading lessons and their performance in PIRLS 2021 in England by gender
Table 38: Pupils' engagement in reading lessons and their performance in PIRLS 2021 inEngland by number of books at home
Table 39: Pupils' engagement in reading lessons and their performance in PIRLS 2021by eligibility for free school meals (FSM) in the past 6 years
Table 40: Pupils' engagement in reading lessons and their performance in PIRLS 2021 by ethnic group 118
Table 41: Pupils' engagement in reading lessons and their performance in PIRLS 2021by English as an Additional Language (EAL) status
Table 42: Pupils' time spent reading outside of school each day in England and comparator systems
Table 43: Percentage of pupils in England and comparator systems whose teachers havereached or exceeded different education levels (2021)129
Table 44: Percentage of pupils in England and comparator systems whose teachers'formal training emphasised different aspects of reading education (2021)
Table 45: Percentage of pupils in England and comparator systems whose teachers havedifferent amounts of teaching experience (2021)
Table 46: Percentage of pupils in England and comparator systems whose teachers have participated in different areas of reading-related professional development (2021)131
Table 47: Percentage of pupils in England and comparator systems whose teachersreport different levels of career satisfaction (2021)132
Table 48: Percentage of pupils in England and comparator systems whose teachers ask them to complete weekly tasks requiring reading skills A-F (2021)

Table 49: Percentage of pupils in England and comparator systems whose teachers ask them to complete weekly tasks requiring reading skills G-L (2021)
Table 50: Percentage of pupils in England and comparator systems whose teachersassign different instructional materials for reading (2021)
Table 51: Percentage of pupils in England who reported that their teacher gives them interesting things to read
Table 52: Performance of England's PIRLS 2021 cohort by the type of school they attend
Table 53: Percentage of pupils in England and comparator systems whose teachers believe different levels of emphasis are placed on academic success at the school138
Table 54: Percentage of pupils in England and comparator systems whose teachersreport different levels of safety and orderliness at their school
Table 55: Percentage of pupils in England and comparator systems whose headteachersreport different levels of problems associated with school discipline
Table 56: Percentage of pupils in England and comparator systems and the frequency ofbullying they experience at school
Table 57: Percentage of pupils in England and comparator systems whose headteachersbelieve that parents are involved in school activities143
Table 58: Percentage of pupils in England and comparator systems whose headteachers believe that parents are committed to ensuring that pupils are ready to learn
Table 59: Percentage of pupils in England and comparator systems whose headteachersbelieve that parents have high expectations for pupil achievement
Table 60: Percentage of pupils in England and comparator systems whose headteachersbelieve that parents support the school in ensuring high pupil achievement
Table 61: Number of weeks in which primary school operations were affected by theCOVID-19 pandemic in 2020-21 (percent of schools by education system)
Table 62: Resources offered when normal school operations were disrupted by theCOVID-19 pandemic (percent of schools by education system)
Table 63: Trends across 2016 and 2021 PIRLS cycles according to data collectionadaptations due to the COVID-19 pandemic

Table 64: Key stage 1 and 2 reading comprehension skills and knowledge and their association with PIRLS International Benchmark levels	7
Table 65: Education systems testing in a single language in PIRLS 2021	2
Table 66: Education systems testing in multiple languages in PIRLS 2021183	3
Table 67: Missing NPD data for pupils in England's PIRLS 2021 sample	1
Table 68: Percentage of pupils in England and comparator systems whose teachersbelieve that parents are involved in school activities18	5
Table 69: Percentage of pupils in England and comparator systems whose teachers believe that parents are committed to ensuring that pupils are ready to learn	5
Table 70: Percentage of pupils in England and comparator systems whose teachersbelieve that parents have high expectations for pupil achievement	5
Table 71: Percentage of pupils in England and comparator systems whose teachers believe that parents support the school in ensuring high pupil achievement	6

Appendices

Appendix A: Methodology

PIRLS sampling methodology and England sample

The sample design in PIRLS is intended to provide a nationally-representative, unbiased and internationally-comparable estimate of the reading performance, attitudes and behaviours of pupils in the target population within each participating education system. The target population is defined as pupils in their fourth year of formal schooling provided their mean age at the time of testing in a particular education system is at least 9.5 years. In England, this means that the target population is year 5, as year 4 (age 8-9 years) would not meet the necessary age criterion.

Education systems transitioning to digitalPIRLS were subject to more complicated sampling considerations to accommodate the need for both computer-based and paper assessment samples to allow for linking across different modes of assessment and measurement of trends from previous assessments. This did not apply in England, where PIRLS 2021 continued with the paper-and-pencil assessment as in previous cycles.

To achieve a nationally-representative sample, a two-stage random sample design was used, first drawing a sample of schools and then drawing a sample of one or more classes within sampled schools. Classes in the sample were included intact, rather than drawing samples of pupils from within classes across a year group. The reasons for this were twofold. First, this approach would cause less disruption to regular school operations as pupils normally in a class together could be assessed together. Second, PIRLS looks at pupils' experiences of reading-related classroom instruction and curriculum, aspects that tend to be organised at the classroom level.

Pearson led on project activities relevant to sampling and data collection in England, in collaboration with the IEA and Statistics Canada. Statistics Canada provided a main sample of 169 schools and about 4,500 pupils based on a sample frame of primary schools in England stratified by type (state-funded, independent, academy) and by attainment (based on attainment at KS2 in quintiles with an extra category for independent schools for which KS2 results were not available). For each of the schools in the main sample, Statistics Canada also provided a matched replacement school with similar characteristics. Some schools were excluded from the sampling frame. Exclusion criteria were:

• Very small schools, specifically those with fewer than 9 pupils in year 5; this criterion was based on IEA's requirement that small school exclusions not exceed 2% of the sample

- Special schools
- Alternative Provision
- Pupil Referral Units

Pearson successfully recruited 153 schools from the main sample followed by 14 replacement sample schools. Two schools from the main sample and 1 replacement school withdrew after signing up, leaving 167 participating schools by the end of the recruitment window. Three more schools withdrew from participating between recruitment and testing, and an additional school dropped out during the testing period, leaving 163 schools. One additional school was removed from the sample after testing as the school returned pupil questionnaires but no assessment booklets, leaving a final analytical sample of 162 schools.

Within participating classes, all pupils in the participating classes were expected to take part in PIRLS data collection activities except for those with functional or learning disabilities and those who were non-native language speakers (according to the terminology of the IEA). Pupils classified as having EAL, for example, were not excluded by default; IEA guidelines suggested that pupils with less than a year of instruction in English should be excluded. Similarly, IEA guidance indicated that pupils with Special Educational Needs and Disabilities (SEND) should not be excluded by default due to low attainment or disciplinary issues, and that accommodations should be put in place wherever possible to allow pupils to participate. Exclusions based on SEND were to be limited to cases in which, based on the professional judgment of headteachers or other qualified school staff, a pupil could not physically perform under PIRLS testing conditions or they could not follow the general instructions of the test. These criteria led to the exclusion of 161 pupils in total, 17 with functional disabilities, 100 with learning disabilities and 44 non-native language speakers, making up 3.5% of the sample in total.

Overall, 4543 pupils were included in the sample, of which 4166 (92%) completed assessment booklets and 4106 (90%) completed pupil questionnaires. Pearson collected information on pupil absences, of which there were 437 in total. Of these, 62 pupils had left their schools permanently, 86 had parental permission denied, and 289 were absent for any other reason. Of 179 teachers in the sample, 167 (93%) returned questionnaires. School questionnaires based on head-teacher responses were completed in 146 out of 163 schools (90%). This meant that the IEA's minimum participation threshold of 85% was met across England's sample of schools, teachers and pupils.

Survey administration

A field trial was conducted in early 2020, prior to the main study. Information from the field trial helped to inform school recruitment processes as well as items included in the data collection instruments in the main study, in order to ensure that assessments and

questionnaires would collect meaningful information and that sampling and recruitment would go as smoothly as possible.

Data collection took place between mid-May and mid-July 2022, following the decision to delay data collection by 12 months from when it was originally planned in England due to the COVID-19 pandemic. An additional step taken as a consequence of the pandemic was to provide the option – not previously offered in PIRLS in England, though offered in some other education systems as standard practice – for schools to self-administer PIRLS tests and questionnaires rather than accommodating external visitors for the purposes of this research. Schools were given an opportunity to express a preference upon being recruited, and later those that initially opted to self-administer were also given 2 further opportunities to change their choice and involve a Pearson Test Administrator (TA) instead. In the final sample of 162 schools, 108 chose to self-administer, and another 54 decided to have a Pearson TA. Each school participating in the main study, whether self-administering or not, nominated a school coordinator who liaised with Pearson contacts to facilitate sampling and data collection processes and to ensure that these adhered to the international guidelines from IEA.

Test materials from the IEA, including assessment booklets and pupil, teacher and headteacher questionnaires, underwent national adaptations to ensure that the materials were appropriate for the English target population. Adaptations largely involved changing American English words and phrases to British English to ensure that they would be accessible to British pupils, and no questions were added or removed from the assessment booklets. Two curriculum experts with extensive key stage 2 experience reviewed the assessment booklets and confirmed that the content was suitable for pupils in year 5 and included appropriate coverage of the relevant national curriculum content. Pupil and teacher questionnaires did not have any national questions added, but 3 questions were added to the school (headteacher) questionnaire addressing the effect of the COVID-19 pandemic on normal school operations and the resources provided when normal school operations were disrupted. These items pertained explicitly to 2020/21, the original year during which data collection was planned, which may have caused some confusion in England given the choice to delay data collection to 2021/22 (see Chapter 9 for more on these questions).

An International Quality Assurance Program was developed by the IEA to ensure that data collection activities followed standardized procedures, and this involved visits by International and National Quality Control Monitors to 15 schools (approximately 10% of the sample) in each participating education system to observe the administration of PIRLS 2021. In England, feedback from Quality Control Monitors indicated a high level of quality of delivery on the part of test administrators and good adherence to administration procedures; this was the case both for school-based and Pearson TAs.

Data from the pupil assessments and questionnaires, teacher questionnaires and school questionnaires were sent to IEA via secure file sharing portal for upload into the IEA software CodingExpert. 22 coders and 5 coding Team Leaders were recruited by Pearson, trained in July 2022 and coded the data between late July and early August 2022. Coded, cleaned and checked data for England's sample was submitted to IEA by Pearson in early September 2022, followed by data checks on the assessment and questionnaire data by IEA. IEA returned only minor queries which were addressed by Pearson and returned with no further queries by late September 2022. Data transfers between collaborating organisations used secure file transfers and uploads to secure servers, and transfer of paper materials between Pearson and schools used a courier service throughout the study. Data protection was carefully monitored throughout the study and there were no breaches.

Data analysis

The IEA merged England's final data with that of other participating education systems into an international database. IEA's analysis of the international results is available in the *PIRLS 2021 International Results in Reading* report (Mullis et al., 2023). The international database was released to participating education systems' National Research Centres on February 2, 2023, and released publicly on June 22, 2023. This National Report is based on these data as well as linked pupil information from the NPD to contextualise the PIRLS results. The linked data from the NPD were delivered directly by secure file transfer to the OUCEA research team, and held securely on a University server.

Analysis undertaken by the OUCEA research team used the IEA's International Database (IDB) Analyzer, IBM SPSS and R software. Appropriate analytical techniques were used to account for the sample design and data structure, including pooling of results across plausible values and use of jackknife replicate weights. Not all pupils in the PIRLS sample could be matched to NPD records, and the analyses using the linked NPD-PIRLS dataset in this report are prefaced by reporting of coverage of the PIRLS sample in the matched NPD sample. In most cases the instances in which a match could not be found in the NPD was due to pupils' enrolment in independent schools not included in the NPD.

Sources of further information

For more information on the broader PIRLS sample design and implementation, instrument development, translation, quality assurance, creation of the international database and scaling methodology, please see <u>https://pirls2021.org/methods/</u>.

For the *PIRLS 2021 Assessment Frameworks*, please see <u>https://pirls2021.org/frameworks/</u>.

For the *PIRLS 2021 Encyclopedia* documenting education policy and curriculum in participating education systems, please see <u>https://pirls2021.org/encyclopedia/</u>.

For further information on PIRLS 2021 and the international report, please see <u>https://www.iea.nl/studies/iea/pirls/2021</u>.

Appendix B: Missing data from the NPD

Table 65: Missing NPD data for pupils in England's PIRLS 2021 sample

Pupil background characteristic	Number of pupils with PIRLS data but missing NPD data	Percent of pupils with PIRLS data but missing NPD data	
No NPD match	373	9%	
Missing ethnic group*	411	10%	
Missing EAL	376	9%	
Missing Ever6FSM	373	9%	
Missing year 1 phonics check score	531	13%	
Missing KS1 reading level	479	12%	

* Ethnic group was defined as missing if no information was available for a particular pupil or if a student was listed as 'unclassified'. Note that counts and percents are not cumulative, for example pupils without any NPD match are also counted as missing ethnic group, EAL, Ever6FSM, year 1phonics screening check and KS1 reading level.

Sources: National Pupil Database (NPD) and IEA's PIRLS 2021

Appendix C: Test languages in PIRLS 2021

Table 66: Education systems testing in a single language in PIRLS 2021

Education System	Test language
Albania	Albanian
Australia	English
Austria	German
Belgium (Flemish)	Dutch
Belgium (French)	French
Brazil	Portuguese
Bulgaria	Bulgarian
Croatia	Croatian
Czech Republic	Czech
Denmark	Danish
Egypt	Arabic
England	English
France	French
Georgia	Georgian
Germany	German
Hong Kong, SAR	Chinese
Hungary	Hungarian
Iran	Persian
Ireland	English
Italy	Italian
Jordan	Arabic
Kosovo	Albanian
Montenegro	Montenegrin (Cyrillic)
Morocco	Arabic
Netherlands	Dutch
Northern Ireland	English
Poland	Polish
Portugal	Portuguese
Russia	Russian
Serbia	Serbian (Cyrillic)
Singapore	English
Slovenia	Slovenian
Sweden	Swedish

Education System	Test language
Taiwan	Chinese
Turkey	Turkish
United States	English

Source: IEA's PIRLS 2021

Table 67: Education systems testing in multiple languages in PIRLS 2021

Education System	Test languages	Percentage of pupils
Azerbaijan	Azerbaijani, Russian	90% , 10%
Bahrain	Arabic, English	68% , 32%
Canada (Alberta)	English, French	98% , 2%
Canada (British Columbia)	English, French	98% , 2%
Canada (Quebec)	French, English	91% , 9%
Cyprus	Greek, English	92% , 8%
Finland	Finnish, Swedish	94% , 6%
Israel	Hebrew, Arabic	75% , 25%
Kazakhstan	Kazakh, Russian	66% , 34%
Latvia	Latvian, Russian	76% , 24%
Lithuania	Lithuanian , Polish, Russian	89% , 6%, 6%
Macao SAR	Chinese , English, Portuguese	90% , 9%, 1%
Malta	English, Maltese	75% , 25%
New Zealand	English, Māori	98% , 2%
North Macedonia	Macedonian, Albanian	66% , 34%
Norway	Nynorsk, Bokmål	92% , 8%
Oman	Arabic, English	92% , 8%
Qatar	English, Arabic	55% , 45%
Saudi Arabia	Arabic, English	95% , 5%
Slovak Republic	Slovak, Hungarian	95% , 6%
South Africa	English , isiZulu, isiXhosa, Sepedi, Afrikaans, Sesotho, Setswana, Xitsonga, siSwati, Tshivenda, isiNdebele	22% , 20%, 15%, 10%, 9%, 8%, 8%, 3%, 2%, 2%, 1%
South Africa (Grade 6)	English, Afrikaans	91% , 9%
Spain	Spanish , Catalan, Basque, Galician, Valencian	70% , 18%, 4%, 4%, 3%

Education System	Test languages	Percentage of pupils
United Arab Emirates	English, Arabic	76% , 24%
United Arab Emirates (Abu Dhabi)	English, Arabic	92% , 8%
United Arab Emirates (Dubai)	English, Arabic	88% , 12%
Uzbekistan	Uzbek (Latin) , Russian, Karakalpak	88% , 9%, 3%

Because of rounding, some results may appear inconsistent. Most common test language is in bold

Appendix D: Teachers' perspectives on parental involvement

Table 68: Percentage of pupils in England and comparator systems whoseteachers believe that parents are involved in school activities

Education system	Very high involvement	High involvement	Medium involvement	Low involvement
England	5%	25%	40%	29%
Singapore	5%	34%	43%	18%
Hong Kong SAR	5%	26%	53%	15%
Australia	8%	21%	39%	32%

Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 69: Percentage of pupils in England and comparator systems whose teachers believe that parents are committed to ensuring that pupils are ready to learn

Education system	Very high commitment	High commitment	Medium commitment	Low commitment
England	5%	24%	45%	26%
Singapore	7%	29%	49%	15%
Hong Kong SAR	1%	21%	54%	24%
Australia	9%	30%	44%	18%

Because of rounding, some results may appear inconsistent.

Source: IEA's PIRLS 2021

Table 70: Percentage of pupils in England and comparator systems whose teachers believe that parents have high expectations for pupil achievement

Education system	Very high expectations	High expectations	Medium expectations	Low expectations
England	11%	31%	41%	17%
Singapore	12%	50%	33%	6%
Hong Kong SAR	12%	47%	36%	6%
Australia	20%	33%	38%	9%

Because of rounding, some results may appear inconsistent.

Table 71: Percentage of pupils in England and comparator systems whoseteachers believe that parents support the school in ensuring high pupilachievement

Education system	Very high support	High support	Medium support	Low support
England	10%	22%	52%	16%
Singapore	6%	33%	47%	13%
Hong Kong SAR	5%	36%	50%	10%
Australia	7%	30%	49%	14%

Because of rounding, some results may appear inconsistent.

Appendix E: Corrections included in this version of the report

The authors have made minor amendments to the original version of this report, including to correct typos and make minor changes to tables or figures, mainly to resolve an issue with weighting that arose due to problems with SRS functionality and timescales, where this affected some of the analyses that used linked PIRLS and NPD data. Corresponding text was updated to reflect changes where relevant. Amendments in this version are:

- p7, p8 and p72, correlation amended from 0.46 to 0.43.
- p44 & p50, repetition of text deleted.
- p49, corrected to reflect 30% 'interpret and integrate', 20% 'evaluate and critique'.
- p65, small amendments to fix typos in Figure 13 and accompanying table.
- p70, 3,621 corrected to 3,619 pupils.
- Tables and figures slightly amended to reflect appropriate weighting: Figure 14, p71; Figure 15, p73; Figure 16, p75; Table 52, p136; Table 65, p180; on p74 and elsewhere, "over 90%" amended to "nearly 90%".
- p81, counts and percentages slightly amended to reflect the use of PIRLS (not NPD) gender data in the regression model.
- p59, Table 12 and p84, Table 15, International Median differences corrected to reflect difference of medians rather than medians of differences.
- p88, p89 & p91, International Median values for gender groups slightly corrected and International Median Gender Gaps corrected to reflect differences of medians rather than medians of differences.
- p105, Table 27, FSM non-eligible pupil score corrected from 560 to 600. *Only this amendment affected direction or substantive meaning of a relationship.*
- p106, Table 28, "Not very confident" for White pupils amended to 21% from 23%.
- p109, Table 30, Singapore points difference amended from 47 to 46.
- p112, Table 34, "Somewhat likes reading" for Other pupils amended from 47% to 48%.
- p114, Figure 26, "Agree a little" for Boys 2021 amended from 31% to 30%.
- p128, Table 43, categories of formal education amended to reflect those in PIRLS (e.g. "Post-secondary, non-tertiary" corrected to "Post-secondary or short-cycle tertiary"); values unchanged.
- p128, p129, p136, p138 & p143, International Medians in text slightly amended, and text referring to the International Median for plays on p134 clarified.
- p133 & p134, "at least weekly" clarified in titles for Tables 48, 49 & 50.
- p135, Table 51 title amended to read "England and comparator systems".
- p138, 'very' to 'somewhat safe and orderly' points difference in text amended from 16 to 14.
- p140, Table 56 final column amended to read "About weekly".
- p141, rounding and label errors resolved with corresponding changes to Figure 29 and table.
- p144, "<2%" corrected to "1%" for consistency in Table 60.
- p148 & p149, small corrections to International Median values in Table 61 & Table 62.

• p152-3, corrections to Table 63 to include significant differences for delayed assessment education systems.



© OUCEA, Department of Education, University of Oxford 2024

This publication is licensed under the terms of the Open Government Licence v3.0 except where otherwise stated. To view this licence, visit nationalarchives.gov.uk/doc/open-government-licence/version/3.

Where we have identified any third-party copyright information you will need to obtain permission from the copyright holders concerned.

Reference: DfE RR1353

ISBN: 978-1-83870-473-5

For any enquiries regarding this publication, contact us at: <u>international.surveys@education.gov.uk</u> or <u>www.education.gov.uk/contactus</u>

This document is available for download at <u>www.gov.uk/government/publications</u>